



Vakgroep Biologie



Krijgslaan 281-S8 9000 Ghent (Belgium)



Phytoplankton community structure and cyanobacterial blooms in the semi-arid highlands of Tigray, Ethiopia

> Proefschrift voorgelegd tot het behalen van de graad van Doctor in de Wetenschappen: Biologie

> > Tsehaye Asmelash Dejene 2009





Krijgslaan 281-S8 9000 Ghent (Belgium)

Phytoplankton community structure and cyanobacterial

blooms in the semi-arid highlands of Tigray, Ethiopia

Promoter:

Prof. Dr. Wim Vyverman

Co-Promoter:

Prof. Dr. Luc De Meester

behalen van de graad van Doctor in de Wetenschappen: Biologie

Proefschrift voorgelegd tot het

Tsehaye Asmelash Dejene

The research reported in this thesis was performed in the laboratory of Protistology and Aquatic Ecology, Biology Department, Ghent University, Krijgslaan 281-S8, B-9000 Ghent, Belgium. http://www.pae.ugent.be/



Vakgroep Biologie



Krijgslaan 281-S8 9000 Ghent (Belgium)

Promoter: Prof. Dr. Wim Vyverman¹ Co-Promoter: Prof. Dr. Luc De Meester²

Overige leden van de examencommisie: Prof. Dr. Bian Moss³ Dr. Steven Declerck² Prof. Dr. Koenraad Muylaert⁴ Dr. Pieter Voorvedediging¹ Dr. K. Van der Gucht¹ Prof. Dr. Adriaens¹ (voorzitter)

- Laboratory of Aquatic Ecology and Protistology, Gent University, Krijgslaan 281, S8, B-9000 Gent, Belgium
- K.U. Leuven, Laboratory of Aquatic Ecology and Evolutionary Biology, Ch. Deb'eriotstraat 32, 3000 Leuven, Belgium
- School of Biological Sciences, University of Liverpool, Derby Building, Liverpool, L69 3BX, UK
- ^{4.} K.U. Leuven Campus Kortrijk, Etienne Sabbelaan 53, 8500 Kortrijk, Belgium

List of Figures

	Figure	Page
1.1 1.2	Map of Ethiopia Monthly rainfall for selected meteorological stations around the study area (Tigray, North Ethiopia) for a period of 4 years (2003-2006), showing variation among years and sites	2 4
13	Map of Tigray with distribution of reservoirs and lakes	6
1.4	Photograph of a cyanobacterial bloom in one of the study reservoirs (Mai Nigus) and microphotographs of cyanobacterial colonies and filaments showing some of their morphological diversity	12
1.5	Structure of the toxic cyclic peptides (microcystin, nodularin, and cylindrospermopsin) and alkaloids (anatoxin-a, saxitoxins, and anatoxin-a(s) produced by cyanobacteria	17
1.6	Schematic model of the different components occurring in pelagic food webs	21
2.1	Map with the geographic location of the sampled reservoirs in Tigray, North Ethiopia	66
2.2	Shannon-Wiener diversity values of phytoplankton communities in the wet and dry seasons of the shallow reservoirs in a semi-arid subtropical region	72
2.3	Biplot of PCA analysis performed on the phytoplankton communities in the wet and dry seasons	73
2.4	Scatter plots of pH versus <i>Microcystis</i> in the rainy season and reservoir depth versus <i>Peridinium</i> in the dry season in reservoirs in Tigray, Ethionia	74
3.1	The geographic location of the eight reservoirs in Tigray, North Ethiopia, selected for the seasonal study of environmental variables and associated phytoplankton communities	99
3.2	First and second axis of a partial RDA on the response of environmental variables in function of month of the year, corrected for the influence of reservoir by including reservoirs as co-variables	104
3.3	Changes in environmental variables during the course of one year in eight reservoirs in Tigray North Ethiopia	107
3.4	Changes in phytoplankton biomass during the course of one year in eight reservoirs in Tigray. North Ethiopia	108
3.5	Partial RDA biplot on changes in phytoplankton taxon in function of time, corrected for the influence of reservoir by including reservoirs as co-variables	110
3.6	Average total phytoplankton biomass in Tigray reservoirs during the course of one year	112
3.7	Standardized RDA plot on environmental variables and phytoplankton community in function of time after correcting for among reservoir differences	113
4.1	Map with geographical location of the studied reservoirs in Tigray, North Ethiopia	126
4.2	Principal component analysis of bacterial community composition in the studied reservoirs in the wet and dry seasons	134
4.3	Variance partitioning of the BCC in the wet season. Both % Microcystis	135

and Copepod biomass explained variation undependently, there were no common effects

- 5.1 PCA and RDA ordination plots, showing the response of limnological 156 variables and zooplankton communities to the presence of fish in enclosures
- 5.2 The response of abiotic and phytoplankton variables to the presence of 157 fish in the enclosure experiments in Gereb Awso and Tsinkanet
- 5.3 Population densities of cladoceran taxa in response to the presence of 158 fish in the enclosure experiments in Gereb Awso and Tsinkanet
- 5.4 The association between *Daphnia carinata* abundance and the relative 161 biomass of *Microcystis* in the experiments in Gereb Awso and Tsinkanet
- 5.5 Biomass of different phytoplankton taxa in Gereb Awso and Tsinkanet 173 during enclosure experiments
- 5.6 Triplot of PCA results performed on biomass of phytoplankton taxa (a) 174 and triplots of RDA results showing the response to the presence of fish (Garra) of the biomass of phytoplankton taxa in the four enclosure experiments (b)
- 5.7 Triplot of PCA results on the Microcystis ITS (a) and RDA triplots on 177 Microcystis ITS (b-d)
- 6.1 Biomass of major phytoplankton taxa in the experimental units on the 191 starting day (day 0) of the experiment
- 6.2 Changes in total biomass and biomass of the major phytoplankton taxa 193 in the different treatments during the course of the experiment
- 6.3 Changes in biomass of major zooplankton taxa during the course of the 194 enclosure experiment
- 6.4 Partial RDA biplot showing the response of phytoplankton taxon 196 composition to the source water and zooplankton treatments correcting for impact of time
- 6.5 Partial RDA biplot on the response of phytoplankton taxon composition 198 in function of time, correcting for among treatment differences
- 6.6 Plots representing associations between microcystin production and 199 zooplankton treatments in the enclosure experiments

List of Tables

	Table	Page
1.1	Mass mortalities of flamingo species (<i>Phoenocopterus</i>) after exposure to ingestion of cyanobacterial biomass	13
1.2	Examples of human exposure to cyanobacterial blooms and toxins, with associated health outcomes	14
1.3	Geographic co-ordinates, altitude, surface area and depth of reservoirs studied by Tadesse et al., 2008	33
1.4	Summary of some limnological variables of the reservoirs in Tigray for the wet and dry seasons of the year 2004/2005	34
2.1	List of environmental variables measured for the reservoirs and their abbreviations with the average value and range	68
2.2	Environmental variables significantly explaining variation in the phytoplankton communities in the rainy and dry season respectively, as identified by a forward selection Redundancy Analysis	75
2.3	Multiple regression results for factors potentially influencing <i>Microcystis</i> biomass	76
3.1	General characteristics of the eight reservoirs selected for assessing seasonal dynamics of environmental variables and associated phytoplankton communities	100
3.2	Results of Friedman ANOVA and Kendall coefficient of concordance calculations on the temporal dynamics of environmental variables recorded for eight reservoirs between September 2005 and August 2006	105
3.3	Results of Friedman ANOVA and Kendall coefficient of concordance calculations on the temporal dynamics of phytoplankton communities recorded for eight reservoirs in Tigray, Ethiopia, between September 2005 and August 2006	111
4.1	Summary statistics of limnological variables in the 32 studied reservoirs during the wet and dry season (2004 and 2005 respectively)	128
4.2	Occurrence of <i>Microcystis</i> blooms in highland reservoirs in Tigray during the wet and dry seasons	133
5.1	Results of repeated measures of ANOVA testing for the effect of fish, time and the interaction on densities of <i>Daphnia carinata</i> and <i>D. barbata</i> during experiments	160
5.2	Results of repeated measures of ANOVA on log transformed biomass data testing for the effect of fish and time on total phytoplankton biomass, biomass of Diatoms, and <i>Microcystis during</i> enclosure experiments	175
6.1	Results of regression analysis testing for the effect of zooplankton and source water on phytoplankton taxa biomass in the enclosure experiment	197
6.2	Results of two way ANOVAs testing for the effect of zooplankton,	107

water source and their interaction on log transformed biomasses of 197 different phytoplankton taxa in the enclosure experiment

Acknowledgments

This dissertation holds far more than the conclusion of the years of my study. It also reflects the relationships with many kind and inspiring people I have met since beginning my research. This list is long and may be far from complete, but I appreciate each contribution made to my success.

Firstly, I would like to thank my parents for giving me the opportunity to go to school and their continued encouragement and support all the way to the successful completion of this long journey. I thank Mekelle University, my home institution, for allowing me a study leave to pursue this research and the invaluable support, logistical or otherwise, during the fieldwork. I thank all my friends and colleagues at MU for their friendship and encouragement. The department of Medical Microbiology, my home department, and the staff were helpful during my stay in Mekelle for my field work.

This research was carried out in the framework of VLIR IUC scientific cooperation program between several Flemish universities (Belgium) and Mekelle University (Ethiopia). I thank the Flemish government for the financial support of this program.

I am also grateful to the MU-IUC program and the coordination office. Dr Mintesinot Behailu and Dr Mulu Bayray served as local program coordinators. Both were helpful in solving practical administrative issues. Thank you both. I would also like to thank Dr Kindeya Gebrehiwot (current program coordinator) for his logistic support; the field experiments we carried out would have been most difficult without his administrative intervention. The MU-IUC provided me all the logistics for the field work including the very reliable vehicles along the most reliable drivers. Berhe, Desta, Haile, Samson, and Kahsu, you make the long field drives enjoyable and safe, thank you all. The MU-IUC staff deserves special note of thanks for the services they gave me during the field work: Dr Tarekegn Tadesse, Dr Nuruhsen Teha, Yirga Marye, Mulugeta Hagos and Selamawit Girmay. I am grateful to all of you. Many people joined us in the field work. My special thanks go to Belay Gebreyohannes, Teklit Gebregiorgis, Hayal Lemma, Sahle Simur, Abraha Gebrekidan, Afework Mulugeta and Mekonen Teferi for their support and encouragement during the fieldwork.

I am greatly indebted to my promoter Prof. Dr. Wim Vyverman for his support, encouragement and guidance. My sincere thanks are also reserved to my copromoter Prof. Dr. Luc De Meester. I enjoyed their company in the field and their comments were crucial to shape the research to its current content. I have no words to thank Dr. Katleen Van der Gucht for her all rounded assistance. She joined me in the field during the selection of the study sites and her contribution on sample collection, preservation, and processing for rDNA profiling was crucial. I am also greatly indebted to Dr Steven Declerck for his assistance in the field. His company together with Prof. Dr Luc De Meester in the fieldwork and setting up the enclosure experiments were memorable and greatly contributed to the contents of the output. Luc De Meester also critically read five of my chapters.

Prof. Dr. Brian Moss, Prof. Dr. K. Muylaart, Dr Steven Declerck, Dr Pieter Vanormelingen, and Dr. Katleen Van der Gucht gave valuable comments and advices on an earlier version of this thesis which significantly increased the quality of the final version. I thank also Prof. Dr. Brian Moss for accepting to be a member of the evaluation committee, and for coming to Belgium.

I get important comments and guidance from Dr Jan Nyssen. The support given by Prof. Dr. Jos Snoeks during the various field trips has also been very useful for me. Comments from Prof. Mitiku Haile and Prof. Sepe Deckers during field visits of the study sites were crucial and inspiring. I would like to express my deepest gratitude to my friend and coworker Tadesse Dejenie. His work on identification of zooplankton in the samples has been crucial to several chapters and the overall interpretation of my results. Tadesse and I together with other colleagues had a very enjoyable time during the extended period of joint fieldwork. Tadesse was a source of encouragement in the field during times of frustration. The humors from Tadesse and Berhe ("shebeto shufer" - the driver with white hair) in the field are unforgettable.

I would like also to convey words of appreciation to Ineke van Gremberghe and Sofie D'hondt for their help in the PCR-DGGE and microcystin analyses of my field collected samples. Ineke's help has been on my side during my stay in the lab. Jeroen Van Wichelem, and others in the lab also deserve words of thanks for their help in identification of phytoplankton taxa. Pieter Vanormelingen provided me unreserved support when called for. Thank you. I thank the Laboratory for Protistology and Aquatic Ecology group at UGent for trying to create an environment for academic discourse and comfort during my stay. Thank you very much indeed. Many thanks are also due to Leen De Groot who showed full cooperation when called for during my stay.

I thank the Geographical and Statistical Information Management section of the Bureau of Finance and Economic Development of the Tigray Regional National State for allowing me to get the digital form of the map of Tigray. Special thanks to Habtom GM for his help in preparing the map of Tigray (Fig. 1.6, chapter 1) with distribution of reservoirs and lakes in the region. I am very grateful to my friend Tesfamicahael Gebreyohannes, for preparing the maps of Tigray with the geographic locations of the study reservoirs.

The city of Gent was my home away from home. I lived in Mr. Guido's house for most of my stay in Gent, a friendly, caring and responsible person. I am thankful to my fellow Ethiopians, Sofia and Rishan Kahsay, who were caring in my stay in Belgium. Many thanks to Andre Crismer and Almaz for their generous offer to visit them and the chance I got to visit parts of Belgium which I would not have done it in my own. I am greatly indebted to Tesfamariam Habtemichael and Arsema Assefa for their moral support. My presence with them and their lovely twin sons, Abenezer and Nahom, is unforgettable.

My fellow Ethiopian students in various universities in Belgium deserve my words of thanks, they were sources of encouragement: Ferede Ali, Seifu Hagos, Tesfamicahel Gebreyohannes, Solomon Gebremariam, Fredu Nega, Zaid Negash, Amanuel Abreha, Bedru Bablo, Mekonnen Gebreselassie, Ayele Teshome, Biniam Zeray, Laekemariam Kassa, Dawit Shawol, Tigist Oicha and many others for their good company and moral support. Aster Assefa deserves special words of thanks. I would like also to thank all the close friends Fitsum Hagos, Kurkura Kabeto, Alem Mebrahtu, Hagos Weldegabriel, Fikru Yifter, Znawi Zerihun, Afework Mulugeta and Nata Tadesse who stood by me in moments of frustration and joy. Fitsum Hagos and Kurkura Kabeto with his family (Haregua, Tamrat and Sakura) were so caring to my family in my absence. Thank you.

My heartfelt thanks goes also to my beloved wife Firewoini and my lovely sons Bruk and Abreham for understanding my long absences and endless working hours and wholehearted support and encouragement in the course of the study. Firewoini has always been of great moral support and happily took the burden of family responsibility so that I could concentrate on my study. I am grateful to my sister-in-law, Samrawit, and my brother, Alem, for taking care of our sons when my wife and I both were absent.

Tsehaye Asmelash Dejene Gent, Belgium April, 2009.

Table of Contents

	Content	Page
	List of figures	i
	List of tables	iii
	Acknowledgments	v
	Table of contents	ix
Chapter 1	Introduction and thesis outline	1
Chapter 2	The phytoplankton communities of shallow reservoirs in a semi-	61
	arid subtropical region (Tigray, North Ethiopia), with an	
	emphasis on the occurrence of bloom-forming cyanobacteria	
Chapter 3	Seasonal dynamics of environmental variables and associated	95
	phytoplankton community structure in tropical semi-arid	
	highland reservoirs of Tigray, North Ethiopia	
Chapter 4	Bacterial community composition in semi-arid highland	121
	reservoirs in Tigray, North Ethiopia – impact of environmental	
	factors and Microcystis blooms	
Chapter 5	Impact of the fish Garra on the ecology of reservoirs and the	147
	occurrence of Microcystis blooms in semi-arid tropical highlands:	
	an experimental assessment using enclosures	
Chapter 6	An experimental assessment of potential top-down effects of	181
	zooplankton on the phytoplankton communities, including toxic	
	cyanobacteria, of eutrophic shallow reservoirs in Tigray, North	
	Ethiopia	
Chapter 7	General Discussion	213
	Summary	239

Chapter 1

1. Introduction and thesis outline

1.1. General introduction

Ethiopia is located in the easternmost part of the African continent. It stretches between 3° N and 15° N Latitudes and 33° E to 48° E Longitudes and is bordered in the north by Eritrea, in the south by Kenya, in the east and southeast by Djibouti and Somalia, and in the west by Sudan (Fig. 1.1). Ethiopia is the third largest country in Africa with an area of over one million km², and has more than 80 million inhabitants (CSA, 2007). The country is endowed with a variety of agro-ecological conditions ranging from desert to rainforest and from 120 meters below sea level (the Denakil Depression) to highlands with altitudes of over 4600 metres above sea level (m.a.s.l). Its complex topography and wide altitudinal variation also ensure a variety of temperature and rainfall patterns. The traditional Ethiopian classification of climate is based on elevation and recognizes at least three zones: 1) the Kolla zone (the hot lowlands) below 1800 metres, with mean annual temperatures of 20-28°C, 2) the Woina Dega (temperate) zone between 1800 and 2400 metres, with mean annual temperatures of 16-20°C, and 3) the Dega (cool) zone above 2400 metres, with mean annual temperatures of 6-16°C. Most of the population inhabits the Woina Dega and Dega zones that are cooler, healthier (lower incidence of parasitic diseases) and more suitable for agriculture.

The Ethiopian highlands (more than 1500 m.a.s.l) cover about 500,000 km². They represent about 43% of the country but support about 88% of the population (MNRDEP, 1994; Mohamed-Saleem, 1995), and account for 95% of the regularly cropped land, more than 70% of the livestock, and 90% of the economic activities of the country (Constable, 1984; FAO, 1986). The population density in the highlands is close to ten times that of the

lowlands. The highlands have been settled for millennia and are known for a similar longstanding agricultural history (McCann, 1995). This long history of settlement and high population pressure has lead to un-sustainability in agriculture. The basis for the early development of agriculture and high human population densities in this agro-ecological zone likely has been the reasonably favourable climatic and ecological conditions, with sufficient rainfall, moderate temperatures, less incidence of tropical diseases, and welldeveloped soils. The highlands of Ethiopia are considered to be amongst the most degraded lands in Africa (El-Swaify and Hurni, 1996). The FAO (1986) estimated that 50% of the highlands are significantly eroded, 25% are seriously eroded and 4% have reached the point of no return.



Fig.1.1 Map of Ethiopia (from: <u>http://www.fao.org/nr/water/aquastat/countries/</u> ethiopia/ethiopia.pdf)

Most farmers in the Ethiopian highlands depend on rain-fed agriculture. Rainfall in Ethiopia is highly erratic with an extreme spatial and temporal variability. The country is often divided into three regions according to seasonal rainfall patterns with distinctive characteristics: 1) an extended single wet season in the south-west 2) a shorter single wet season further north and 3) a bi-modal pattern in the east with a short wet season in March-May preceding the main wet season. The erratic nature and extreme spatial and temporal variability of rainfall causes major socio-economic problems in the country (Shanko and Chamberlain 1998). As a result, Ethiopia has become known by the recurring droughts and the resulting famine that have occurred in 1888 – 1892 and more recently, in 1984 – 1985 (Pankhurst, 1966; Mehari and Vahlquist, 1976). There are recurrent periods of drought every 3-5 years in the northern parts of Ethiopia and every 6-8 years over the whole country (Haile, 1988). Part of the variability of seasonal and annual rainfall across time and space in the drought prone tropical African regions is known to be associated with the El Nino-Southern Oscillation (ENSO) phenomenon (Quinn, 1993; Seleshi and Demaree, 1995; Camberrlin, 1997; Camberlin, et al., 2001; Comenetz and Caviedes, 2002). The main wet season droughts in Ethiopia, for example, are more likely to occur during the years of warm ENSO events (Seleshi and Demaree, 1995; Seleshi and Zanke, 2004). The El Nino phase of the cycle is reported to relate to a low index phase of the Southern Oscillation and to be associated with an eastern and northern Australian drought, an east monsoon drought over Indonesia, deficient monsoon rainfall over India, and deficient summer monsoon rainfall over the highlands of Ethiopia (Quinn, 1993; Camberlin, 1997). El Nino has been related to a decrease in discharge of the Nile River, which mainly originates from the Ethiopian highlands (Griffiths, 1972; Quinn, 1993).

Extreme spatial and temporal variation in rainfall is also characteristic for the Tigray region, located in the northern tip of the country (Fig. 1.2). Realising the problems

and the potentials, agricultural development through irrigation has been a priority for the Ethiopian Government for the last two decades.



* Only two year data was available for Axum station.

Fig. 1.2 Monthly rainfall for selected meteorological stations around the study area (Tigray, North Ethiopia) for a period of 4 years (2003-2006), showing variation among years and sites [Source: MU - IUC, 2007: Digital database of climatological and stream flow data of Geba catchment, obtained from National Meteorological Services Agency and Ministry of Water Resources. VLIR (Belgium) - Mekelle University Institutional University Cooperation Programme, Mekelle (Ethiopia) and Leuven (Belgium)].

To achieve their goal, the Regional Government of Tigray established the 'Commission for Sustainable Agriculture and Environmental Rehabilitation for the Tigray Region (Co-SAERT)' in 1994. The commission's target was to achieve food self-sufficiency to the area mainly by development of irrigated agriculture through the planning, design and construction of about 500 dams within ten years (SAERT, 1994, Sustainable agriculture and environmental rehabilitation in Tigray, unpublished), and accomplished only about 14%.

Reservoirs provide various services to the local people, such as the production of electricity, an increased availability of water for irrigation and drinking purpose, and/or benefits with respect to flood protection, navigation, fisheries, the attraction of tourists, etc. (Zwahlen, 2003). Ethiopia has many small, medium and large reservoir dams constructed for hydropower generation, irrigation and drinking water supply or a combination of them. Koka reservoir, constructed in the 1950's on the Awash River in central Ethiopia is used for commercial fisheries (Tudorance *et al.*, 1999) in addition to its function for hydropower generation and irrigation water supply. According to Frenken (2005), small dams are defined to be less than 15 meter high and to have a capacity of less than 3 million m^3 . The height of the medium and large dams in Ethiopia is 15 - 50 meters and their capacity ranges from 4 to 19000 million m^3 . Currently several (seven) medium and large reservoirs are under construction for hydropower generation, irrigation and/or water supply in Ethiopia.

Multi-purpose small dams (micro-dams) constructed for irrigation supply are concentrated in the Amhara and Tigray regional states. The micro-reservoirs are intended to be permanent bodies of standing water between five and 50 hectares in size, established by the government and used by villages near the reservoirs. In Tigray regional state, there are more than 70 reservoirs (see fig. 1.3) and their ages vary between five and 20 years (Hurni, 1993; De Wit, 2003; Nigussie *et al.*, 2006; Tsehaye *et al.* 2007).



Fig. 1.3 Map of Tigray with distribution of reservoirs and lakes (Map from the Geographical & Statistical Information Office, Regional National State of Tigray, Ethiopia)

In addition to the role of these reservoirs in boosting agricultural development and preventing erosion, they have also potential to create added value for local residents, local nature value and biodiversity: 1) the water from the reservoirs can directly be used for household purposes, 2) some reservoirs hold the promise of enabling the culture of fish as a source of proteins and 3) they can also function as sites attracting bird life, amphibians, aquatic invertebrates and aquatic plants. In addition to its intrinsic value, biodiversity is an important indicator of ecosystem health and sustainability (Wilson, 1988) and shows important relationships with ecosystem functioning and ecosystem services (Myers, 1996; Holmlund & Hammer, 1999; Moss, 2000; Tilman, 1999; Luck *et al.*, 2003; Worm et al., 2006).

1.2. Reservoirs and cyanobacterial blooms in semi-arid regions

Reservoirs are man-made lakes. The requirement to store water to supply a wide range of human needs, and the operational procedures to fulfil that demand, may influence their ecosystem functioning and make them different from natural lakes (Straskraba & Tundisi, 1999). Reservoirs frequently differ from natural lakes in the shape of their longitudinal profiles. Natural lakes are normally deepest near the middle; reservoirs are almost always deepest just upstream from the dam (Baxter, 1977). As a result, reservoirs have been referred to as "half-lakes". Reservoirs often have a limited diversity because they are young and artificial systems, and because they show less development of vegetation along the littoral zone because of artificial changes in water level. Some authors consider reservoirs to occupy an intermediate position between rivers and natural lakes by combining numerous features of these two types of ecosystems (Kimmel *et al.*, 1990). The extent of shoreline modification in a reservoir is likely to be greater than in comparable natural lakes because the annual drawdown exposes a large area to the effects of shoreline development (Baxter, 1977).

One key characteristic of reservoirs is that they experience pronounced changes in water level. The water level consistently rises during the rainy season(s), and goes down when the water is used without being refilled during the dry season. Also year-to-year variations can be very pronounced in the African semi-arid sub/tropics as a result of the large inter-annual variability in rainfall, river flow and lake level (Dagnachew *et al.*, 2002). The seasonal and inter-annual fluctuations of water level are significantly larger in shallow lakes and reservoirs (Vallet-Coulomb *et al.*, 2001). Different authors associate eutrophication and development of blooms of specific phytoplankton species with water level variations of reservoirs (Harris & Baxter, 1996; Bouvy *et al.*, 1999, 2000). Dominance of a particular species within a phytoplankton community depends upon a

complex interplay of factors that include physical (e.g. retention time), chemical (e.g. nutrient loading) and biological (e.g. grazing) processes. Cyanobacteria are commonly encountered in many eutrophic reservoirs, and are known to contribute by their blooms to the deterioration in the aquatic environment (Chorus and Bartram, 1999). Harris & Baxter (1996) reported that cyanobacterial blooms are linked to a decrease in total water capacity in a subtropical reservoir.

Because of massive erosion linked to land degradation (Nyssen *et al.*, 2005), there is an excessive nutrient loading associated with sediment influx (Nigussie *et al.* 2006) to the reservoirs in Tigray. Many reservoirs are therefore expected to be characterized by high nutrient loads and phytoplankton blooms, including cyanobacteria blooms. This has indeed been observed in a field survey of reservoirs (Tsehaye *et al.* 2007; Dejenie *et al.* 2008).

Phytoplankton blooms are transient departures from quasi-equilibrium when the primary productivity temporally exceeds the losses and transports and the population grows rapidly and reaches exceptionally high biomass (Paerl, 1988). Typically, only one or a small number of phytoplankton species are involved, and some blooms may be recognized by discoloration of the water resulting from the high density of pigmented cells. Phytoplankton blooms usually are not single discrete events but rather are a series of fluctuations in which the biomass and the species composition of the phytoplankton population change rapidly. Blooms can be very generally classified into three types (Cloern, 1996): 1) recurrent seasonal events that usually persist over periods of weeks, 2) aperiodic events that often persist for periods of days and 3) exceptional events that are typically dominated by a few species and persist for months. Cyanobacterial blooms in the tropical regions fall into the persistent bloom type (3) (Sommer et al., 1986, Zohary & Robarts, 1989, Sivenon & Jones, 1999, Akin-Oriola, 2003.

Although there is no officially recognized threshold level, algae can be considered to be blooming at concentrations of hundreds to thousands of cells per millilitre, depending on the severity. Algal bloom concentrations may reach millions of cells per millilitre. Algal blooms are often green, but they can also be yellow-brown or red, depending on the species of algae. Algal blooms are monitored using biomass measurements coupled with the examination of species present. A widely-used measure of algal and cyanobacterial biomass is the chlorophyll concentration. Peak values of chlorophyll *a* for an oligotrophic lake are about 1-10 μ g/l, while in a eutrophic lake they can reach 300 μ g/l. In cases of hypertrophy, maxima of chlorophyll a can be as high as 3,000 μ g/l. (Zohary and Roberts, 1990).

The considerations of cyanobacterial bloom threshold have been different in different studies. In marine systems, cyanobacterial blooms have been defined as surface accumulations of cyanobacterial filaments by analysing satellite images (e.g. Kahru et al., 1994) and the intensity of bloom is characterised by spatial coverage. When microscopic analysis is conducted the biomass of species or group of species is used. Bloom intensity can be described as the maximum biomass observed. Wasmund (1997) took the biomass value 0.2 mg Γ^{-1} as a threshold in his analysis as this "differences from 'normal' levels of species abundance" and because at that concentration aggregated cyanobacteria became visible in the water and appeared as a "bloom" to the beholder. Lips (2005) characterized the intensity of cyanobacterial blooms by integrated biomass values of *Nodularia*, *Aphanizomenon* and Anabaena over a bloom period defined bloom period as the period when cyanobacterial biomass exceeds 0.5 mg Γ^{-1} at one or more sampling points in the Baltic Sea. In this study we defined blooms of *Microcystis* when *Microcystis* accounts for 65% or above of the total phytoplankton community. In doing so, we look at the total intensity of bloom across the water column rather than surface accumulations.

1.3. Cyanobacteria and cyanotoxins

Cyanobacteria or blue green algae are among the oldest and most ubiquitous oxygenic photosynthetic organisms (Falconer 1989; Carmichael, 1994; Chorus & Bartram, 1999; Castenholz, 2001). Their habitats range from hot springs to temporarily frozen ponds in Antarctica (Whitton, 1992). They are common members of the plankton of marine, brackish and freshwaters throughout the world. They also occur on rocks and soils and in symbiosis with plants and animals. Cyanobacteria possess characteristics similar to both bacteria and algae. They are similar to algae in size; they contain blue-green and green pigments and can perform photosynthesis and produce oxygen (Castenholz, 2001). Like eubacteria they are prokaryotes, and lack a nucleus and other membrane bound organelles.

Cyanobacteria are morphologically diverse. Unicellular and filamentous forms are commonly found, with both morphotypes able to form large colonies and bundles of filaments (fig. 2) (Whitton and Potts, 2000). Cells and filaments in colonies may be arranged in different ways, e. g. radially, in strict planes, or irregularly. Filaments can be branching, coiled, or straight. Cells growing in colonies may be packed in a mucilaginous sheath (e.g. *Microcystis* species) or in the case of filamentous species grow as floating mats (e.g. *Aphanizomenon*) or as free-floating strands (e.g. *Oscillatoria*). Further differentiation amongst the cyanobacteria includes the ability of certain filamentous genera, such as *Anabaena, Aphanizomenon, Gloeotrichia, Planktothrix,* and *Nodularia,* to enzymatically fix atmospheric nitrogen (called biological nitrogen fixation) in specialized cells called heterocysts. Several of the filamentous genera also produce other differentiated cells called akinets (spore stages) which are useful for survival in periods of harsh conditions like cold and drought.

Blooms of cyanobacteria are increasingly frequent in aquatic ecosystems around the world as a result of eutrophication (Chorus & Bartram 1999, FRW, 2000; Huisman et al.

2005). Cyanobacteria have evolved many adaptations to survive, compete and achieve dominance in freshwater environments. Several explanations have been forwarded for the formation of cyanobacterial blooms in water bodies. These include: elevated temperature, nutrient enrichment, low N/P ratios, low light energy requirements, high pH and/or low carbon dioxide concentration, selective zooplankton grazing, excretion of compounds that suppress the growth of competing algae, and the capacity to float and form scums (Porter, 1973; Shapiro, 1973, 1984, 1990; Murphy et al., 1976; Keating, 1978; Flett et al., 1980; Tilman et al., 1982, 1986; Smith, 1983, 1986; Haney, 1987; McQueen & Lean, 1987; Varis, 1993; Chorus & Bartram, 1999; Huisman et al., 2005). Shapiro (1990) suggested that a low carbon dioxide concentration and associated high pH is the key important factor for cyanobacteria dominance, the other factors (high temperature, low available light, lower N/P ratios, buoyancy, and low grazing losses) being related to CO₂ in some aspects. However, all factors may play a role in determining cyanobacteria dominance, and taking one factor in isolation may not help to understand the complex phenomenon of cyanobacterial bloom formation. Although the initiation of the blue-green maximum does not depend upon conditions of low CO₂ concentration or high pH (Shapiro, 1997), once they become abundant they may ensure their dominance by reducing concentrations of CO₂ to levels available only to themselves. The large growths and accumulations of cyanobacteria are often aesthetically undesirable because they discolour the water and cause turbidity. In addition, cyanobacteria are able to synthesize a number of low molecular weight compounds that cause taste and odour problems. These substances, including geosmin and methyl isoborneol often result in complaints regarding recreational and other water bodies and the quality of raw and treated drinking water. Production of low molecular weight compounds with high toxicity to vertebrates, including mammals by some taxa of cyanobacteria causes more concern. A number of common cyanobacteria can

produce a wide range of toxins in water, called cyanotoxins, causing a potential risk to livestock and human health (Chorus & Bartram 1999). The cyanotoxins are classified functionally into hepato-, neuro-, and cytotoxins. In addition to the cyanotoxins, cyanobacteria also produce lipopolysaccharides (LPS) as well as secondary metabolites that are potentially pharmacologically useful.



Fig. 1.4 Photograph of a cyanobacterial bloom in one of the study reservoirs (Mai Nigus) (a) and microphotographs of cyanobacterial colonies and filaments showing some of their morphological diversity (b-g). (b-c) colonies of *Microcystis sp.* (d) colonies of *Merismopedium* and filaments of *Anabaenopsis. (d)* coiled trichome of *Anabaena sp. (e) straight* trichome of *Anabaena sp.* and (f) trichome of *Aphanizomenon*. Heterocystis (h) and akinets (a) are indicated by arrows. Photos a and d are taken by Tsehaye Asmelash and photos b, c, a, d, e-g are taken by Jeroen Van Wichelen.

Defined by their chemical structure, cyanotoxins fall into three groups: cyclic peptides (the hepatotoxins microcystins and nodularin), alkaloids (the neurotoxins anatoxin and saxitoxins), and lipopolysaccharides (LPS). Of the cyanobacterial genera that include toxin forming species, the ones of particular concern when mass populations occur include

Microcystis, *Anabaena*, *Planktothrix* (formerly know as *Oscillatoria*), *Aphanizomenon*, *Cylindrospermopsis*, *Pharmidium*, *Nostoc*, *Anabaenopsis* and *Nodularia* (NHMRC 1994, Chorus and Bartram, 1999; FWR, 2004). These genera of cyanobacteria can produce a wide range of cyanobacterial toxins that vary in structure, toxicity and mode of action. Examples of toxic mass populations of cyanobacteria and their toxic effects on animal and/or humans have been reported in fresh, brackish and marine waters on a global basis (see Table 1.1 and 1.2)

Cyanobacterial toxins are generally divided into categories based on their modes of action in mammalian test systems (Codd, 1995; Sivonen and Jones, 1999) as hepatotoxins and neurotoxins. Although the most studied cyanobacterial toxins are the hepatotoxins and neurotoxins, cyanobacterial endotoxins (LPS) have also been implicated in human illness.

Location	Flamingo species	Numbers dying	Candidate toxigenic cyano-	Presence cyanobact toxins ^c in:	of terial	Reference
			bacteria ^b	Blooms, mats, scum	Bird tissues, excreta	-
Zoo pond, Orlando, Florida, USA	Chilean (P. chilen- sis)	10 (adults)	Μ	MC	MC	Chittick et al. (2002)
Wetland lagoon, Doñana National Park, Spain	Greater (P. ruber)	579 (out of 943 chicks)	M, Ana, Osc	МС	МС	Alonso- Andicober ry et al. (2002)
Lakes Nakuru and Bogoria, Kenya	Lesser (P. minor)	Thous- ands ^a	Ana, An, Osc, Ph, (others ?)	MC, ANA	MC, ANA	Ballot et al. (2002) Krienitz et al. (2003)

Table 1.1 Mass mortalities of flamingo species (*Phoenocopterus*) after exposure to ingestion of cyanobacterial biomass (from Codd et al., 2005)

^a Tens of thousands, 1991-1999 (see Krienitz et al., 2003); ^b M, Microcystis; Ana, Anabaena; Osc, Oscillatoria; An, Anabaenopsis; Ph, Phormidium; ^c MC, microcystins; ANA, anatoxins.

Table	1.2	Examples	of	human	exposures	to	cyanobacterial	blooms	and	toxins,	with
associa	ated	health outco	ome	s (from	Codd et al.	, 20	05)				

Year	Location (source ^a)	Cyano- bacteria ^b	Toxins	Health outcomes ^d	References
Drinking wa	ter				
1. 1975	USA (DWR)	S, L, Ph	?	GI	Lippy and Erb (1976), Keleti (1981)
2. 1979	Australia (DWR)	С	CYN	GI, LD, KD, ID	Byth (1980), Griffiths et al. (1998)
3. 1981	Australia (DWR)	М	MC	LD	Falconer et al. (1983)
4. 1972 – ca. 1990	China (SWR)	М	MC	PLC, D	Yu (1989, 1995)
5. 1988	Brazil (DWR)	M, Ana	?	GI, D	Teixeira et al. (1993)
6. 1994	Sweden (DWR, CR)	Pl	MC	GI, F, AP, MP	Annadotter et al. (2001)
Recreational 7. 1989	/occupational w UK (FR, SW, KY)	ater contact M	MC	GI, ST, BM, V, AP, F, PC	Turner et al. (1990)
8. 1995	Australia (FR, SW, BA)	M, Ana, Aph, Nod	Hepato -toxins	GI, FLS, BM, F, EE	Pilotto et al. (1997)
9. 1996	UK (FR, BO)	Pl	МС	R, F	G. A. Codd & K. A. Beattie, Environment Agency report
10. 1996- 1998	Australia (MR, SW, BA, FI)	L	?	CD, EE, RI	Dennison et al. (1999)
Haemo-dialys	sis				
1. 1974	USA (DC)	present	LPS	F, MY, C, V	Hindman et al. (1975)
12. 1996	Brazil (DC)	present	MC, CYN	VID, TIN, N, V, LD, D	Jochimsen et al. (1998), Pouria et al. (1998), Carmichael et al (2001), Azevedo et al. (2002)
13. 2001	Brazil (DC)	Ana, Mic	MC	Same as in 12 (no D reported)	Carmichael et al (2002)

bathing; BO, boating; MR, marine; FI, fishing; DC, dialysis clinic water.
b S, Schizothrix; L, Lyngbya; Ph, Phormidium; C, Cylindrospermopsis; M, Microcystis; Ana, Anabaena; Pl, Planktothrix; Aph, Aphanizomenon; Nod, Nodularia.
⁶ CYN, cylindrospermopsin; MC, microcystin(s); LPS, lipopolysaccharide endotoxin(s).
^d D, deaths; GI, gastroenteritis; LD, liver damage; KD, kidney damage; ID, intestinal damage; PLC, primary liver cancer; F, fevers; AP, abdominal pain; MP, muscular pain; ST, sore throat; BM, blistered mouth; V, vomiting; PC, pulmonary consolidation; FLS, flu-like symptoms; EE, eye and/or ear irritation; R, rashes; CD, contact dermatitis; RI, respiratory irritation; MY, myalgia: C, chills; VID, visual disturbance: TIN, tinitus: N, nausea.

1.3.1. Hepatotoxins

The hepatotoxins are low molecular weight cyclic peptide toxins. These are the toxins that have been most frequently linked to animal intoxication. Cyanobacterial hepatotoxines are reported the most common toxins worldwide and can cause gastro-enteritis, liver-damage and hepatocellular carcinoma's (MacKintosh *et al.*, 1990; Repavich *et al.*, 1990; Ohtani *et al.*, 1992; Elder *et al.*, 1993; Carmichael 1994; Chorus and Bartram, 1999; Humpage & Falconer, 1999). Some of the hepatotoxins (e.g. cylindrospermopsin) may also cause kidney damage in addition to liver damage. Based on their chemical structure, hepatotoxins can be divided into three categories: microcystins, nodularins and cylindrospermopsins.

Microcystins have been described from the genera *Microcystis, Anabaena, Planktothrix, Nostoc,* and *Anabaenopsis* (Botes et al., 1982; Carmichael et al., 1988a; Meriluoto et al., 1989; Namikoshi et al., 1990; Namikoshi et al., 1992). They are cyclic molecules (Fig. 1.5) that contain unusual amino acids and are synthesized nonribosomally via peptide synthetases (Meissneret al., 1996; Dittmann et al., 1997). Their molecular weight ranges from 800 to 1,000 Da (Botes et al., 1982a; 1982b).

Microcystin is a 7 amino acid ring, of which two, N-methyl-dehydroalanine (Mdha) and a 3-amino-9-methoxy-2, 6, 8-trimethyl-10-phenyldeca-4, 6-dienoic acid (ADDA), are almost unique to microcystin (Fig. 1.5). The structure of microcystin was first described from an isolate of *M. aeruginosa*. Meanwhile, more than 70 structural variants of this toxin are known (Botes et al., 1982a; 1982b, 1985; Rinehart et al., 1994; Sivonen, 1996) (which differ in their L amino acid components and the presence of certain methyl groups.) The current nomenclature names the most common structural variants, for instance microcystin LR (L, L-leucine; R, L-arginine) or microcystin-LW (W, L-tryptophane) (Carmichael et al., 1988b). Microcystin LR is the toxin most frequently associated with documented cases of animal poisoning (Eriksson *et al.*, 1990; MacKintosh *et al.*, 1990; De Mott & Dhawale,

1995; Zurawell *et al.*, 2005). Nodularins are structurally similar to the microcystins (Fig. 1.5). They are cyclic peptides that also contain ADDA, but with only 5 amino acids in total. Unlike microcystins that are produced by a number of freshwater cyanobacterial genera, nodularins have only been documented in cyanobacterial blooms and strains of *Nodularia*, a bloom forming cyanobacteria typical of brackish and marine waters.

Cylindrospermopsins (Fig. 1.5) are alkaloids produced by *Cylndrospermopsis raciborskii, Umezkia natans* and *Aphanizomenon ovalisporum* (Sivenon & Jones, 1999). This toxin induces pathological changes in the liver but also in the kidneys, spleen, thymus and heart (Hawkins et al., 1985, 1997).

Cyanotoxins are usually found inside the cell, and are degraded very slowly after cell death (Sivonen, 1990; Rapala et al., 1997; Orr and Jones; 1988). Even cyanobacterial scums that accumulate and dry out on lake shores may retain high levels of microcystins for a few months (Jones et al., 1995). Animal intoxication may occur either by direct ingestion of living cells or by drinking contaminated water after the bloom collapses and toxins are released from the cells. Hepatotoxins are absorbed across the ileum and are then transported to the liver and taken up by hepatocytes. These toxins are potent inhibitors of protein phosphatases type 1 and 2A, enzymes that are crucial to cell growth and tumour suppression (MacKintosh et al., 1990; Matsushima et al., 1990; Yoshizawa et al., 1990; Runneger et al., 1995). Acute poisoning by hepatotoxins causes weakness, anorexia, pallor of the mucous membranes, vomiting, cold body extremities and diarrhoea (Carmicahel, 1992). Death due to intrahepatic haemorrhage and hypovolaemic shock may occur within a few hours. The LD₅₀ for microcystin LR in mice is 50-160 μ g Kg⁻¹ body weight (bw) by intraperitoneal injection and about 5000 μ g Kg⁻¹ bw by oral route. Humans and animal mortalities due to cyanobacterial hepatotoxins have been reported from many countries including Africa (Joshimsen *et al.*, 1998; Ginkel, 2004; Jayatissa, *et al.* 2005; Kreintz *et al.* 2003).



Fig.1.5. Structure of the toxic cyclic peptides (microcystin, nodularin, and cylindrospermopsin) and alkaloids (anatoxin-a, saxitoxins, and anatoxin-a(s) produced by cyanobacteria

1.3.2. Neurotoxins

There are three known cyanobacterial neurotoxins (fig. 1.5): anatoxin-a, anatoxin-a(s), and paralytic shellfish poisons (PSPs, also known as saxitoxins) (Carmichael, 1994; Sivonen & Jones, 1999).

Cyanobacterial neurotoxins disrupt the normal propagation of neural impulses to muscles, causing paralysis and death via respiratory failure and asphyxia in animals (Carmichael, 1994). Anatoxin-a is a low molecular-weight secondary amine that mimics acetylcholine. It binds to the nicotinic-acetylcholine receptors with higher affinity than acetylcholine, and is not susceptible to hydrolysis by acetylcholinesterase. It induces muscle twitching and cramping, followed by fatigue and paralysis (Gorham & Carmichael, 1988a; Carmichael, 1994). Anatoxin-a(s) is an organophosphate that also causes muscle fatigue and failure, yet through a different mechanism. It binds to acetylcholinesterase, rendering it incapable of breaking down acetylcholine, which results in overstimulation of the muscle cells (Carmichael, 1994). Saxitoxins inhibit nerve impulse propagation along axons by blocking sodium ion entry into nerve cells through sodium channels, effectively suppressing stimulation of muscles, including those of the respiratory system (Adelman et al., 1982; Sivonen & Jones, 1999).

Anatoxin-a has been described in members of Anabaena, Planktothrix Oscillatoria, Aphanizomenon, Cylindrospermum (Devlin et al., 1977; Carmichael et al., 1990; Sivonen et al. 1989; Edwards et al., 1992; Skulberg et al. 1992; Mahmood & Carmichael 1986a,), and in lower concentrations in Microcystis (Carmichael et al., 1975; Park et al, 1993; Sivonen and Jones, 1999). Anatoxin a(s) is the only naturally occurring organophosphate and has first been isolated from Aanbaena flos-aquae and A. lemmermannii (Matsunaga et al., 1989; Mahmood & Carmichael, 1987; Henriksen et al., 1997; Onodera et al., 1997). Saxitoxins are well known from marine dinoflagellates (red tide) where they are responsible for paralytic shellfish poisoning after consumption of contaminated shellfish. However, they have also been detected in relevant amounts in freshwater cyanobacteria such as Aphanizomenon flos-aquae, A. circinalis, Cylindrospermopsis raciborskii, and Lyngbya wollei (Mahmood and Carmichael, 1986; Humpage et al. 1994; Negri and Jones, 1995; Lagos et al., 1999; Kaas & Henriksen, 2000; Pereira et al., 2000; Pomati et al., 2000; Ferreira et al., 2001; Li et al., 2003). Anatoxin-a, anatoxin-a(s), and saxitoxins are all lethal to animals, with LD50 based on intraperitoneal (ip) injection of mice of 200, 20, and 10 μg/kg, respectively (Carmichael, 1992; Monserrat et al., 2001).

1.3.3. Lipopolysaccharides (LPS)

Lipopolysaccharides, also called endotoxins, are structural components of the outer layer of the cell envelope of all gram negative bacteria. LPS is heat stable and toxic to mammals, with its toxicity varying among different bacterial and cyanobacterial isolates. The LPS molecule consists of three main parts: the O antigen, a core polysaccharide and lipid A. The lipid A is responsible for the biological effects and responses.

Unlike LPSs of most bacteria, cyanobacterial LPS lack phosphate in the lipid A (Keleti and Sykora, 1982). In animal experiments, cyanobacterial LPS is about 10 times less toxic than enterobacterial LPS (Codd, 1984). There are conflicting results on the effect of the endotoxins on mice. Some studies report lethal effects upon injection of the toxin (Keleti et al., 1979; Keleti and Sykora, 1982), while others report no effect (Weise et al., 1970; Keleti et al., 1979).

1.4. The aquatic food web

Knowledge on how species interact within ecosystems is necessary to understand how natural and anthropogenic pressures will affect ecosystem structure and functioning (Lampert & Sommer, 1997, Moss, 1998, 2000; Scheffer, 1998, Begon et al., 2006). We need information on abiotic environmental conditions as well as on the structure of the food web in the system to understand the processes and factors that cause algal blooms in reservoirs, for example. Below, we provide a brief overview of the aquatic food web structure in lakes and reservoirs and the interaction between the different food web components.

1.4.1. Planktonic food webs in standing freshwater systems

The classic concept of food webs states that energy and nutrients flow up through well defined trophic levels (Krebs, 1994; Begon *et al.*, 1996). In lakes, phytoplankton cells at

the bottom of the food web fix carbon; hence they are primary producers that support the upper trophic levels of secondary producers. The principal grazers, such as copepods and cladocerans, feed on phytoplankton and are in turn consumed by larger predators, such as insects and fish (Fig. 1.6). In many lakes secondary carnivores, including piscivorous fish, are at the top of the food web (Valiela, 1991).

The efficiency of the transfer of energy from one trophic level to the next is low. In most cases only 10% or less of the energy flowing into one trophic level is transferred to the next (Begon *et al.*, 1996). As a result, a limited number of trophic steps can be sustained in open water food webs. So much energy is dissipated by respiration of organisms at each trophic level that by the fifth trophic step, virtually no energy remains to be passed on to an additional trophic level (Begon *et al.*, 1996). In this classic view of food webs, bacteria are assumed to have only the role of degrading the dead organic matter that was not consumed. Bacteria (and fungi) are indeed the principal remineralizers of the nutrients that are sequestered in the organic matter synthesized by primary and secondary producers. These remineralized nutrients, plus externally supplied nutrients, keep the system going (Valiela, 1991).

The traditional view of the role of bacteria in plankton communities as remineralizers of nutrients became modified during the 1970s and 1980s. New techniques revealed that heterotrophic bacteria in the plankton were much more abundant and productive than previously thought. Moreover, several studies showed that in marine environments there was an alternative pathway of carbon flow that led from bacteria to protozoa to metazoa, with dissolved organic matter (DOM) being utilized as substrate by the bacteria (Steele, 1974; Pomeroy, 1974). This food web paradigm was called the 'microbial loop' (Azam *et al.*, 1983). The microbial loop was first defined as a separate 'loop' in which bacteria utilize dissolved organic matter and are consumed by protozoa,

with energy and nutrients thereby returning to the phytoplankton-zooplankton-fish food chain (Azam *et al.*, 1983). However, the current view of the microbial food webs is one of a complex interacting community, including phytoplankton, bacteria, and protozoans, which collectively account for primary carbon fixation, nutrient regeneration and production to support metazoans (Sherr & Sherr, 1991). In addition to bacterial uptake of nonliving organic matter, many direct links exist among algae, bacteria, and heterotrophic protozoa. By these numerous pathways, fixed organic carbon can be recovered, often into larger-sized microorganisms that may be more available for consumption by metazoans (Sherr & Sherr, 1988). Metazoans consume organisms of the microbial food web by grazing heterotrophic flagellates and ciliates (Wetzel, 2001).



Figure 1.6 Schematic models of the different compartments occurring in pelagic food webs. Arrows represent the flux of matter and energy from one compartment to another. The left half transfers represent the classical food web, whereas the right transfers represent the microbial food web (Modified from Azam, 1983).

The importance of bacterial production and microplanktonic consumers to macrobial food webs is still poorly resolved, especially in freshwater environments (Neill,

1994). Many authors seem to agree that the high metabolic rates of bacteria dissipate almost all microbial production prior to consumption by macro-organisms, so that bacteria and their consumers serve mostly to remineralize inorganic nutrients subsequently used by primary producers (Currie, 1990). In this view, bacterial and protist biomasses may provide emergency rations or casual by-catch for suspension feeders, except under "unusual" conditions, such as in ecosystems driven by the input of external organic carbon. The opposing perspective suggests that major uncertainties still remain, particularly at small spatial and temporal scales (e.g. hotspots of production, mutualistic facilitation) and that productive freshwaters may be especially conductive to high fluxes of microbial production to higher trophic levels, despite apparent low efficiency (Neill, 1994).

1.4.2. Bottom-up, top-down and the trophic cascade in lakes

Plankton communities are structured by the simultaneous impact of bottom-up and topdown effects (Rothhaupt 2000, Begon et al., 2006). Top-down control refers to stimulations where the structure (abundance, distribution, and/or diversity) of lower trophic levels depends directly or indirectly on trophic activities of the higher trophic levels. And bottomup control refers to direct or indirect dependence of community structure on factors producing variation at lower trophic levels or their resources. Discriminating between these two types of driving forces is important, as it determines which measures may or may not be successful when one wants to control biomass or species composition at a given trophic level. For example, if bottom-up factors are the main determinants of phytoplankton biomass, then the occurrence of phytoplankton blooms can only be controlled by reducing nutrient levels (e.g. Schindler, 1977; Wetzel, 2001). On the other hand, zooplankton grazing may control both algal biomass and species composition (top-down) (Timms & Moss, 1984, Carpenter et al., 1985, Carpenter & Kitchell, 1993, Scheffer et al., 1993). In this situation, it may be possible to reduce the occurrence of algal blooms by changing zooplankton community structure through a manipulation of predation pressure on zooplankton by fish (Shapiro & Wright, 1984; Gulati, 1990; Scheffer et al., 1993; Scheffer, 1998, 1999).

The bottom-up hypothesis states that population densities and community structure of a given trophic level is determined by the availability of resources. In pelagic food webs, the importance of nutrients such as phosphorus and nitrogen for phytoplankton primary production is well-known (Schindler, 1977, Wetzel, 2001). They control the potential flux of carbon and energy that can be transferred to higher trophic levels. Several studies have shown a positive correlation between phosphorus concentrations and phytoplankton production and biomass (Currie 1990). However, residual variance in such relationships remains high. In the intermediate to high phosphorus range, phosphorus availability tends to mainly determine the maximal potential phytoplankton biomass, whereas the actual phytoplankton biomass is often much lower (Carpenter *et al.* 1985). This indicates that phytoplankton biomass cannot be explained solely by nutrient availability.

The top-down hypothesis postulates that population density and structure of the community at a given trophic level are controlled by the one-higher trophic level (Carpenter et al. 1985). In pelagic food webs, piscivorous fish control the densities of planktivorous fish and planktivorous fish control zooplankton densities. Grazing by zooplankton in turn will control phytoplankton density and community composition (Elser 1992).

The trophic cascade hypothesis (Carpenter et al. 1985) integrates both the bottomup and top-down hypotheses. The top-down control of the highest trophic level on the second-highest one combined with the alternation of top-down / bottom-up control as one moves down along the food web is called a trophic cascade (Carpenter et al., 1985;
Rothhaupt, 2000). It argues that whereas the potential biomass and production of organisms at a given trophic level is controlled by resource availability, the actual biomass and dynamics of organisms at each trophic level is regulated by the adjacent higher trophic level via top-down control. Trophic cascades occur because predators in a food chain suppress the abundance of their prey, thereby releasing the next lower trophic level from predation (Carpenter et al. 1985, Carpenter & Kitchel 1993). Planktivorous fish can become very abundant when piscivorous fish are scarce, affecting the abundance and structure of zooplankton communities by selectively eliminating large-bodied zooplankton, leading to increased phytoplankton biomass (Shapiro & Wright 1984, Vanni 1986, Carpenter & Kitchell 1993, Lampert & Sommer, 1997, Vanni & Layne 1997).

1.4.3. Fish, zooplankton and macrophyte interactions in eutrophic tropical shallow lakes Shallow lakes are characterized by the absence of stratification and as a consequence an intense sediment-water interaction. As shown for temperate shallow lakes, the food web structure in these lakes can basically show two contrasting stable states, the so-called clearwater and turbid states, each stabilized by their own positive feedback mechanisms (Scheffer, 1998). A clear water state is largely characterized by submerged plants and sufficient piscivorous fish biomass to exert strong control on planktivorous fish enabling large zooplankton (e.g., Daphnia) to control phytoplankton (Jeppesen et al., 1997a, Moss et al., 1997, Scheffer et al., 1997a). Submerged plants promote clear water state through several stabilizing buffer mechanisms namely: nutrient uptake, reduction of sediment resuspension, provision of refuge (especially for *Daphnia* against fish predation), enhancement of denitrification, and allelopathic effects on phytoplankton (Timms & Moss, 1984, Gumbricht, 1993, Hamilton & Mitchell, 1996). In the turbid water state, there are

few or no submerged plants, and there is total dominance of the fish community by planktivorous fish (Jeppesen et al., 1997a, Moss et al., 1997, Scheffer, 1997a). And the predation by planktivorous fish leads to changes in the zooplankton communities towards smaller body size. Bentivorous fish also promote phytoplankton growth by recycling nutrients (Attayde and Hansson, 2001). States dominated by floating macrophytes and filamentous cyanobacteria (see below) are also possible however (Scheffer and Van Nes 2007). Floating plants have primacy in competition for light but need high nutrient concentrations (Portielje & Roijackers, 1995). By contrast, rooted submerged macrophytes are susceptible to shading, but less dependent on nutrients in the water column as they may take up a large part of their nutrients from the sediment (Chambers et al., 1989). Still submerged plants can also use their shoots effectively for nutrient uptake from the water column (Robach et al., 1996) and by various mechanisms reduce nitrogen concentrations in the water column to below detection levels (Van Donk et al., 1993). This interaction may result in two alternative stable states: a floating plant dominated state in which invasion by submerged plants is prevented by shading, and a situation dominated by submerged plants in which invasion by free-floating plants is prevented by reduced nutrient availability (Scheffer et al., 2003). An alternative stable state for the phytoplankton community in shallow lakes may also result from the shade tolerance of filamentous cyanobacteria. Dominance by Oscillatoriaceae can be an alternative stable state of the algal community of shallow lakes because these shade-tolerant cyanobacteria are able to cause an increase in turbidity that favours their competitive advantage (Scheffer et al., 1997b). The relative inedibility of filaments to zooplankton may further enhance the stability of blue-green dominance. It has been reported that high flush rates reduce the probability of blue-green dominance because of their relatively slow growth rates (Scheffer et al., 1997b). But in less

turbid conditions, other groups of phytoplankton such as the green algae have a competitive advantage and can become dominant over blue green algae.

During the past centuries, increased nutrient loading on many shallow lakes has resulted in major changes in the biological structure and dynamics of lakes and often in a shift from a clear-water to a turbid state world-wide (Jeppesen et al, 2005). Food webmanipulation at higher trophic levels (using planktivorous and piscivorous fish) has become widely used practices to reverse the shift from turbid to a clear-water state in the temperate region (Carpenter et al., 1987, Jeppesen et al., 2005, Hart, 2006). However, several factors indicate that fish stock manipulations would not have the same positive effect on the environmental state in tropical lakes as in temperate lakes (e.g. Jeppesen et al., 2005): 1) the fish species richness is often higher in tropical and subtropical lakes and many of the fish species show partial niche overlap, which increases predator control of prey items (Lazzaro, 1997, Aguiaro & Caramaschi, 1998), 2) the fish stock in tropical and subtropical lakes is often dominated by omnivorous species that feed on zooplankton but also consume phytoplankton, periphyton, benthic invertebrates, and detritus (Lazzaro, 1997). The subtropical and tropical fish stock that has the potential of feeding on zooplankton may thus attain a higher carrying capacity than obligate zooplanktivores, which augments the potential control of the zooplankton; 3) top-down control by piscivores is most likely weaker in subtropical/tropical lakes than in temperate lakes. Only few large strictly piscivores and small-sized carnivores occur and sit-and-wait predators are often more frequent (Quirós, 1998), 4) fish density, but not necessarily biomass, is substantially higher in subtropical and tropical lakes than in comparable north temperate lakes (Scasso et al. 2001, Meerhoff et al., 2003).

Due to high predation by fish, the zooplankton communities in tropical and subtropical lakes are frequently dominated by small cladocerans (like *Diaphanosoma*,

Ceriodaphnia and *Bosmina*) and rotifers, and by juveniles and small copepodites among the copepods (Dumont, 1994, Garcia et al., 2002). When fish are absent, large *Daphnia* spp. may sometimes develop (Mazzeo et al., 2003). Omnivorous copepods usually dominate in terms of biomass in oligo-mesotrophic systems, whereas microzooplankton prevails in more eutrophic systems. The high temperatures, the daily fluctuations in physical and chemical conditions or sudden environmental changes due to heavy rains may add to the predominance of fast-recovering forms such as protozoans and rotifers in the zooplankton community. The classic control of phytoplankton by large zooplankton in temperate lakes is therefore not usually found in tropical lakes.

All but the smallest zooplanktivorous fish have a strong preference for large-bodied zooplankton because they are easier to detect visually (Zaret, 1980, Lampert, 1987a). A reduction in fish density will result in a shift in body size of the zooplankton towards larger sized individuals (Brooks & Dodson, 1965, Shapiro & Wright 1984, Lampert, 1987a). This is both because large zooplankton are stronger competitors than small zooplankton (Gliwicz, 1990) but also because most invertebrate predators, who are the main predators in the absence of fish, have a strong preference for small zooplankton due to gape limitation or problems to handle larger prey (Dodson, 1974, Zaret, 1980, Lair, 1990). Large-bodied zooplankton (Lampert, 1987b). As a result, the shift towards larger-bodied zooplankton upon release from fish predation pressure may strongly increase the top-down impact of zooplankton on phytoplankton. This shift in size structure is therefore a key element of the control of phytoplankton biomass and structure by manipulating fish communities (Shapiro & Wright, 1984), a management approach that has been applied successfully in quite some temperate lakes (e.g. Kairesalo et al., 1999).

Aquatic plants play a very important structuring role in most freshwater ecosystems, and especially in shallow lakes where they can potentially cover the whole lake bottom surface (Scheffer et al., 1993). In temperate nutrient-rich lakes, submerged plants often act as daytime refuges for zooplankton against fish predators (Timms & Moss, 1984, Lauridsen et al., 1996, Burks et al., 2002). At night, when the risk of predation is lower zooplankton migrate to the open water for feeding and thereby contribute to maintaining clear-water conditions in lakes with high macrophyte coverage (Jeppesen et al., 1997b). However, in the tropics and subtropics, the effects of macrophytes on trophic interactions are more complex. The few studies conducted so far in the subtropics and tropics indicate that fish, particularly the smallest species and individuals, aggregate in high numbers in the vegetation (Conrow et al., 1990, Meerhoff et al., 2003, Branco et al., 2007).One might therefore expect that the vegetation is a poor refuge for large-bodied zooplankton in warm lakes (Bachmann et al., 2002, Meerhoff et al., 2003; Meerhoff, 2006), with as a consequence a less important structuring role of macrophytes in tropical than in temperate shallow lakes. On the other hand, macrophytes may perform better in tropical turbid lakes as they are not forced to regrow each spring from the bottom as in a temperate climate. Also, contrasting management approaches can alter the role of macrophytes in lake functioning and trophic interactions. While in temperate systems the introduction and development of aquatic plants are considered a key step in a restoration process (Moss et al., 1996), many aquatic plants in the tropics and subtropics are often considered as a nuisance and subject to severe eradication measures.

1.4.4. Zooplankton and cyanobacteria blooms

Cyanobacterial blooms are increasingly frequent in aquatic ecosystems around the world as a result of eutrophication (Chorus & Bartram, 1999; Huisman et al., 2005) and are likely to

increase in frequency with global warming. The study of the interactions between bloomforming cyanobacteria and zooplankton has become important because of the potentials for top-down control of cyanobacterial blooms by zooplankton (Burns, 1987, Lampert, 1987b; Sommer, 1989). A large body of research has been directed at understanding the mechanisms by which zooplankton affect cyanobacteria and vice versa.

Cyanobacteria are characterized by several adaptations that may make them less sensitive to zooplankton grazing than other phytoplankton groups. Many cyanobacterial species form colonies that are often too large to be ingested by zooplankton (Hawkins & Lampert, 1989, Gliwicz, 1990, Gliwicz & Lampert, 1990). In addition, several species produce toxins (Eriksson et al., 1990, MacKintosh et al., 1990, DeMott & Dhawale, 1995, Ferreira et al., 2001, Pereira et al. 2000, Plumley, 1997), the effect of which ranges from a reduction in feeding activity (Haney et al., 1994) to immediate death (Reinikainen et al., 1994). For example, the filamentous morphology of certain cyanobacteria has been shown to negatively affect large cladocerans more than small cladocerans through reduced fecundity (Webster and Peters, 1978, Gilbert, 1990). Two major mechanisms have been proposed for the greater susceptibility of large cladocerans to colonial or filamentous bluegreen algae (Webster & Peters 1978, Richman & Dodson, 1983, Porter & McDonough, 1984): 1) colonial or filamentous algae clog the filtering appendages of larger cladocerans, reducing their feeding rates on co-occurring nutritious food sources or increasing their respiration rates or both, 2) larger cladocerans ingest more readily colonial or filamentous blue-greens and are thereby more strongly affected by any toxic chemicals that these algae may possess. Finally, mucilaginous sheets may reduce digestibility, and many cyanobacteria are reported to be poor food in terms of biochemical composition (Porter & Orcutt, 1980, Lampert, 1987b; DeMott, 1989, Hawkins & Lampert, 1989; De Bernardi & Giussani, 1990). These characteristics suggest that cyanobacteria may be quite resistant to zooplankton grazing, and this has indeed been reported by many studies (Sarnelle, 1993; DeMott, 1999). Yet, several studies report that the development of cyanobacteria can be suppressed by zooplankton grazing (Haney, 1987, Matveev et al., 1994).

A number of studies suggest that a dominance of *Daphnia* in the zooplankton community results in a reduced likelihood of cyanobacterial blooms (MacKay & Elser, 1998, Smith, 1983). This may at first sight seem counter-intuitive, because it has been shown that especially large-bodied cladocerans such as *Daphnia* suffer from clogging of their filtration apparatus by cyanobacteria filaments (Gliwicz, 1990). Yet, *Daphnia* are also the most efficient grazers of phytoplankton, and one likely scenario may be that grazing by *Daphnia* may prevent cyanobacteria to develop dense populations and produce filaments (Gliwicz, 1990). If so, timing of the presence of *Daphnia* can be important, because efficient grazing could then prevent bloom formation but not suppress an existing bloom. According to Elser (1999), *Daphnia* also reduces the importance of cyanobacteria in the phytoplankton because of its high need for phosphorus, which results in a relatively higher N:P ratio in the medium (Andersen & Hessen, 1991, Elser & Urabe, 1999; Sterner, 1990). As such, *Daphnia* would promote dominance of green algae over cyanobacteria indirectly, through a bottom-up effect.

Cyanobacteria species such as *Microcystis* have been shown to be polymorphic with respect to the production of toxins, with non-toxic and toxic strains often coexisting and showing seasonal changes in relative abundance (Kirk and Gilbert, 1992, Barreiro et al., 2007). There is a need for studies that systematically investigate the relationship between zooplankton, and especially *Daphnia*, and the occurrence of cyanobacteria in the phytoplankton community, their tendency to bloom, and the degree to which these blooms are toxic. It is currently insufficiently known whether grazing of zooplankton actually reduces or enhances cyanobacteria blooms and to what extent this is dependent on species

composition of the zooplankton community. The importance of cyanobacteria in many tropical lakes, for instance, is attributed to a lack of top-down control of phytoplankton by large zooplankton, due to the dominance of omnivorous copepods in oligo-mesotrophic systems and of microzooplankton in eutrophic systems (Jeppesen et al., 2005). Especially with respect to toxicity, this is a burning question: will intensive grazing by zooplankton prevent or enhance toxic blooms? If toxin production is a defence against zooplankton, one may expect that dense zooplankton populations may enhance cyanobacteria blooms to become toxic. On the other hand, if *Daphnia* can prevent bloom formation, they may perhaps also prevent the cyanobacteria populations to become toxic. It is therefore possible that the outcome of this interaction depends on initial conditions. Insight in this matter may be crucial in our efforts to reduce the occurrence of cyanobacteria blooms.

1.5. Study Area

Tigray, the northernmost Regional State of Ethiopia, lies between 12° N and 15° N Latitudes and 37° 10' E and 40° 10' E Longitudes (Fig. 6). The Tigray Regional State has an area of 50,078 km², out of which 19 per cent is cultivable (CSA, 2000). The Tigray region is droughtprone and suffered from high land degraded.

The annual rainfall pattern in North Ethiopia is uni-modal with a short rainy season and a prolonged dry season (Fig. 1.2). Rains are erosive as well as unreliable (Lemma, 1996; Nyssen *et al.*, 2005). Some authors divide the year into three periods, with October to May being dry, the period from June to September as rainy but with moderately spaced heavy rains, and July and August as rainy with very intensive heavy rains (Daniel 1977; Nyssen et al., 2005). The climate is generally characterized as tropical semiarid (Virgo and Munro, 1978) with an annual rainfall range from 450 mm in the north, east and central zones to 980 mm in the southern and western parts of Tigray. In addition, within a given year, there is strong spatial variation in the amount of rainfall (Fig. 1.2).

1.5.1. General characteristics of the reservoirs in Tigray

Dejenie et al. (2008) studied the limnological characteristics of a set of 32 reservoirs in Tigray (see Table 2 for the list of reservoirs; see also Tsehaye et al. 2007 for a description). The reservoirs are found at altitudes of 1833 to 2747 m.a.s.l., with most of the reservoirs (31 /32) located at an altitude \geq 2000 m.a.s.l. The reservoirs vary in their abiotic characteristics. Generally they are small (1.8 to 45 ha. in surface area) and shallow. Depth varies among seasons, being reduced during the dry season, resulting from the use of water for irrigation (see Table 1.3 for a summary of the abiotic ecological characteristics of the reservoirs). A fraction of reservoirs dries up during the dry season. This fraction is higher than suggested in Table 1.3, as Dejenie et al. (2008) explicitly avoided sampling too many reservoirs that would fall dry during the dry season. The water in the reservoirs has a higher conductivity during the dry season due to evaporation. The reservoirs are alkaline, their pH ranging from 7.45 to 9.6. The trophic state of the reservoirs varies from mesotrophic to hypereutrophic. *Daphnia* is present in a majority of the reservoirs. Dejenie et al., (2008) identified a total of fifteen cladoceran species and the genus *Daphnia* was reported from all reservoirs as the most abundant.

The small riverine fish (*Garra sp.*) is the dominant fish in most reservoirs. Two introduced fish species, *Tilapia zillii* (Gervais, 1848) and *Oreochromis niloticus cancellatus* (Nichols, 1923), were reported from eight of the reservoirs (namely Dibla, Enda Gabriel, Haiba, Korir, Laelay Wukro, Mai Sessella, Shilenat IV and Tsinkanet).

Table 1.3 Geographic co-ordinates, altitude, surface area and depth of reservoirs (Dejenie

et	al.,	2008)	
		/	

Reservoir names	No.	Long.	Lat	Altitude	Area	Depth	Year of
			Lut.	(m)	(ha)	(m)	construction
Adi Amharay	1	39.57	13.41	2,354	9.3	2.5	1996/1997
Adi Asme'e	2	38.96	13.65	1,833	2.1	4.5	1993/1994
Adi Gela	3	39.51	13.13	2,044	17.0	4.3	1997/1998
Adi Kenafiz	4	39.41	13.25	2,161	11.6	2.2	1997/1998
Betequa	5	39.34	13.32	2,260	6.0	4.0	1996/1997
Bokoro	6	39.57	14.20	2,673	3.8	3.5	1985/1986
Dibdibo	7	39.08	14.26	2,015	14.8	8.8	1998/1999
Dibla	8	39.49	14.23	2,446	1.8	0.5	1986/1987
Dur Anbesa	9	39.44	13.27	2,133	11.1	5.0	2000/2001
Enda Gabriel	10	39.58	14.18	2,673	3.8	3.3	1985/1986
Era Quihila	11	39.60	13.45	2,321	6.4	2.5	1996/1997
Gereb Awso	12	39.56	13.43	2,283	2.4	3.0	1997/1998
Gereb Beati	13	39.48	13.45	2,151	14.0	9.5	1999/2000
Gereb Mihiz	14	39.47	13.29	2,123	17.7	4.9	1997/1998
Gum Selasa	15	39.54	13.24	2,116	23.1	1.5	1994/1995
Haiba	16	39.28	13.28	2,263	45.4	3.5	1997/1998
Hashenghe	17	39.67	13.48	2,400	5.4	1.0	1996/1997
Hizaeti Wedi Cheber	18	39.54	13.36	2,245	32.6	2.5	1996/1997
Korir	19	39.61	13.75	2,022	13.7	4.0	1995/1996
Laelay Wukro	20	39.61	13.81	2,023	9.8	4.5	1997/1998
Mai Delle	21	39.52	13.22	2,105	8.0	2.1	1997/1998
Mai Gassa I	22	39.49	13.29	2,131	14.7	3.9	1996/1997
Mai Gassa II	23	39.49	13.29	2,130	9.1	1.3	1996/1997
Mai Leba	24	39.23	13.69	2,231	8.2	3.0	1997/1998
Mai Nigus	25	38.66	14.12	2,056	30.8	11.0	1996/1997
Mai Sessella	26	39.03	14.07	2,068	21.8	9.0	1999/2000
Mai Seye	27	38.80	14.06	2,007	15.9	10.0	2000/2001
Meala	28	39.35	13.29	2,370	14.5	4.5	1997/1998
Ruba Feleg	29	39.73	13.95	2,747	10.7	11.0	1995/1996
Shilanat IV	30	39.49	13.10	2,080	17.7	9.5	1997/1998
Tegh'ane	31	39.73	13.89	2,735	9.6	6.5	1996/1997
Tsinkanet	32	39.54	14.01	2,320	7.0	4.1	1993/1994

	Wet season (2004)		Dry season (2005)			
Variable	Mean	Minimum	Maximum	Mean	Minimum	Maximum
Altitude and morphometrics						
Altitude (m.a.s.l.)	2,235	1,833	2,747			
Mean depth (m)	3.4	0.5	10.3	1.5	0.2	7.9
Surface area (ha)	13.1	1.8	45.4			
Physicochemical variables						
pH	8.46	7.47	9.30	8.66	7.45	9.58
Conductivity (μ Scm ⁻¹)	203	75	471	283	135	824
Dissolved Oxygen (%)	109	82	162	153	84	304
Temperature (°C)	19.4	15.1	21.9	20.9	14.4	29.9
Secchi disc transparency (m)	0.46	0.05	1.50	0.44	0.12	2.00
Total phosphorus (µgl ⁻¹)	52	11	221			
Total nitrogen (µgl ⁻¹)	850	232	2,411			
Suspended matter (mg 1 ⁻¹)	42	6	737	61	6	596
Biotic components						
Chlorophyll-a (µgl ⁻¹)	187	36	773	204	12	1,716
Cyanobacteria (mg C 1 ⁻¹)	1,281	0	13,143	1,230	0	10,675
Zooplankton (µgl ⁻¹ dry weight)	295	9	1,603	722	8	4,137
Rotifers (µgl ⁻¹ dry weight)	16	0	135	177	0	4,127
Copepods (µgl ⁻¹ dry weight)	144	2	683	120	2	755
Cladocerans (µgl ⁻¹ dry weight)	135	1	1,128	308	1	2,065
Daphnia (µgl ⁻¹ dry weight)	18	0	254	117	0	1,597
Garra (CPUE; kg net ⁻¹ day ⁻¹	2.4	0	24.6			
fresh weight)	2.4	0	24.0			
Tilapia (CPUE)	0.6	0	5.9			
Vegetation cover (% cover)	6	0	50	9	0	50

Table 1.4 Summary of some limnological variables of the reservoirs in Tigray for the wet and dry season of the year 2004/2005 (Dejenie et al. 2008)

1.6. Objectives and thesis outline

Reservoirs are important sources of water for agriculture and other domestic purposes. Although the direct use of the reservoirs is vivid, the occurrence of potentially toxic cyanobacteria blooms is a major threat to the use of water. Cyanobacteria are well documented as being able to potentially synthesize an array of toxic substances. These toxins have been shown to have high toxicity to vertebrates including mammals. Although a large body of literature is available on bloom-forming cyanobacteria from temperate regions, research on these potentially toxic cyanobacteria in sub/tropical Africa is still limited. There is a need for studies on the occurrence of cyanobacteria blooms and the mechanisms that lead to bloom formation and toxin production. It is therefore the main aim of the present work to gain better understanding of the conditions that favor cyanobacteria bloom formation in the reservoirs in Tigray. To achieve this, we carried out field surveys on the occurrence and species composition of phytoplankton in 32 reservoirs in the region, with special emphasis on cyanobacteria. The descriptive information collected was supplemented with an experimental approach, designed to obtain insight in the relative importance of biotic (fish, zooplankton) and abiotic factors in determining the phytoplankton community composition. In addition, seasonal monitoring was carried out for eight selected reservoirs during a one-year period. We also explored bacterioplankton community structure in the reservoirs and to what extent they relate to environmental conditions and cyanobacterial blooms.

In the first part of this thesis (Chapter 2, 3 and 4), we focus on patterns of phytoplankton and bacterioplankton community structure as observed in the field. In a second part (Chapter 5 and 6), we present the results of experiments designed to test for specific mechanisms underlying the occurrence of cyanobacteria blooms in the reservoirs.

Chapter 2 deals with the phytoplankton community in 32 reservoirs with an emphasis on the dominant cyanobacteria of these communities. This study accompanies a large-scale survey in which the reservoirs were studied for a wide range of abiotic variables (Dejenie et al., 2008) and zooplankton. In an appendix to this chapter, we present the list of

phytoplankton genera that were observed in the study reservoirs, as there were no previous studies on the biota of these reservoirs.

Chapter 3 presents the seasonal changes in the phytoplankton community and their relations to changes in environmental variables of eight reservoirs selected to represent the range of environmental gradients in the region. This chapter presents data collected over a one year period (September 2005 - August 2006).

Chapter 4 describes the bacterioplankton community structure of the reservoirs in relation to abiotic and biotic environmental conditions in the reservoirs. More specifically, we also focused on whether the bacterioplankton taxon composition differed among reservoirs with and without *Microcystis* blooms.

Chapter 5 presents the results of two enclosure experiments designed to determine the impact of the presence of fish (*Garra*) on phytoplankton community composition and the occurrence and abundance of *Microcystis*. The experimental design allowed us to assess the relative importance of bottom-up and top-down control for phytoplankton species composition and cyanobacteria blooms.

Chapter 6 presents the results of a field experiment in which we manipulated zooplankton biomass so as to explore the extent to which phytoplankton community structure and the abundance of *Microcystis* can be influenced by top-down effects exerted by zooplankton.

In a final chapter (Chapter 7), we briefly discuss our research findings, and translate our observations to suggestions for future study and for management of the reservoirs with the aim of reducing the occurrence and toxicity of cyanobacterial blooms.

References

- Adelman, W.J., J.F. Fohlmeister, J.J. Sasner, and M. Ikawa, 1982. Sodium channels blocked by aphantoxin obtained from the blue-green alga. *Aphanizomenon flos-aquae. Toxicon*, **20**:513–516
- Aguiaro, T., and E.P. Caramaschi, 1998. Trophic guilds in fish assemblages in three coastal lagoons of Rio de Janeiro State (Brazil). *Verh. Internat. Verein. Limnol*, **26**: 2166-2169
- Akin-Oriola, G.A., 2003. On the phytoplankton of Awab reservoir, Ibadan, Nigeria. *Rev. Biol. Trop.*, **51**: 99-106
- Andersen, T. and D.O Hessen, 1991. Carbon, nitrogen, and phosphorus content of freshwater zooplankton. *Limnology and Oceanography*, 36: 807-814
- Attayde, J.L. and L-A. Hansson, 2001. Fish-mediated nutrient recycling and the trophic cascade in lakes. *Can. J. Fish Aquat. Sci.*, **58**: 1924-1931
- Azam, F., T. Fenchel, J.G. Field, J.S. Gray, L.A. Meyer-Reil, and F. Thingstad, 1983. The ecological role of water-column microbes in the sea. *Marine Ecology Progress Series*, 10: 257-263
- Bachmann, R.W., C.A. Horsburgh, M.V. Hoyer, L.K. Mataraza and D.E. Canfield, 2002. Relations between trophic state indicators and plant biomass in Florida lakes. *Hydrobiologia*, **470** : 219–234
- Barreiro, A., C. Guisande, I. Maneiro, A.R. Vergara, I. Riveiro, P. Iglesias, 2007.
 Zooplankton interactions with toxic phytoplankton: Some implications for food web studies and algal defence strategies of feeding selectivity behaviour, toxin dilution and phytoplankton population diversity. *Acta Oecologica*, 32: 279 290
- Brooks, J.L. and S.I. Dodson, 1965. Predation, body size, and composition of plankton. *Science*, **150**: 28-35
- Begon M., J.L. Harper and C.R. Townsend, 1996. Ecology. Blackwell Science, Oxford. pp.888
- Begon, M., J.L. Harper and C.R. Townsend, 2006, Ecology: From Individuals to Ecosystems. Oxford, Blackwell, pp. 738

- Botes, D.P., H. Kruger, and C.C. Viljoen, 1982a. Isolation and characterization of four toxins from the blue-green alga *Microcystis aeruginosa*. *Toxicon*, **20**:945-954
- Botes, D.P., C.C. Viljoen, H. Kruger, P.L. Wessels, and D.H. Williams, 1982b.
 Configuration assignments of the amino acid residues and the presence
 Mmethyidehydroalanine in toxins from the blue-green alga, *Microcystis aeruginosa*.
 Toxicon, 20:1037-1042
- Botes, D.P., Wessels P.L., Kruger H., Runnegar M.T.C., Santikarn S., Smith R.J., Barna J.C.J., and Williams D.H., 1985. Structural studies on cyanoginosins-LR, -YR, -YA, and -YM, peptide toxins from *Microcystis aeruginosa*. J. Chem. Soc. Perkin. Trans., 1: 2747-2748
- Bouvy, M., R. Molica, S. De Oliveira, M. Marinho, and B. Beker, 1999. Dynamics of a toxic cyanobacterial bloom (*Cylindrospermopsis raciborskii*) in a shallow reservoir in the semi-arid region of north-east Brazil. *Aquatic Microbial Ecology*, **20**: 285-297
- Bouvy, M., D. Falcao, M. Marinho, M. Pagano, and A. Moura, 2000. Occurrence of *Cylindrospermopsis* (Cyanobacteria) in 39 Brazilian tropical reservoirs during the 1988 drought. *Aquatic Microbial Ecology*, 23: 13-27
- Branco, C.W.C., B. Kozlowsky-Suzuki, and F.A. Esteves, 2007. Environmental changes and zooplankton temporal and spatial variation in a disturbed Brazilian coastal lagoon. *Braz. J. Biol.*, 67: 251-262
- Burks, R.L., E. Jeppesen, D.M. Lodge, T. Lauridsen, 2002. Diel Horizontal Migration of Zooplankton: Costs and benefits of inhabiting littoral zones. *Freshw. Biol.*, 47: 343-365
- Burns, C.W., 1987. Insights into zooplankton-cyanobacteria interactions derived from enclosure studies. New Zealand Journal of Marine and Freshwater Research, 21: 477-482
- Camberlin, P., 1997. Rainfall anomalies in the source region of the Nile and their connection with the Indian summer monsoon. *Journal of Climate*, **10**: 1380-1392

- Camberlin, P., S. Janicot and I. Poccard, 2001. Seasonality and atmospheric dynamics of the teleconnection between African rainfall and tropical sea-surface temperature: Atlantic vs. ENSO. *International Journal of Climatology*, 21: 973 – 1005
- Carmichael, W.W., 1994. The toxins of cyanobacteria. Scentific America. 270:78-86
- Carmichael, W.W., D.F. Briggs, and P.R. Gorham, 1975. Toxicology and pharmacological action of *Anabaena flos-aquae* toxin. *Science*, **187**: 542-544.
- Carmichael, W.W., J.T. Eschedor, G.M.L. Patterson, and R.E. Moore, 1988b. Toxicity and partial structure of a hepatotoxic peptide produced by the cyanobacterium *Nodularia spumigena* Mertens emend. L 575 from New Zealand. *Appl. Environ. Microbiol.*, 54: 2257-2263
- Carmichael, W.W., V. Beasley, D.L. Bunner, J.N. Eloff, I. Falconer, P. Gorham, K. Harada, T. Krishnamurthy, M.J. Yu, R.E. Moore, et al. 1988b. Naming of cyclic heptapeptide toxins of cyanobacteria (blue-green algae) [Letter to the Editor]. *Toxicon*, 26: 971-973
- Carmichael, W.W., N.A. Mahmood, E.G. Hyde. Natural toxins from cyanobacteria. In: Marine Toxins, Origin, Structure, and Molecular Pharmacology. Vol ACS Symposium Series 418 (Hall S., Strichartz G., eds). Washington, DC:American Chemical Society, 1990;87-106
- Carmicahel, W.W. 1992. Cyanobacteria secondary metabolites: the cyanotoxins. *Journal of Applied Bacteriology*, **72**: 445-459
- Carpenter, S.L. and J.F. Kitchell (eds.), 1993. The trophic cascade in lakes. Cambridge University Press, Cambridge, UK. pp. 385
- Carpenter S.R., J.F. Kitchell and J.R. Hodgson, 1985. Cascading trophic interactions and lake productivity. *Bioscience*, **35**: 634-638
- Carpenter, S.R., J.F. Kitchell, J.R. Hodgson, P.A. Cochran, J.J. Elser, M.M. Elser, D.M. Lodge, D. Kretchmear, X. He, and C.N. Von Ende, 1987. Regulation of lake primary productivity by food web structure. *Ecologl*, 68: 863-1876

- Castenholz, R.W., 2001. Phylum VX. Cyanobacteria. Oxygenic photosynthetic bacteria. In The Archaea and the deeply branching and phototrophic Bacteria. Boone, D.R., Castenholz, R.W., and Garrity G.M. (eds.). Bergey's manual of systematic bacteriology, 2nd edition, vol. 1. Berlin, Germany: Springer-Verlag, pp. 473-599
- Chambers, P.A., E.E. Prepas, M.L. Bothwell and H.R. Hamilton, 1989. Roots versus shoots in nutrient uptake by aquatic macrophytes in flowing waters. *Can. J. Fish Aquat. Sci.*, 46: 435–439
- Chorus, I. and J. Bartram (eds.), 1999. Toxic cyanobacteria in water: a guide to their public health consequences, monitoring and management. World Health Organization.
- Cloern, J. E., 1996. Phytoplankton bloom dynamics in coastal ecosystems: a review with some general lessons from sustained investigation of San Francisco Bay, California. *Rev. Geophys.*, **34**: 127–168
- Codd, G.A., 1984. Toxins of freshwater cyanobacteria. Microbiological Science, 1: 48-52
- Codd, G.A., 1995. Cyanobacterial toxins: occurrence, properties and biological significance. *Wat. Sci. Tech.* **32**: 149-156.
- Codd, G.A., J. Lindsay, F.M. Young, L.F. Morris, and J.S. Metcalf, 2005. Harmful cyanobacteria from mass mortalities to management measures. Pp. 1-23. In Hiusman, J. Maththijs, HCP
- Comenetz, J., and C. Caviedes, 2002. Climate variability, political crises, and historical population displacements in Ethiopia. *Environmental Hazards*, **4**: 113-127
- Conrow, R., A.V. Zale, and R.W. Gregory, 1990. Distributions and abundances of early stages of fishes in a Florida lake dominated by aquatic macrophytes. *Trans. Am. Fish. Soc.*, **119**: 521-528
- Constable, M., 1984. Resources of rural development in Ethiopia. Ethiopian highlands reclamation study. Working paper 17, FAO/Ministry of Agriculture, Addis Ababa; Ethiopia

- Constanza, R., R. d'Arge, R. De Groot, S. Farber, M. Grasso, B. Hannon, K. Limburg, S. Naeem, R.V. O'Neill, J. Paruelo, R.G. Raskin, P. Sutton and M. van den Belt, 1997.
 The value of the world's ecosystem services and natural capital. *Nature*, 387: 253-260
- Central Statistics Authority (CSA), 2000. Statistical abstracts 2000. Federal Democratic Republic of Ethiopia, Central Statistical Authority: Addis Ababa; 403 pp.
- Central Statistics Authority (CSA), 2007. Projected population of Ethiopia, July 2007. Federal Democratic Republic of Ethiopia, Central Statistical Authority (http://www.csa.gov.et/text_files/national%20statistics%202007/Population.pdf)
- Currie, D.J., 1990. Large-scale variability and interactions among phytoplankton, bacterioplankton and phosphorus. *Limnology and Oceanography*, **35**: 1437-1455
- Dagnachew, L., F. Gasse, O. Radakovitch, C. Vallet-Coulomb, R. Bonnefille, D. Verschuren, E. Gibert, and D. Barker, 2002. Environmental changes in a tropical lake (Lake Abiyata, Ethiopia) during recent centuries. *Palaeogeography, Palaeoclimatology, Palaeoecology*, **187**: 233–258.
- Daniel, G., 1977. Aspects of climate and water budget in Ethiopia. Addis Ababa printing press, 77 pp.
- De Bernardi, R., and G. Giussani., 1990. Are blue-green algae a suitable food for zooplankton? An overview. *Hydrobiologia*, **200/2001**: 29-41
- Dejenie, T., Tsehaye A., L. De Meester, Afework M., Abreha G., S. Risch, A. Pals, K. Van der Gucht, W. Vyverman, J. Nysen, S. Deckers, and J. Decklerck, 2008, Limnological and ecological characteristics of tropical highland reservoirs in Tigray, Northern Ethiopia. *Hydrobiologia*, 610:193–209
- DeMott W.R., 1989. The role of competition in zooplankton succession. In: Sommer, U. (Ed.). Plankton ecology: succession in plankton communities. Springer-Verlag

- De Mott, W.R., and S. Dhawale, 1995. Inhibition of in vitro protein phosphatase activity in three zooplankton species by microcystin-LR, a toxin from cyanobacteria. *Arch. Hydrobiol.*, **134**: 417-424
- DeMott, W.R., 1999. Foraging strategies and growth inhibition in five daphnids feeding on mixtures of a toxic cyanobacterium and a green alga. *Freshwater Biology*, **42**: 263– 274
- Devlin, J.P., O.E. Edwards, , P.R. Gorham, N.R. Hunter, R.K. Pike, and B. Stavric, 1977. Anatoxin-a, a toxic alkaloid from *Anabaena flos-aquae* NCR-44 h. *Can. J. Chem.* **55**: 1367–1371
- De Wit, J., 2003. Stuwmeren in Tigray (Noord-Ethiopië): kenmerken, sedimentatie en sedimentbronnen. Unpublished M.Sc. thesis, Department of Geography, K.U.Leuven.
- Dittmann, E., B. Neilan, M. Erhard, H. von Dohren, T. Borner, 1997. Insertional mutagenesis of a peptide synthetase gene that is responsible for hepatotoxin production in the cyanobacterium *Microcystis aeruginosa* PCC 7806. *Mol. Microbiol.*, 26: 779-787
- Dodson S.I., 1974. Zooplankton competition and predation, an experimental test of the size efficiency hypothesis. *Ecology*, 55: 605–613
- Dumont, H.J., 1994. On the diversity of the Cladocera in the tropics. *Hydrobiologia*, **272**: 27-38
- Edwards, C., K.A. Beattie, C.M. Scrimgeour, and G.A. Codd, 1992. Identification of anatoxin-a in benthic cyanobacteria (blue-green algae) and in associated dog poisonings at Loch Insh, Scotland. *Toxicon*, **30**:1165–1175
- Elder, G.H, P.R. Hunter, and G.A. Codd., 1993. Hazardous freshwater cyanobacteria (blue green algae). *Lancet*, **341**:1519-1520
- Ellis M.M. 1941. Freshwater impoundments. In: Baxter R.M. 1977. Environmental effects of dams and impoundments. *Ann. Rev. Ecol. Sys.*, **8:** 255 283

- Elser J.J. 1992. Phytoplankton dynamics and the role of grazers in Castle Lake, California. *Ecology*, **73**: 887-902
- Elser, J.J., 1999. The pathway to noxious cyanobacteria blooms in lakes: the food web as the final turn. *Freshwater Biology*, **42**: 537-543
- Elser, J.J., and J. Urabe, 1999. The stoichiometry of consumer-driven nutrient recycling: theory, observations, and consequences. *Ecology*, **80**: 745-751
- El-Swaify, S.A and H. Hurni, 1996. Trans-boundary effects of soil erosion and conservation in the Nile basin. *Land Husbandry*, 1: 6-21
- Eriksson, J.E., D. Toivola, J.A. Meriluoto, H. Karaki, Y.G. Han, and D. Hartshorne, 1990.
 Hepatocyte deformation induced by cyanobacterial toxins reflects inhibition of protein phosphatases. *Biochem. Biophys. Res. Commun.*, 173: 1347-1353
- Falconer I.R., 1989. Effects on human health of some toxic cyanobacteria (blue green algae) in reservoirs, lakes and rivers. *Tox. Assessment*, **4**: 175-184
- FAO, 1986. Highlands reclamation study Ethiopia. Final report, Volumes 1 and 2. Food and Agriculture Organization, Rome.
- Ferreira, F.M.B., J.M.F. Soler, M.L. Fidalgo, and V.P. Fernandez, 2001. PSP toxins from *Aphanizomenon flos-aquae* (cyanobacteria) collected in the Crestuma-Lever reservoir (Douro River, Northern Portugal). *Toxicon*, **39**: 757–761
- Flett, R.J., D.W. Schindler, R.D. Hamilton and N.E.R. Campbell, 1980. Nitrogen fixation in Canadian Precambrian shield lakes. *Can. J. Fish. Aquat. Sci.* **37**: 494- 505
- Frenken K. (ed.), 2005. Irrigation in Africa in figures: AQUASTAT survey. FAO water reports. Rome.
- FRW: Eutrophication of freshwaters. Review of current knowledge, FR/R0002. Foundation for Water Research, Marlow, Buckinghamshire, 19p, (2000).
- FRW: Cyanobacterial toxins in the water environment. Review of current knowledge, FR/R0009. Foundation for Water Research, Marlow, Buckinghamshire, 36p, (2004).

- Garcia, P.R., S. Nandini, S.S.S. Sarma, E.R. Valderrama, I. Cuesta, M.D. Hurtado, 2002. Seasonal variations of zooplankton abundance in the freshwater reservoir Valle de Bravo (Mexico). *Hydrobiologia*, **467**: 99-108
- Gilbert, J.J., 1990. Differential effects of *Anabaena affinis* on cladocerans and rotifers: mechanisms and implications. *Ecology*, **71**: 1727-1740
- Gliwicz, Z.M., 1990. Why do cladocerans fail to control algal blooms? *Hydrobiologia*, **200/201**: 83-98.
- Gliwicz, Z.M. and W. Lampert, 1990. Food threshold in *Daphnia* species in the absence and presence of blue-green filaments. *Ecology*, **71**: 691-702
- Griffiths, J.F., 1972. Ethiopian highlands. In: World survey of climatology (J.F. Griffiths, Ed.), Elsevier, New York.
- Gorham, P. R., and W. W. Carmichael, 1988. Hazards of freshwater blue-green algae (cyanobacteria). In Algae and human affairs, (Eds.) C. A. Lembi and J. R. Waaland, pp. 403–431. Cambridge University Press.
- Gulati, R.D., 1990. Structural and grazing responses of zooplankton community to biomanipulation of some Dutch water bodies. *Hydrobiologia*, **200/201**: 99-118
- Gumbricht, T., 1993. Nutrient removal process in freshwater submerged macrophyte systems. Ecological Engineering, Elsevier Publishers.
- Habtamu, G., 1990. Problems and constraints in the study, constructions and management of small-scale irrigation projects. National Irrigation (Ethiopia).
- Haile, T., 1988. Causes and characteristics of drought in Ethiopia. J. Agri. Sci., 10: 85-97
- Hamilton, D.P. and S.F. Mitchell, 1996. An empirical model for sediment resuspension in shallow lakes. *Hytdrobiologia*, **317**:209-220
- Haney, J.F., 1987. Field studies on zooplankton-cyanobacteria interactions . N.Z.J. Marine & Freshw. Res. 2 1: 467 -475
- Haney, J.F., D.J. Forsyth and M.R. James., 1994. Inhibition of zooplankton filtrating rates by dissolved inhibitors produced by naturally occurring cyanobacteria. *Archiv fur Hydrobiologie*, **132**: 1-13

- Harris, G.P., and G. Baxter, 1996. Interannual variability in phytoplankton biomass and species composition in a subtropical reservoir. *Freshwater Biology*, **35**:545-560
- Hart, R.C., 2006. Food web (bio-)manipulation of South African reservoirs viable eutrophication management prospect or illusory pipe dream? A reflective commentary and position paper. http://www.wrc.org.za
- Hawkins, P.R, M.T.C. Runnegar, A.R.B. Jackson, I.R. Falconer, 1985. Severe hepatotoxicity caused by the tropical cyanobacterium (bluegreen alga) (Woloszynska) Seenaya and Subba Raju isolated from a domestic water supply reservoir. Appl. Environ. Microbiol., 50: 1292-1295
- Hawkins, P. and W. Lampert, 1989. The impact of *Daphnia* body size on filtering rate inhibition in the presence of filamentous cyanobactrium. *Limnology and Oceanography*, 34: 1084–1089
- Hawkins, P.R., N.R. Chandrasena, G.J. Jones, A.R. Humpage and I.R. Falconer 1997.
 Isolation and toxicity of *Cylindrospermopsisraciborskii* from an ornamental lake. *Toxicon.*, 35: 34 1-346
- Henriksen, P., W.W. Carmichael, J. An, and Ø. Moestrup, 1997. Detection of an anatoxina(s)-like anticholinesterase in natural blooms and cultures of cyanobacteria/bluegreen algae from Danish lakes and in the stomach contents of poisoned birds. *Toxicon*, **35**: 901–913
- Holmlund, C.M. and M. Hammer, 1999. Ecosystem services generated by fish and population. *Ecological economics*, **29**: 253-268
- Huisman, J., H.C.P. Matthijs, and P.M. Visser (Eds), 2005. Harmful cyanobacteria. Springer, Berlin.
- Humpage, A., J. Rositano, A. Bretag, R. Brown, P. Baker, B. Nicholson, and D. Steffensen,
 1994. Paralytic shellfish poisons from Australian cyanobacterial blooms. *Aust. J. Mar. Freshwat. Res.*, 45: 761-771

- Humpage, A.R, and I.R. Falconer, 1999. Microcystin-LR and liver tumour promotion: effects on cytokinesis, ploidy, and apoptosis in cultured hepatocytes. *Env. Toxicol.*, 14: 61-75
- Hurni, H., 1993. 'Land Degradation, Famines and Resource Scenarios in Ethiopia', inPimental D. (ed.), World Soil Erosion and Conservation, Cambridge: Cambridge University Press.
- Jayatissa, L.P., E.I.L. Silva, J. McElhiney, and L.A. Lawton, 2005. Occurrence of toxigenic cyanobacterial blooms in freshwaters of Sri Lanka. Systematic and Applied Microbiology, 29: 156-164
- Jeppesen, E., T.L. Lauridsen, T. Kairesalo, and R.M. Perrow, 1997a. Impact of submerged macrophytes on fish-zooplankton interactions in lakes: the structuring role of submerged macrophytes in lakes. Ecological studies, 131, Springer.
- Jeppesen, E., J.P. Jensen, M. øndergaard, T.L. Lauridsen, L.J. Pedersen, and L. Jensen, 1997b. Top-down control in freshwater lakes: the role of nutrient state, submerged macrophytes and water depth. *Hydrobiologia*, 342/343: 151-164
- Jeppesen, E., J.P. Jensen, M. Søndergaard, T. Lauridsen, and F. Landkildehus, 2000. Trophic structure, species richness and biodiversity in Danish lakes: changes along a phosphorus gradient. *Freshwater Biology*, **45**: 201-218
- Jeppesen, E., M. Sondergaard, N. Mazzeo, M. Meerhoff, C. Branco, V. Huszar and F. Scasso, 2005. Lake restoration and biomanipulation in temperate lakes: relevance for subtropical and tropical lakes. In Reddy, M. V. (ed.), Tropical Eutrophic Lakes: Their Restoration and Management. Science Publishers, Enfield, 331–359.
- Jones, G.J., I.R. Falconer, and R.M. Wilkins, 1995. Persistence of cyclic peptide toxins in dried cyanobacterial crusts from Lake Mokoan, Australia. *Environ. Toxicol. Wafer Quality*, **10**: 19-24
- Jochimsen, E.M., W.W. Carmichael, J.S. An, D.M. Cardo, S.T. Cookson, C.E.M. Holmes, M. B. Antunes, D.A. de Melo-Filho, T.M. Lyra, V.S.T. Barreto, S.M.F.O. Azevedo,

and W.R. Jarvis, 1998. Liver failure in death after exposure to microcystins at a haemodialysis centre in Brazil. *N. Eng. J. Med.*, **338:** 873-878

- Kaas, H., and P. Henriksen, 2000. Saxitoxins (PSP toxins) in Danish lakes. *Water Res.* 34: 2089–2097
- Kahru, M., U. Horstmann, and O. Rud, 1994. Increased cyanobacterial blooming in the Baltic Sea detected by satellites: Natural fluctuation or ecosystem change? *Ambio*, 23: 469–472
- Kairesalo, T., S. Laine, E. Luokkanen, T. Malinen and J. Keto, 1999. Direct and indirect mechanisms behind successful biomanipulation. *Hydrobiologia*, **395/396**: 99–106
- Keating, K.I., 1978. Blue-green algal inhibition of diatom growth: transition from mesotrophic to eutrophic community structure. *Science*. **199**: 971-973.
- Kimmel, B.L., O.T. Lind, and L.J. Paulson, 1990. Reservoir primary production. *In*: Thornton, K. W., Kimmel B. L. and Payne F. E. (eds), Reservoir limnology: Ecological perspectives. John Wiley and Sons, Inc., New York.
- Kirk, L.K., and J.J. Gilbert, 1992. Variation in herbivore response to chemical defences: zooplankton foraging on toxic cyanobacteria. *Ecology*, **73**: 2208–2217
- Keleti, G., and J.L. Sykora, 1982. Production and properties of cyanobacterial endotoxins. *Applied and Environmental Microbiology*, **43**: 104 – 109
- Keleti, G., J.L. Sykora, E.C. Lippy, and M.A. Shapiro, 1978. Composition and biological properties of lipopolysaccharides isolated from *Schizothrix calcicola* (Ag.) Gomont (Cyanobacteria). *Applied and Environmental Microbiology*, **38**: 471-477
- Krebs C.J. 1994, Ecology: the experimental analysis of distribution and abundance. Harper Collins, New York. 800 pp.
- Krienitz, L., A. Ballot, K. Kotut, C. Wiegand, S. Pütz, J.S. Metcalf, G.A. Codd, and S. Pflugmacher, , 2003. Contribution of hot spring cyanobacteria to the mysterious

deaths of Lesser Flamingos at Lake Bogoria, Kenya. FEMS Microbiol. Ecol., 43: 141-148

- Lair, N., 1990. Effects of invertebrate predation on the seasonal succession of a zooplankton community: a two year study in Lake Aydat, France. *Hydrobiologia*, 198: 1-12
- Lagos, N., H. Onodera, P. A. Zagatto, D. Andrinolo, S.M. Azevedo, and Y. Oshima, 1999. The first evidence of paralytic shellfish toxins in the freshwater cyanobacterium *Cylindrospermopsis raciborskii*, isolated from Brazil. *Toxicon* **37**: 1359–1373
- Lampert, W., 1987a. Predictability in lake ecosystems: the role of biotic interactions. In Schulze, E.-D. & H. Zwôlfer (eds), Ecological Studies, Vol. 61. Springer-Verlag Berlin, Heidelberg: 333-345.
- Lampert, W., 1987b. Laboratory studies on zooplankton–cyanobacteria interactions. New Zealand Journal of Marine and Freshwater Research, 21: 483–490
- Lampert, W. and U. Sommer, 1997. Limnology: The Ecology of lakes and Streams. Oxford University Press, New York, pp. 400.
- Lauridsen, T.L., L.J. Pedersen, E. Jeppesen, and M. Søndergaard, 1996. The importance of macrophyte bed size for cladoceran composition and horizontal migration in a shallow lake. J. Plank. Res., 18: 2283-2294
- Lazzaro, X., 1997. Do the trophic cascade hypothesis and classical biomanipulation approaches apply to tropical lakes and reservoirs? *Ver. Int. Verein. Limnol.*, **26**: 719-730
- Lemma, G., 1996. Climate Classification of Ethiopia. Bulletin 3. National meteorological Services Agency: Addis Ababa.
- Li, R., W.W. Carmichael, , and P. Pereira, 2003. Morphological and 16S rRNA gene evidence for reclassification of the paralytic shellfish toxin producing *Aphanizomenon flos-aquae* LMECYA 31 as *Aphanizomenon issatschenkoi* (Cyanophyceae). J. Phycol. **39**: 814–818
- Lips, I, 2005. Abiotic factors controlling the Cyanobacterial bloom occurrence in the gulf of Finland. PhD thesis, Tartu University Press.

- Luck, G.W., G.C. Daily, and P.R. Ehrlich, 2003. Population diversity and ecosystem services. *Trends in Ecology and Evolution*, **18**: 331-336
- Mackintosh, C., K.A. Beattie, S. Klumpp, P. Cohen, and G.A. Codd, 1990. Cyanobacterial microcystin-LR is a potent and specific inhibitor of protein phosphatases 1 and 2A from both mammals and higher plants. *FEBS Lett.*, **264**:189–192
- Mackay, N.A. and J.J. Elser, 1998. Nutrient recycling by *Daphnia* reduces N₂ fixation by cyanobacteria. *Limnology & Oceanography*, **43**: 347-354.
- Mahmood, N.A., and W.W. Carmichael, 1986. Paralytic shellfish poisons produced by the freshwater cyanobacterium *Aphanizomenon flos-aquae* NH-5. *Toxicon*, 24:175-186Mahmood, N.A., and W.W. Carmichael, 1987. Anatoxin-a(s), an anticholinesterase from the cyanobacterium *Anabaena flos-aquae* NRC-525-17. *Toxicon*, 25:1221–1227
- Matsunaga, S., R. Moore, W. Niemczura, W.W. Carmichael, 1989. Anatoxin-a(s), a potent anticholinesterase from *Anabaena flosa-quae*. J. Am. Chem. Soc. **111**: 8021-023
- Matsushima, R., S. Yoshizawa, M.F. Watanabe, M. <u>Furusawa</u>, W.W. <u>Carmichael</u>, and H. <u>Fujiki</u>, 1990. In vitro and in vivo effects of protein phosphatise inhibitors; microcystin and nodularin, on mouse skin and fibroblasts. *Biochemical and Biophysical Research Communications*, **171**: 867-874
- Matveev, V, L. Matveeva, and G.J. Jones, 1994. Study of the ability of *Daphnia carinata* king to control and resist cyanobacterial toxicity: implications for biomanipulation in Australia. *Australian Journal of Marine and Freshwater Research*, **45**: 889–904
- Mazzeo, N., L. Rodríguez-Gallego, C. Kruk, M. Meerhoff, J. Gorga, C. Lacerot, F. Quintans, M. Loureiro, D. Larrea, and F. García-Rodrígues, 2003. Effect of *Egeria densa Planch*. beds on a shallow lake without piscivorous fish. *Hydrobiologia*, **506–509**: 591–602
- McCann, J., 1995. People of the plow: An agricultural history of Ethiopia, 1800-1900. University of Wisconsin Press, Madison, USA. 298 pp.

- McQueen D.J. and D.R.S. Lean, 1987. Influence of water temperature and nitrogen to phosphorus ratio on the dominance of blue-green algae in Lake St. George, Ontario. *Can. J. Fish. aquat. Sci.* 44: 598-604
- Meerhoff, M., N. Mazzeo, B. Moss, and L. Rodríguez-Gallego, 2003. The structuring role of free-floating versus submerged plants in a subtropical shallow lake. *Aquat. Ecol.*, **37**: 377-391
- Meerhoff, M., C. Fosalba, C. Bruzzone, N. Mazzeo, W. Noordoven and E. Jeppesen, 2006.
 An experimental study of habitat choice by *Daphnia*: plants signal danger more than refuge in subtropical lakes. *Freshwater Biology*, **51**: 1320–1330
- Meissner, K, E., Dittmann, and T. Borner, 1996. Toxic and non-toxic strains of the cyanobacterium *Microcystis aeruginosa* contain sequences homologous to peptide synthetase genes. *FEMS Microbiol., Lett.* 135:295-303
- Mehari, G-M. and B. Valquist, 1976. Famine in Ethiopia a brief review. *Journal of Nutrition*, **29**: 1016-1020
- Meriluoto, J.A., A. Sandstrom, J.E. Eriksson, G. Remaud, A.G. Craig, J. Chattopadhyaya, 1989. Structure and toxicity of a peptide hepatotoxin from the cyanobacterium *Oscillatoria agardhii. Toxicon*, **27**:1021-1034
- MNRDEP, 1994. National Conservation Strategy. Volume 2: National Policy in Natural Resources and the Environment. Ministry of Natural Resources Development and Environmental Protection. Addis Ababa.
- Mohamed-Saleem, M.A., 1995. Fragile East African highlands: A development vision for smallholder farmers in the Ethiopian highlands. *Outlook on Agriculture*, **24**: 111-116
- Monserrat, J. M., J.S. Yunes, and A. Bianchini, 2001. Effects of *Anabaena spiroides* (cyanobacteria) aqueous extracts on the acetylcholinesterase activity of aquatic species. *Environ. Toxicol. Chem.* **20**:1228–1235

- Moss B., M. Beklioglu, L. Carvalho, S. Kilinc, S. McGowan and D. Stephen, 1997. Vertically-challenged limnology, contrasts between deep and shallow lakes. *Hydrobiologia*, **342/343**: 257-267
- Moss, B., 1998. Ecology of fresh waters, man and medium. Blackwell Scientific Publications, London, pp. 557.
- Moss B., 1990. Engineering and biological approaches to the restoration from eutrophication of shallow lakes in which aquatic plant communities are important components. *Hydrobiologia*, **200/201**, 367–377
- Moss, B. 2000. Biodiversity in freshwaters an issue of species preservation or system functioning? *Environmental conservation*, **27**: 1-4
- Murphy, T.P., D.R. Le and C. Nalewajko, 1976. Blue-green algae: their excretion of iron selective chelators enables them to dominate other algae. *Science*, **192**: 900-902
- Myers, N., 1996. Environmental services of biodiversity. Proc. Natl. Acad. Sci., 93: 2764-2769
- Namikoshi, M., K.L. Rinehart, R. Sakai, K. Sivonen, W.W. and Carmichael, 1990. Structures of three new cyclic heptapeptide hepatotoxins produced by the cyanobacterium (blue-green alga) *Nostoc sp.* strain 152. *J. Org. Chem.* **55**:6135-6139
- Namikoshi, M., K. Sivonen, W.R. Evans, W.W. Carmichael, F. Sun, L. Rouhiainen, R. Luukkainen, K.L. Rinehart, 1992. Two new L-serine variants of microcystins-LR and -RR from *Anabaena sp.* strains 202 Al and 202 A2. *Toxicon*, **30**: 1457-1464
- Negri, A.P., and G.J. Jones, 1995. Bioaccumulation of paralytic shellfish poisoning (PSP) toxins from the cyanobacterium *Anabaena circinalis* by the freshwater mussel *Alathyria condola. Toxicon*, **33**:667-678
- Neill, W.E., 1994. Spatial and temporal scaling and organization of limnetic communities.In: P.S. Guiller, A.G. Hildrew & D.G. Raffaelli (eds.) Aquatic ecology: scale, patterns and process. Blackwell Scientific, Oxford, New York.

- NHMRC (National Health and Medical Research Council), 1994. Health Effects of Toxic Cyanobacteria (Blue Green Algae). Canberra, Australia: Australian Government Publishing Service.
- Nigussie, H., J. Poesen, J. Nyssen, J. De Wit, Mitiku H., G. Govers and J. Deckers, 2006. Reservoirs in Tigray: characteristics and sediment deposition problems. *Land Degradation and Development*, **17**: 211–230
- Nyssen, J., H. Vandenreyken, J. Poesen, J. Moeyersons, J. Deckers, Mitiku H., C. Salles, and G. Govers, 2005. Rainfall erosivity and variability in the Northern Ethiopian Highlands. *Journal of Hydrology*, **311**: 172-187
- Ohtani, L., R.E. Moore, and M.T.C. Runnegar, 1992. Cylindrospermopsin: a potent hepatotoxin from the blue green alga *Cylindrospermopsis raciborskii*. J. Am. Chem. Soc., **114**: 7941-7942.
- Onodera, H., Oshima, Y., Henriksen, P., and Yasumoto, T. 1997. Confirmation of anatoxin-a(s) in the cyanobacterium *Anabaena lemmermannii*, as the cause of bird kills in Danish Lakes. *Toxicon*, **35**:1645–1648
- Orr, P.T. & G.J. Jones, 1998. Relationship between microcystin production and cell division rates in nitrogen-limited *Microcystis aeruginosa* cultures. *Limnol. Oceanogr.*, 43: 1604-1614
- Paerl, H. W., 1988. Nuisance phytoplankton blooms in coastal, estuarine, and inland waters. *Limnol. Oceanogr.*, 33: 823–847
- Pankhurst, R., 1966. The great Ethiopian famine of 1888 1892: a new assessment Part two. *Journal of the History of Medicine and Allied Sciences*, **21**: 271-294
- Park, H.D., M.F. Watanabe, K.I. Harada, H. Nagai, M. Suzuki, M. Watanabe, and H. Hayashi, 1993. Hepatotoxin (microcystin) and neurotoxin (anatoxin-a) contained in

natural blooms and strains of cyanobacteria from Japanese freshwaters. *Nat. Toxins*, 1:353-360

- Pereira, P., H. Onodera, D. Andrinolo, S. Franca, F. Araujo, N. Lagos, and Y. Oshima, 2000. Paralytic shellfish toxins in the freshwater cyanobacterium *Aphanizomenon flos-aquae*, isolated from Montargil reservoir, Portugal. *Toxicon*, **38**: 1689–1702
- Plumley F.G., 1997. Marine algal toxins: biochemistry, genetics, and molecular biology. *Limnology Oceanography*, **42**: 1252–1264
- Pomati, F., S. Sacchi, C. Rossetti, S. Giovannardi, H. Onodera, Y. Oshima, and B.A. Neilan, 2000. The freshwater cyanobacterium *Planktothrix* sp. FP1: Molecular identification and detection of paralytic shellfish poisoning toxins. *J. Phycol.* 36: 553–562
- Pomeroy, L.R. 1974. The ocean's food web, a changing paradigm. Bioscience, 24: 499-504
- Porter, K., 1973. Selective grazing and differential digestion of algae by zooplankton. *Nature*, 244: 179-1 80
- Porter, K.G., and J.D. Orcutt, 1980. Nutritional adequacy, manageability, and toxicity as factors that determine the food quality of green and blue-green algae for *Daphnia*. In Kerfoot W.C. (ed.), Evolution and ecology of zooplankton communities. University Press of New England
- Porter, K.G., and R. McDonough, 1984. The energetic cost of response to blue-green algal filaments by cladocerans. *Limnol. Oceanogr.*, **29**: 365-369
- Portielje, R. and R.M.M. Roijackers, 1995. Primary succession of aquatic macrophytes in experimental ditches in relation to nutrient input. *Aquatic Botany*, **50**: 127–140
- Quinn, W.H. 1993. The large scale ENSO event, the El Nino and other important regional features. *Bull. Inst. fr. Etudes andines*, **22:** 13-34
- Quirós, R., 1998. Fish effects on trophic relationships in the pelagic zone of lakes. *Hydrobiologia*, 361: 101-111

- Rapala, J., K. Sivonen, C. Lyra and S.I. Niemela., 1997. Variation of microcystins, cyanobacterial hepatotoxins in *Anabaena* spp. as a function of growth stimuli. *App. Environ. Microbiol.*, 64: 2206-22 12
- Repavich, W., W.C. Sonzogni, J.H. Standridge, R.E. Wedepohl, and L.E. Meisner, 1990.Cyanobacteria (blue green algae) in Wisconsin waters: acute and chronic toxicity.*Water Res.*, 24: 225-231
- Rinehart, K.L., M. Namikoshi, and B.W. Choi, 1994. Structure and biosynthesis of toxins from blue-green algae (cyanobacteria). *J. Appl. Phycol.* **6**:159-176
- Reinikainen, M., M. Ketola and M. Walls, 1994. Effects of the concentrations of toxic Microcystis aeruginosa and an alternative food on survival of *Daphnia pulex*. *Limnology and Oceanography*, **39**: 424-432
- Richman, S., and S.I. Dodson, 1983. The effect of food quality on feeding and respiration by *Daphnia* and *Diaptomus*. *Limnol. Oceanogr.*, **28**: 948-956
- Robach, F., S. Merlin, T. Rolland and M. Tremolieres, 1996. Ecophysiological approach of water quality bioindication using aquatic plant materials: the role of phosphorus. *Ecologie-Brunoy*, 27: 203–214
- Rothhaupt, K.O., 2000. Plankton population dynamics: food web interactions and abiotic constraints. *Freshwater Biology*, **45**: 105- 109
- Runneger, M., N. Berndt, and N. Kaplowitz, 1995. Microcystin uptake and inhibition of protein phosphatases: effects of chemoprotectants and self-inhibition in relation to known hepatic transporters. *Toxicology and Applied Pharmacology*, **134**: 264-272
- Sarnelle, O., 1993. Herbivore effect on phytoplankton succession in eutrophic lake. *Ecological Monographs*, **63**: 129 – 149
- Scasso, F., N. Mazzeo, J. Gorga, C. Kruk, G. Lacerot, J. Clemente, D. Fabián, S. Bonilla, 2001. Limnological changes of a subtropical shallow hypertrophic lake during its restoration. Two years of whole-lake experiments. *Aquat Conserv: Mar Freshwater Ecosys*, **11**: 31-44

- Scheffer, M., S.H. Hosper, M.-L. Meijer, B. Moss and E. Jeppesen, 1993. Alternative equilibrium in shallow lakes. *Trends in Ecology and Evolution*, **8**: 275–279
- Scheffer, M., S. Rinaldi, Y. A. Kuznetsov and E. H. Van Nes, 1997a. Seasonal dynamics of Daphnia and algae explained as a periodically forced predator-prey system. Oikos, 80: 519–532
- Scheffer, M., S. Rinaldi, A. Gragnani, L.R. Mur and E.H. Van Nes, 1997b. On the dominance of filamentous cyanobacteria in shallow, turbid lakes. *Ecology*, 78: 272–282
- Scheffer, M., 1998. Ecology of shallow lakes. Chapman and Hall, London, pp. 357.
- Scheffer, M., 1999. The effect of aquatic vegetation on turbidity: how important are the filter feeders? *Hydrobiologia*, **408/409**: 307–316
- Scheffer, M., S. Szabo, A. Gragnani, E.H. van Nes, S. Rinaldi, N. Kautsky, J. Norberg, R.M.M. Roijackers and R.J.M. Franken, 2003. Floating plant dominance as a stable state. Proceedings of the National Academy of Science of the United States of America 100: 4040–4045.
- Scheffer, M., and E.H. van Nes, 2007. Shallow lakes theory revisited: various alternative regimes driven by climate, nutrient, depth and lake size. *Hydrobiologia*, **584**: 455-466
- Schindler, D.W., 1977. Factors regulating phytoplankton production and standing crop in the world's freshwaters. *Limnology and Oceanography*, 23: 478 – 486
- Seleshi, Y., and G-R. Demaree, 1995. Rainfall variability in the Ethiopian and Eritrean highlands and its links with the Southern Oscillation Index. *Journal of Biogeography*, 22: 945-952
- Seleshi, Y., and U. Zanke, 2004. Recent changes in rainfall and rainy days in Ethiopia. International Journal of Climatology, 24: 973-983
- Shanko, D. and P. Chamberlain, 1998. The effects of the southwest Indian Ocean tropical cyclones on Ethiopian drought. *International Journal of Climatology*, **18**: 1373–1388
- Shapiro, J., 1973. Blue-green algae: why they become dominant. Science, 179: 382-384
- Shapiro, J., 1984. Blue-green algal dominance in lakes: the role and management significance of pH and CO₂. *Int. Revue ges. Hydrobiol*, **69**: 765-80

- Shapiro, J. and D.I. Wright, 1984. Lake restoration by manipulation: Round Lake, Minnesota the first two years. *Freshwater Biology*, **14**: 371-383
- Shapiro, J., 1990. Current beliefs regarding dominance by blue-greens: the case of importance of CO₂ and pH. *Verh. int. Ver. Limnol*, **24**: 38-54
- Shapiro, J., 1997. The role of carbon dioxide in the initiation of and maintenance of bluegreen dominance in lakes. *Freshwater Biology*, **37**: 307-323
- Sherr, E.B., and B.F. Sherr, 1988. Role of microbes in pelagic food web: A revised concept. *Limnol. Oceanogr.*, **33**: 1225-1227
- Sherr, E.B. & B.F. Sherr, 1991. Planktonic microbes: tiny cells at the base of the ocean's food web. *Trends in Ecology and Evolution*, 6: 50-54
- Sivonen, K., K. Himberg, R. Luukkainen, S.I. Niemelä, G.K. Poon, and G.A. Codd, 1989.
 Preliminary characterization of neurotoxic blooms and strains from Finland. *Toxicol.*Assess. 4:339–352
- Sivonen, K. 1990. Effects of light, temperature, nitrate, orthophosphate and bacteria on growth of and hepatotoxin production by *Oscillatoria agardhii* strains. *App. Environ. Microbiol.*, **56**: 2658-2666
- Sivonen, K, 1996. Cyanobacterial toxins and toxin production. Phycologia, 35:12-24
- Sivonen, K., and G. Jones, 1999. Cyanobacterial toxins. In, I. Chorus and J. Bartram (eds.), Toxic cyanobacteria in water, E and F.N Spon, London, pp.41-111
- Skulberg, O.M., W.W. Carmichael, R.A. Anderson, S. Matsunaga, R.E. Moore, and R. Skulberg, 1992. Investigations of a neurotoxic oscillatorialean strain (cyanophyceae) and its toxin. Isolation and characterization of homoanatoxin-a. *Environmental Toxicology and Chemistry*, **11**, 321–329
- Smith, V.H., 1983. Low nitrogen to phosphorus ratio favour dominance by blue-green algae. *Science*, **221**: 669-671

- Smith, V.H. 1986. Light and nutrient effects on the relative biomass of blue-green algae in lake phytoplankton. *Can. Fish Aquat. Sci.* **43**: 148-153
- Sommer, U., Z.M. Gilwicz, W. Lampert, and A. Duncan, 1986. The PEG model of seasonal succession of planktonic events in fresh waters. *Archiv für Hydrobiologie*, 106: 433-471
- Sommer, U., 1989. Plankton Ecology. Springer-Verlag.
- Steele, J.H., 1974. The structure of marine ecosystems. Harvard University press, Cambridge, Mass. 128 pp.
- Sterner, R.W. 1990. The ratio of nitrogen to phosphorus resupplied by herbivores: zooplankton and the algal competitive arena. *Am. Nat.*, *136*, 209-229
- Straskraba, M., and J. G. Tundisi, 1999. Reservoir Ecosystem Functioning: Theory and Application. In Tundisi, J. G. & M. Straskraba (eds), Theoretical Reservoir Ecology and its Applications. Backhuys Publishers, Leiden: 565–597.
- Tilman, D., S.S. Kilham and P. Kilham, 1982. Phytoplankton community ecology: the role of limiting nutrients. *Annu. Rev. Ecol. Syst.*, **13**: 349-382
- Tilman, D., R. Kiesling, R. Sterner, S.S. Kilham, F.A. Johnson, 1986. Green blue-green and diatom algae: taxonomic differences in competitive ability for phosphorus, silicon, and nitrogen. *Arch. Hydrobiol.*, 106: 473-485
- Tilman, D., J., Knops, D. Wedin, P, Reich, M. Ritchie, and E. Siemann, 1997. The influence of functional diversity and composition on ecosystem processes. *Science*, 277: 1300–1302
- Tilman, D., 1999. Ecological consequences of changes in biodiversity: a search for general principles. *Ecology*, **80**: 1455-1474
- Timms R.M. and B. Moss B. 1984. Prevention of growth of potentially dense phytoplankton populations by zooplankton grazing, in the presence of

zooplanktivorous fish, in a shallow wetland ecosystem. *Limnol. Oceanogr.*, 29: 472–486

- Tudorancea, C., G-M. Zinabu, and D. Elias, 1999. Limnology in Ethiopia. In: Wetzel. R.G. and Gopal, B. (Eds.) Limnology in Developing Countries. International Association for Limnology.
- Tsehaye A., Tadesse D., Declerck, S., Nyssen, J., Van der Gucht, K., Risch, S., Rousseaux, S., De Wit, J., Afework M., Nigussie H., Abreha G., Poesen, J., Deckers, J., Vyverman, W., De Meester, L., 2007. Ecological atlas of reservoirs in Tigray, Northern Ethiopia. Tigray Livelihood Papers No. 4, VLIR, Mekelle University IUC Programme and Zala-Daget Project, 80 pp.
- Valiela, I., 1991. Ecology of Water Columns. In: Barnes R.SK. L.K.H. Mann. Fundamentals of Aquatic Ecology. Blackwell, Oxford, UK. p. 270.
- Vallet-Coulomb, C., Daghachew L., F., Gasse, Y. Travi, and Tesfaye T., 2001. Lake evaporation estimates in tropical Africa (Lake Ziway, Ethiopia). *Journal of Hydrology*, 245: 1-18
- Van Donk, E., R.D. Gulati, A. Iedema and J.T. Meulemans, 1993. Macrophyte-related shifts in the nitrogen and phosphorus contents of the different trophic levels in a biomanipulated shallow lake. *Hydrobiologia*, 251: 19–26
- van Ginkel, C.E., 2004. A national survey of the incidence of cyanobacterial blooms and toxin production in major impoundments. Rapport South Africa.
- Vanni, M.J., 1986. Fish predation and zooplankton demography: indirect effects. *Ecology*, **62**: 337–354
- Vanni, M.J. and C.D. Layne, 1997. Nutrient recycling and herbivory as mechanisms in the 'top-down' effect of fish on algae in lakes. *Ecology*, **78**: 21–40
- Vanni, M.J., C.D. Layne and S.E. Arnott, 1997. Top-down trophic interactions in lakes: effects of fish on nutrient dynamics. *Ecology*, **78**: 1-20

- Varis, O., 1993. Cyanobacteria dynamics in a restored Finish lake: a long term simulation study. *Hydrobiologia*, **268**: 129-145
- Virgo, K.J, and R.N. Munro, 1978. Soil and erosion features of the central plateau of Tigray, Ethiopia. *Geoderma*, 20: 131–157
- Wasmund, N., 1997. Occurrence of cyanobacterial blooms in the Baltic Sea in relation to environmental conditions. *Int. Revues ges. Hydrobiol.*, 82: 169–184
- Webster, K.E., and R.H. Peters, 1978. Some size-dependent inhibitions of larger cladoceran filterers in filamentous suspensions. *Limnology and Oceanography* **23**: 1238-1245
- Weise, G., G. Drews, B. Jann, and K. Jann, 1970. Identification and analysis of a lipopolysaccharide in cell of the blue-green algae *Anacystis nidulans*. Archives Microbiology, 71: 89-98
- Wetzel, R.G., 2001. Limnology, lake and river ecosystems. 3rd ed. Academic Press, San Diego. 1006 p.
- Whitton, B., 1992. Diversity, ecology and taxonomy of the cyanobacteria. In: Mann H., Carr N. (ed.) Photosynthetic prokaryotes. New York, Plennum Press
- Whitton, B. A., and M. Potts, 2000. Introduction to the cyanobacteria. In The ecology of cyanobacteria: Their diversity in time and space. B. A. Whitton, & M. Potts (eds). Dordrecht, Kluwer Academic Publishers.
- Wilson, E.O. (ed.), 1988. Biodiversity. National Academy Press, Washington.
- Worm, B., E.B. Barbier, N. Beaumont, J.E. Duffy, C. Folke, B.S. Halpern, J.B.C. Jackson, H.K.
 Lotze, F. Micheli, S.R. Palumbi, E. Sala, K.A. Selkoe, J.J. Stachowicz, and R. Watson, 2006.
 Impacts of Biodiversity Loss on Ocean Ecosystem Services. *Science*, **314**: 787-790
- Yoshizawa, S., R. Matsushima, M.F. Watanabe, K. Harada, A. Ichihara, W.W. Carmichael, and H. Fujiki, 1990. Inhibition of protein phosphatises by microcystins and nodularins associated with hepatotoxicity. *Journal of Cancer Research*, 116: 609-614
- Zurawell, R.W., H.R. Chen, J.M. Burke, and E.E. Prepas, 2005. Hepatotoxic cyanobacteria: A review of the biological importance of microcystins in freshwater environments. *Journal of Toxicology and Environmental Health-Part B-Critical Reviews*, **8**: 1-37
- Zaret, T.M., 1980. Predation and freshwater communities. Yale University Press, New Haven, pp. 187.
- Zohary, T. and R.R. Robarts, 1989. Diurnal mixed layers and the long term dominance of *Microcystis aeruginosa*. In: Zurawell, R.W., H.R. Chen, J.M. Burke, and E.E. Prepas, 2005. Hepatotoxic cyanobacteria: A review of the biological importance of microcystins in freshwater environments. *Journal of Toxicology and Environmentl Health-Part B-Critical Reviews*, 8: 1-37
- Zohary, T. and Roberts, R.D. 1990 Hyperscums and the population dynamics of *Microcystis aeruginosa. J. Plankton Res.*, **12**, 423
- Zwahlen, R., 2003. Identification, assessment, and mitigation of environmental impact of dam Projects. In R. S. Ambasht and N. K. Ambasht (eds), Modern Trends in Applied Aquatic Ecology.

Chapter 2

The phytoplankton communities of shallow reservoirs in a semi-arid subtropical region (Tigray, North Ethiopia), with an emphasis on the occurrence of bloom-forming cyanobacteria

Tsehaye Asmelash¹, Tadesse Dejenie¹, Pieter Vanormelingen², Luc De Meester³, Ineke Van Gremberghe², Katleen Van der Gucht², Steven Declerck³ and Wim Vyverman²

(Unpublished manuscript)

¹Mekelle University, PO Box 231, Mekelle, Ethiopia

²Laboratory of Protistology and Aquatic Ecology, Ghent University, Krijgslaan 281 – S8, 9000 Gent, Belgium

³Laboratory of Aquatic Ecology, Katholieke Universiteit Leuven, Ch De Beriotstraat 32, 3000 Leuven, Belgium

Abstract

Small shallow reservoirs are often constructed in tropical regions to ensure water availability for local communities. Knowledge of the aquatic food web in general and phytoplankton in particular, in such reservoirs is generally lacking, even though their shallowness and possible eutrophication implies a risk of cyanobacterial bloom development. Here, we describe the phytoplankton communities, with an emphasis on the occurrence of bloom-forming cyanobacteria, in 32 small reservoirs in a subtropical semiarid region (Tigray, North Ethiopia), which were sampled once in both the wet and dry season. Local phytoplankton richness was low with most reservoirs dominated by a single genus of cyanobacteria (mostly *Microcystis*), chlorophytes, euglenophytes, cryptophytes or dinophytes. Despite their recent construction, spatial configuration of the reservoirs didn't have a significant influence on phytoplankton community structure. In contrast, local reservoir characteristics were strongly associated with the phytoplankton communities. Environmental variables selected by redundancy analysis were zooplankton and pH. Microcystis was associated with a high pH, especially in the rainy season. In the dry season, the most important environmental factors in this exceptionally dry year were reservoir depth and conductivity, with deeper lowerconductivity reservoirs typically dominated by the dinophyte Peridinium. Despite their occurrence in several reservoirs, submerged macrophytes apparently didn't influence phytoplankton community structure, in contrast to the situation in temperate shallow water bodies. The apparent importance of zooplankton for phytoplankton community structure is surprising for (sub) tropical regions and may be related to the absence of fully developed fish communities in these reservoirs.

Keywords: Phytoplankton, shallow reservoir, semiarid subtropical climate, cyanobacteria, *Microcystis*, *Peridinium*

62

Introduction

Phytoplankton is among the main primary producers in aquatic environments. In the tropics, study of phytoplankton communities has focused on natural lakes and rivers (Hecky & Kling 1981, Talling 1986, Elizabeth & Amha 1994, Tudorancea et al., 1999). However, in (semi-) arid regions, many small shallow reservoirs have been constructed to serve as a water source for irrigation and other purposes during the dry season. While the ecology and food web structure of shallow lakes in temperate regions is well established (Meijer et al., 1999, Scheffer & van Nes, 2007), such knowledge is scarce for the (sub)tropics (Tudorancea et al. 1999, Dejenie et al., 2008). The study of these small reservoirs would contribute significantly to the understanding of tropical freshwater ecosystems and the impact of anthropogenic activities on them (Araoye, 2002), including the possible development of harmful algal blooms. In small meso- to eutrophic water bodies in temperate regions, phytoplankton communities are mainly influenced by the structure of the aquatic food web (Schriver et al., 1995, Scheffer, 1998, Muylaert et al., 2003, Vanormelingen et al., 2008), and by seasonal changes in abiotic conditions and the strength of food web interactions (Sommer et al., 1986, Pilkaityté & Razinkovas, 2007). However, important differences in the factors controlling phytoplankton community structure can be expected in a (sub)tropical climate characterized by a more constant high temperature. For instance, top-down control of phytoplankton by zooplankton rarely occurs in tropical lakes, most likely due to the high predation pressure of abundant and continuously reproducing fish on large zooplankton (Dumont, 1994, van Leeuwen et al., 2007), the association of fish with macrophyte beds preventing their use as zooplankton refuge (Meerhoff et al., 2003, 2005), and the higher contribution of, often inedible, cyanobacteria to the phytoplankton in warmer climates (Rondel et al., 2008).

The intensive use of small man-made reservoirs in a (semi-)arid climate often results in their eutrophication, making them especially vulnerable to the occurrence of cyanobacterial blooms, which are increasingly frequent in aquatic ecosystems around the world (Chorus & Bartram, 1999, Van Ginkel, 2003, Huisman et al., 2005, Paerl and Huisman, 2008). The contribution of cyanobacteria to water quality deterioration is widely documented. Next to other negative aspects of phytoplankton bloom formation, a number of common cyanobacteria can produce a wide range of toxins, causing a risk to livestock and human health (Chorus & Bartram, 1999, Codd et al., 2005). Hypotheses to explain cyanobacterial dominance in eutrophic lakes are rather diverse and can be related to several factors: (1) An elevated water temperature, (2) Low photon irradiance, (3) a stable water column, (4) Zooplankton grazing, (5) low carbon dioxide/high pH, and (6) a low N/P ratio (Shapiro, 1990, Paerl & Huisman, 2008). Moreover such blooms are expected to only occur in the phytoplankton-dominated turbid state of shallow lakes, as opposed to the clear-water state dominated by submerged macrophytes (Scheffer, 1998). However, cyanobacterium identity is important; different cyanobacterial species have different physiological traits and requirements and are favored by different combinations of environmental variables (Kardinaal & Visser, 2005). Once established, a cyanobacterial bloom may itself influence (a)biotic conditions in such a way that positive feedback loops are established (increased internal phosphorus loading, low light conditions, high temperature, CO₂ depletion), which stabilize cyanobacterial dominance (Scheffer et al., 1997, Bicudo et al., 2007). This makes it difficult to disentangle cause and consequence of cyanobacterial bloom development.

In Ethiopia, virtually no studies on the phytoplankton communities of small reservoirs and ponds exist, and the scientific information about these reservoirs is very limited (Melaku *et al.*, 1988, Tudorancea *et al.*, 1999). Independent studies on the occurrence and mechanisms of cyanobacterial blooms in small reservoirs are also almost non-existent except for some reports from the Ethiopian Rift Valley natural lakes. Cyanobacteria (including the genus *Microcystis*) were reported as dominant members of the phytoplankton community from these Ethiopian natural lakes (Baxter & Wood, 1965, Wood & Talling, 1988, Elizabeth, *et al.*, 1992, Elizabeth & Amha, 1994, Elizabeth & Willen, 1996, Elizabeth, 1996). Here, we describe the phytoplankton communities of 32 small highland reservoirs in Tigray, North Ethiopia, and identify the main environmental factors correlated with phytoplankton community variation, with an emphasis on the importance of the aquatic food web structure. In addition, we tested specific hypotheses on the factors influencing the abundance of *Microcystis*, the main bloom-forming cyanobacterium in Tigray.

Material and methods

Reservoir characteristics

This study was carried out in 32 shallow semi-arid tropical reservoirs in the highlands of Tigray, the northernmost regional state of Ethiopia (figure 2.1). A detailed limnological description of the reservoirs is given in Dejenie et al. (2008). The sampled reservoirs are situated at altitudes between 1833 and 2747 meters above sea level and are all quite small (surface area ranged between 7.78 and 45.4 ha) and shallow (maximum depth in the wet season ranged between 1 and 11 meters). The water influx to the reservoirs is confined mainly to run-off rain water in the wet season (July to August). In the remaining months of the year there is no inflow to compensate for the strong drawdown caused by evaporation and water usage to supply agriculture. Mean lake depth was 1.5 m and 3.4 m in the dry and wet season respectively. In the dry season, three of the 32 reservoirs (Hashenge, Gum Selasa and Mai Leba) had dried out completely, whereas others became very shallow (Mai Delle, Adi Kenafiz, Mai Gassa II, Hizaeti Wedicheber, Adi Gela). Reservoir trophic state varied from mesotrophic to hypereutrophic with total phosphorus concentrations measured in the wet season varying between 11 and 221 μ g P L⁻¹, whereas total nitrogen ranged between 230 and 2400 μ g N L⁻¹.



Figure 2.1 Map with the geographic location of the sampled reservoirs in Tigray, North Ethiopia.

Sampling methodology

The thirty two reservoirs were sampled twice; once at the end of the wet season (September 2004) and once near the end of the dry season (April 2005). The same sampling procedure was followed for all reservoirs (see Dejenie et al., 2008). Pelagic water samples were taken from a boat at a fixed location in the middle of the reservoir from three depths (surface, middle, and near to the bottom) using a 3-1 Heart valve sampler and pooled. The pooled samples were transported to the shores immediately after their retrieval and 250 ml sub-samples were fixed with acid Lugol's solution to 0.1 % final concentration for phytoplankton counting. The fixed samples were kept cool during transport and stored in the dark till their analysis. Conductivity, dissolved oxygen, pH and temperature were measured *in situ* using a portable meter WTW Multi 340 I electrode. Conductivity was measured for the pooled samples, but dissolved oxygen, pH and temperature were recorded at the three depths. Depth was estimated with a pre-labeled rope and transparency was determined using a Snell's tube.

The concentration of chlorophyll a was measured in the field with a fluorometer (Turner Aquafluor; average of three measurements).

Zooplankton samples were collected with a Schindler-Patalas plankton trap (12L). Counting was done with a stereoscopic microscope (Dejenie et al., 2008). Individual counts were used to calculate densities (individuals per liter) and biomass. Suspended matter was determined as the dry weight of particles retained on a pre-weighed Whatman GF/C filters after filtration of depth-integrated water samples. Water samples for nutrient analysis were taken to the laboratory in a cool box with ice cubes and frozen at -18°C until further analysis. The concentrations of total phosphorus and total nitrogen were measured following the ascorbic acid method and the kjeldahl method, respectively (Anderson & Ingram, 1989). The average values and ranges for the environmental parameters are listed in Table 2.1.

Phytoplankton was enumerated using an inverted microscope (Lund, 1958, Wetzel & Likens, 2000). Bengal Rose B dye solution was added to facilitate distinction between detritus particles and living phytoplankton cells. The surface of the sedimentation chamber was scanned during enumeration and absolute phytoplankton abundance in the samples was calculated. Cells as well as colonies (filaments, coenobia) were considered as a single 'unit' during the counts. At least 200 units were counted for each sample analyzed at 200X magnification. Identification of the taxa was made to the genus level following standard keys (John, 2005, Komarek & Anagnostidis, 2000, 2005). For biovolume measurements, 15-50 units of each taxon were measured in each sample during the counts. If less than 15 units of a given species were encountered during the counts all units encountered were measured. Biomass estimates were obtained by using published biovolume calculations (Hillebrand et. al, 1999) and conversion of biovolumes to carbon biomass following Menden-Duer and Lessard (2000).

		Wet season (2004)	Dry season (2005)
Variable	Code	Mean (range)	Mean (range)
Altitude (m.a.s.l.)	ALTITUDE	2235 (1833-2747)	
Mean depth (m)	DEPTH	3.4 (0.5-10.3)	1.5 (0.2-7.9)
Surface area (ha)	AREA	13.1 (1.8 - 45.4)	
рН	рН	8.46 (7.47-9.30)	8.66 (7.45-9.58)
Conductivity (μ S cm ⁻¹)	COND	203 (75-471)	283 (135-824)
Dissolved Oxygen (%)	02	109 (82 -162)	153 (84-304)
Temperature (°C)	TEMP	19.4 (15.1-21.9)	20.9 (14.4-29.9)
Snell's tube transparency (cm)	TRANSP	16.26 (3.00-40.00)	14.23 (3.00-40.00)
Total phosphorus (µg l ⁻¹)	ТР	52 (11-221)	
Total nitrogen (µg l ⁻¹)	TN	850 (232-2411)	
Total nitrogen to total phosphorus ratio	TN/TP	24 (6-74)	
Suspended matter (mg l ⁻¹)	SM	42 (6-737)	61 (6-596)
Chlorophyll- <i>a</i> (μ g l ⁻¹)	Chla	187 (36-773)	204 (12-1716)
Zooplankton (µg l ⁻¹ dry weight)	ZOOPB	295 (9-1603)	722 (8-4137)
Copepods ($\mu g l^{-1}$ dry weight)	COPB	144 (2-683)	120 (2-755)
<i>Daphnia</i> (μ g l ⁻¹ dry weight)	DAPHB	18 (0-254)	117 (0-1597)
Fish (CPUE; kg net ⁻¹ day ⁻¹ fresh weight)	FISH	2.4 (0-24.6)	
Fish eating birds	BIRD/PELIC	+/-	+/-
Cattle trampling (as derived from the number			
of cattle dung along the shoreline) (number m ⁻²)	CATTLE	26.4 (6.4-79.2)	41.8 (5-104.8)
Submerged vegetation cover (% cover)	VEG	6 (0-50)	9 (0-50)

Table 2.1 List of environmental variables measured for the reservoirs and their abbreviations with the average value and range

Statistical analysis

Multivariate statistical analyses were performed with CANOCO version 4.5 for Windows (Biometrics, Plant Research International, Wageningen, The Netherlands), except the forward selection of environmental variables, which was done using the package "packfor" in R (available from <u>https://r-forge.r-project.org/R/?group_id=195</u> and <u>http://www.r-project.org/</u> respectively). Absolute abundance data were used for the phytoplankton. Before multivariate analysis, phytoplankton genera occurring in a single pond were excluded and phytoplankton abundances and environmental variables (except pH, ZX1 and ZX2, see below) were log(x+1) transformed to reduce the effect of high values (ter Braak and Smilauer, 1998).

The variation in the phytoplankton communities was visualized with principal component analysis (PCA) for both the rainy and the dry season. Next, associations between the local environmental conditions (related to the aquatic food web structure and abiotic conditions) of the reservoirs and phytoplankton community structure were identified using Redundancy Analysis (RDA). In RDA, a forward selection procedure was used. To avoid inflation of the Type I error due to a large number of environmental variables, an a priori selection of environmental variables was made, and the significance of a global model including all variables checked before applying the forward selection procedure (Blanchet et al., 2008). We additionally checked whether the R²_{adjusted} of the forward selection model didn't exceed that of the full model. The following variables related to the aquatic food web structure were selected *a priori*; ZX1 and ZX2 (the two first axes of a PCA of the log(x+1)transformed absolute biomasses of the different zooplankton taxa, representing the zooplankton communities, taken from Dejenie et al., 2008), FISH (total fish biomass), VEG (the percentage cover by submerged macrophytes), and TRANSP (water clarity). Also CATTLE (as derived from the number of cattle dung along the shoreline) was selected, as the presence of cattle can have a marked influence on the phytoplankton communities of small water bodies (Villena-Alvarez et al. unpubl.). Potentially important abiotic variables included ALTITUDE (Altitude), DEPTH (average depth), TN (total nitrogen concentration), TP (total phosphorus), TN/TP, pH and COND (conductivity). Due to missing values for ZX1 and ZX2, the reservoirs ADIGEL, HIZWED and MAIGAS2 were excluded from the RDA in the dry season. The statistical significance in RDA was assessed by Monte-Carlo permutation tests (999 permutations). The contribution of the spatial configuration of the reservoirs to the phytoplankton community structure, and its (in)dependency of spatially correlated environmental variation, was also identified using RDA in a variation partitioning approach (Borcard et al., 1992). Spatial variables used for the RDA were obtained using the principal coordinates of neighboring matrix (PCNM) approach, in which principal coordinate analysis of a truncated Euclidean distance matrix is applied to construct eigenvectors, corresponding to positive Eigen values, which are then used as spatial descriptors (Borcard & Legendre, 2002, Borcard et al., 2004, Dray et al., 2006). The same procedure for the forward selection of variables was applied as described above for the environmental variables.

A simple Mantel test (Mantel, 1967; Mantel and Valand, 1970) was used to investigate the relation between the phytoplankton communities of the reservoirs in the wet season and the dry season. The test was carried out using the Zt software tool (Bonnet and Van de Peer, 2002) using the Bray-Curtis similarity matrices of the phytoplankton community data constructed in PRIMER).

Regressions and Pearson and Spearman Rank correlations were carried out with Statistica version 6.0 for Windows (StatSoft Inc., Tulsa, USA). A step-wise forward multiple regression approach was applied to identify environmental factors determining *Microcystis* biomass. Dry and rainy season data from the same reservoir were considered independent because of the fast turn-over of phytoplankton populations (e. g. Sommer et al., 1986, Chambouvet et al., 2008). Environmental variables included are related to (1) temperature

(TEMP, ALTITUDE), (2) nutrient level (TN, TP, TN/TP), (3) phytoplankton productivity (pH, Chla), (4) the clear-turbid alternative stable states in shallow lakes (Daph, TRANSP, VEG, FISH) and (5) water column stability (Season, DEPTH, COND). The same procedure for the forward selection of variables was applied as described above for the redundancy analyses (Blanchet et al., 2008).

Results

Phytoplankton community structure

27 out of 32 reservoirs in the wet season and 27 out of 29 reservoirs in the dry season were dominated by a single taxon (Fig. 2.2). Cyanobacteria dominated the phytoplankton biomass in 11 of the 32 reservoirs sampled in the wet season. *Microcystis* was the dominant taxon in ten of these reservoirs. Chlorophytes and Euglenophytes each dominated six reservoirs, and Dinoflagellates five. The dominance of the phytoplankton taxa shifted to dinoflagellates in the dry season; *Peridinium* dominated ten of the 29 reservoirs. Cyanobacteria were dominant in another ten reservoirs with *Microcystis* being the dominant genus in seven of them. No significant correlation was found between the phytoplankton community composition of the two season samples (simple Mantel test, r = -0.021, p = 0.442). This may be because of a high and inconsistent turn-over of phytoplankton communities in the reservoirs. The Shannon-Wiener diversity values of phytoplankton communities with low values of Shannon-Wiener diversity index in both the wet (on average 0.829, range 0.026 - 1.9) and dry (0.529, 0.027 - 1.469) seasons.



Fig.2.2 Shannon-Wiener diversity values of phytoplankton communities in the wet (a) and dry (b) seasons of the shallow reservoirs in a semi-arid subtropical region (Tigray, North Ethiopia). Genera behind bars are the dominant taxa and figures in parenthesis represent the % contribution to the total phytoplankton biomass in that particular reservoir and sampling.

The first axis of a PCA of the phytoplankton communities in the wet season explained 20% of the total variation and separated communities characterized by euglenophytes, most cyanobacterial, and some green algal genera (*Scenedesmus, Coelastrum, Pediastrum*) from the other phytoplankton communities (Fig. 2.3a). The second axis explained 15% and was mainly associated with *Microcystis* (Pearson correlation, $R^2 = 0.65$, p < 0.0001). In the dry

season, the first axis explained 25% and was mainly associated with *Peridinium*, several euglenophyte genera, and *Cryptomonas* (Fig. 2.3b). The second axis (explaining 17%) was, similar to the situation in the rainy season, mainly associated with *Microcystis* ($R^2 = 0.62$, p = 0.0003).



Figure 2.3 Biplot of PCA analysis performed on the phytoplankton communities in the wet (a) and dry (b) seasons. Environmental variables are plotted as supplementary variables. Variables significantly explaining phytoplankton variation as indicated by an RDA analysis are in black, other variables in gray. Filled circles represent the sampled reservoirs.

Structuring factors

The spatial configuration of the reservoirs was apparently unimportant for the phytoplankton community structure as the full RDA model including all PCNM eigenvectors was insignificant for both seasons (wet season, trace = 0.502, p = 0.092; dry season, trace = 0.191, p = 0.33). The full models remained insignificant when only the first five PCNM eigenvectors were selected *a priori*.

The full RDA model with all environmental variables included was highly significant for both seasons (Table 2.2), permitting a forward selection procedure to identify the most important environmental variables. Environmental variables selected for the RDA in the wet season were zooplankton community structure (ZX1) and pH (Table 2.2, Fig. 2.3a). ZX1 was correlated with the first axis of the phytoplankton PCA ($R^2 = 0.16$, p = 0.025), while pH was correlated with the second axis ($R^2 = 0.38$, p = 0.0002), associated with *Microcystis*. A scatter plot of the correlation between *Microcystis* biomass and pH in the rainy season is shown in Fig. 2.5. In the dry season, average reservoir depth (DEPTH) and conductivity (COND) were selected (Table 2.2, Fig. 2.3b). Both variables were strongly correlated with the first axis of the phytoplankton PCA ($R^2 = 0.67$, p < 0.0001 and $R^2 = -0.68$, p < 0.0001 respectively), which is mainly associated with *Peridinium* (Fig. 2.3b). A scatter plot of the correlation between *Peridinium* and reservoir depth is shown in Fig. 2.4.



Figure 2.4 Scatter plots of pH versus *Microcystis* in the rainy season (left) and reservoir depth versus *Peridinium* in the dry season (right) in reservoirs in Tigray, Ethiopia. Both correlations are significant at p < 0.01 as shown by non-parametric Spearman Rank tests.

Table 2.2 Environmental variables significantly explaining variation in the phytoplankton communities in the rainy and dry season respectively, as identified by a forward selection Redundancy Analysis. "All" indicates the full model with all variables included, the other variables were selected by a forward selection procedure.

Season	Variable	R ²	(cumulative) adjusted R^2	F	<i>p</i> -value
Wet season	All	0.504	0.146	1.41	0.006
	ZX1	0.070	0.039	2.26	0.017
	рН	0.053	0.063	1.76	0.045
Dry season	All	0.593	0.152	1.46	0.009
	COND	0.120	0.083	3.26	0.001
	DEPTH	0.080	0.130	2.30	0.003

Microcystis

Microcystis was the single most dominant cyanobacterium in both the wet and dry season. Results of multiple regressions on the different classes of explanatory variables which can potentially influence *Microcystis* blooms only showed a positive association of *Microcystis* with phytoplankton productivity and more in particular pH (Table 2.3). The other factors tested, being temperature, nutrients, the clear-water versus turbid stable state, and water column stability appeared unimportant. Pearson correlation analysis also showed, apart from a highly significant correlation with pH ($R^2 = 0.12$, p = 0.0060), some trend for a negative correlation of *Microcystis* with *Daphnia* biomass ($R^2 = -0.065$, p = 0.047), but this variable wasn't included in the multiple regression analysis due to an insignificant total clear-turbid model (Table 2.3).

Table 2.3 Multiple regression results for factors potentially influencing *Microcystis* biomass. In the selection model, only variables having a *P*-value < 0.05 after a forward stepwise selection procedure were included. (+) positive correlation, (-) negative correlation.

	Model	Variables included	R ²	R^2_{adj}	Р
Temperature	All variables	TEMP, ALTITUDE	0.05	0.02	0.221
Nutrients	All variables	TN, TP, TN/TP	0.02	-0.03	0.785
Phytoplankton productivity	All variables	pH, Chla	0.14	0.11	0.011
	Selection	pH (+)	0.12	0.11	0.006
Clear/Turbid	All variables	Daph, TRANSP, VEG, FISH	0.12	0.06	0.11
Water column stability	All variables	SEASON, DEPTH, COND	0.04	0.00	0.470

Discussion

The phytoplankton assemblage of the study reservoirs was generally composed of taxa associated with meso- to hypertrophic freshwater systems. In eutrophic lakes cyanobacteria are favored relative to other phytoplankton (Tilzer, 1987, Reynolds, 1998). They are among

the phytoplankton assemblages that are considered eutrophic indicators. Several studies indicate perennial dominance of phytoplankton communities by cyanobacteria in eutrophic subtropical lakes and the formation of *Microcystis* water blooms over a range of temperatures in all seasons of the year in the tropics (Sommer et al., 1986, Zohary & Robarts, 1989, Sivonen & Jones, 1999, Akin-Oriola, 2003). Under conditions of nutrient enrichment, cyanobacteria are known to proliferate and form noxious blooms in freshwater environments (Reynolds, 1984). Overall, our observations show that colonial cyanobacteria (Microcystis) dominated in most of the cyanobacteria dominated lakes. This tends to defy the general hypothesis that colony forming species are more commonly dominant in deeper lakes with a stable water column and that long-term dominance by filamentous cyanobacteria is related to shallow lake depth (Schreurs, 1992, Scheffer, 1998). Other dominant phytoplankton groups, chlorophytes and euglenophytes are also associated with eutrophic conditions, and the last group more specifically with organically enriched shallow ponds and lakes (Reynolds, 1984). This agrees well with the previous characterization of the reservoirs as eutrophic based on nutrient concentrations (Dejenie et al. 2008). A few of the shallowest reservoirs in the dry season were dominated by the diatom Nitzschia, which probably has a tychoplanktonic existence in these extremely shallow waters.

An exception is the dominance of *Peridinium* in a large number of reservoirs, especially in the dry season. The RDA for the dry season samples showed a positive correlation of *Peridinium* with depth; while it was negatively correlated with conductivity. *Peridinium* tended to dominate relatively deep and clear waters. Reynolds (1984, 1998) has put *Peridinium* in the oligotrophic to mesotrophic phytoplankton assemblages. This indicates that the deeper reservoirs in Tigray might be less affected by eutrophication, probably due to a lower recruitment of nutrients from the sediment due to less mixing.

The local phytoplankton diversity of the reservoirs in Tigray is low with most reservoirs being dominated by a single genus of cyanobacteria (mostly *Microcystis*), chlorophytes, euglenophytes, cryptophytes or dinophytes. Low diversity environments normally correspond to Shannon-Wiener indices lower than 2.5 (Margalef, 1972, May, 1975). Lower species diversity and richness is associated with eutrophication of freshwater ecosystems (Huszar et al., 1998, Leibold, 1999, Dudgeon et al., 2006, Lo et al., 2008). The shallow reservoirs in Tigray are eutrophic in nature (Dejenie et al., 2008). A change of trophic state towards hypertrophy has been reported to lead to a reduction of species diversity (Huszar et al., 1998, Lo et al., 2008). Cyanobacteria are known to proliferate and form noxious blooms in freshwater environments under nutrient enriched conditions (Reynolds 1984).

The phytoplankton community data show no significant correlation between the data of the two seasons (correlation tested for the 29 reservoirs sampled twice). This may be due to the variation observed in lake depth (see Dejenie et al., 2008). Different reservoirs may respond differently to the environmental changes they face during the dry season. The difference in the pattern of the cyanobacterial blooms is an indication of the response of the reservoirs to the environmental changes. Seasonal monitoring of a subset of the 32 reservoirs studied here also showed the varied nature of the reservoirs (Tsehaye et al., unpublished, Chapter 3 of this thesis). The two high-altitude reservoirs (Ruba Feleg and Teghane) seem to be distinct from each other and from the remaining reservoirs. Whereas Teghane was dominated by cyanophytes (*Anabaenopsis, Aphanizomenon,* and *Merismopedium*), cryptomonads (mainly *Cryptomonas*), and green algae (like *Scenedesmus*), Ruba Feleg was mainly dominated by *Peridinium* and *Cosmarium*. The major determining factor for the observed difference may be the effect of depth as Ruba Feleg is much deeper than Teghane, a factor favouring *Peridinium*.

No effects of the spatial configuration of the reservoirs on phytoplankton communities were observed despite the recent construction of these reservoirs in a region with a historical low density of freshwater bodies; only local environmental variables significantly explain the phytoplankton community composition. Traditionally, studies on the phytoplankton communities of ponds and shallow lakes have emphasized the role of local environmental factors (Leibold, 1999; Dodson et al., 2000; Jeppesen *et al.*, 2000). Recently regional processes have been considered in studies of the phytoplankton metacommunities (Soininen et al., 2007, Vanormelingen et al., 2009). Structuring roles from local and regional factors on phytoplankton and zooplankton communities was reported by Soininen et al. (2007), while others report no structuring role (or only marginal impact) of spatial distance (Van der Gucht et al., 2007, Vanormelingen et al., 2008).

Top-down factors may be important in determining the phytoplankton community composition in the semi-arid subtropical reservoirs in Tigray, North Ethiopia. The phytoplankton community composition showed a correlation with the zooplankton community structure in the wet season. Zooplankton community structure in Tigray reservoirs is mainly associated with an altitudinal gradient in nutrient levels and chlorophyll a concentrations (Dejenie et al. 2008). While this productivity gradient may also directly influence phytoplankton community structure, the selection of zooplankton community structure in the RDA suggests that zooplankton causes changes in phytoplankton community structure with altitude. However, zooplankton might be unable to also control phytoplankton biomass, as suggested by the increasing chlorophyll a concentration with increasing nutrient concentrations (Dejenie et al., 2008). The impact of predators on freshwater plankton community structure and productivity has been reported in many studies (Carpenter, 1988, Arner et al., 1998). In many temperate lakes, top-down control of phytoplankton by large herbivorous zooplankton is reported from many studies to cause spring clear-water phase (Lampert et al., 1986, Sarnelle, 1993, Hanson & Butler, 1994, Jurgens & Stolpe, 1995). In contrast, studies on tropical lakes indicate that chances for an effective top-down control over phytoplankton are lower than in temperate lakes (Fernando, 1994, Lazzaro, 1997). Reports of bottom-up regulation of phytoplankton biomass exist from both field observation and enclosure experiments (Havens, 2002). Zooplankton species in tropical lakes and ponds are generally much smaller than in temperate zones and large herbivorous zooplankton of the genus Daphnia seems to be rare or absent at lower latitudes (Hebert, 1978, Foran, 1986, Dumont, 1994b, Gillooly & Dodson, 2000, Brucet et al., 2005, Gyllstrom et al., 2005). Many tropical fish species reproduce throughout the year, and this can reduce the chances for topdown control of phytoplankton by Daphnia. Continuous reproduction leads to permanent presence of planktivorous young individuals, and thus a continuously high predation pressure on large zooplankton (Van Leeuwen et al., 2007). Our results, however, showed the presence of the large-bodied filter feeder, Daphnia especially in the dry season. This is surprising for a tropical region, and might be related to the absence of fully developed fish communities. The reservoirs in Tigray mainly host Garra, a small opportunistic and mainly benthivorous fish (Dejenie et al., 2008). Stomach analysis indicates that the fish do consume zooplankton but also seem to rely strongly on detritus (Mekonnen et al., in prep.). Garra may thus act as a nutrient pump through direct resuspension of sediments or the consumption of benthic resources which are then recycled in the water column, thus affecting water chemistry (Persson & Svensson, 2006). On the other hand, also bottom-up effects of phytoplankton on zooplankton are possible, and may (partly) explain the significant correlation between phytoplankton and zooplankton community structure. For instance, Daphnia is usually replaced by small-sized cladocerans and copepods when cyanobacterial blooms occur in eutrophic lakes (Jarvis, 1986, Paerl, 1988, Guo & Xie, 2006). We observed a similar shift in

zooplankton community composition in a separate enclosure experiment (Tsehaye et al., unpublished, Chapter 4 of this thesis).

The abundance of submerged macrophytes is a key variable in shallow lakes, affecting important functions like nutrient cycling, phytoplankton productivity and a wide range of food web interaction (Scheffer et al., 1993, Jeppensen et al., 1997, Scheffer, 1998, Bicudo et al., 2007). In both seasons we observed no important association of macrophytes in structuring the phytoplankton community composition. In shallow temperate lakes, submerged plants often provide refuge for pelagic zooplankton against fish predation, a mechanism with potential strong cascading effects on water transparency and on the entire ecosystem. In (sub)tropical lakes, however, the interaction between aquatic plants and predation may be more complex, particularly because fish density is high within the plant beds in such systems (Schriver et al, 1995, Meerhoff et al, 2003, Meerhoff et al., 2006).

Microcystis occurrence showed a significant correlation with pH. There is a controversy on the relationship of phytoplankton and pH; whether pH is a cause or a consequence of blooms. Shapiro (1997) supported the hypothesis forwarded by King (1972) which describes that blue-greens do well at high pH because that is when free CO_2 concentrations are sufficient for them but less sufficient for other groups. Shapiro (1997) also stated that cyanobacteria are the cause for low CO_2 concentrations by their photosynthetic use. Rapid withdrawal of CO_2 by algal growth leads to raised pH levels (Reynolds, 1998). The highest nutrient levels are encountered at higher altitudes, where *Microcystis* bloom occurrence might be less likely due to a lower temperature, although we failed to find significant altitude effects on *Microcystis* biomass. *Microcystis* could have taken the advantage of low water transparency as altitude was negatively correlated with water column transparency (see Dejenie et al., 2008). Success of *Microcystis* in aquatic systems is attributed to the passive mechanism of positive buoyancy which allows maximal exploitation of light

and gives a significant growth advantage in eutrophic waters (Humphries and Lyne, 1998, Dokulil & Teubner, 2000). The decrease of mean annual water temperature recorded with altitude might also decrease the chance of *Microcystis* bloom formation. *Microcystis* is reported to form blooms at higher temperatures (Robarts & Zohary, 1987). Effects of temperature and altitude on *Microcystis* abundance might be more pronounced during the winter months, when temperature is lowest, than in the months of sampling however. A seasonal study of the same reservoirs showed that high *Microcystis* biomasses were absent in the winter period, although little consistent temporal variation was observed in cyanobacterial abundance due to large differences among reservoirs (Tsehaye et al., unpublished; Chapter 3 of this dissertation).

Microcystis also showed a negative correlation with *Daphnia* biomass. A similar finding was reported by Dejenie et al. (2008) for total cyanobacterial biomass. Cladocerans, especially species of the genus *Daphnia*, are the most efficient grazers under eutrophic conditions (Lampert, 1987). Together with the trend for a positive correlation with fish biomass, the correlation of *Microcystis* with *Daphnia* may indicate a top-down control of *Microcystis* by *Daphnia*. Alternatively, the relationship may indicate the reverse, *Daphnia* populations suppressed by blooms of cyanobacteria (Gliwicz & Lampert, 1990, Deng, 2008). From our field observations it is not clear which mechanism is responsible, but results of an enclosure experiments with contrasting zooplankton treatments suggest *Microcystis* suppresses *Daphnia* (Tsehaye et al., unpublished, Chapter 6 of this thesis). This is in line with reports from both field and experimental studies (Deng et al., 2008), as seasonal and long-term changes in community structure of crustacean zooplankton in lakes have been reported to result from cyanobacterial blooms.

Acknowledgements

This study is part of the "Aquatic Ecology" project of the VLIR IUC scientific cooperation program between several Flemish universities and Mekelle University. We thank the Flemish government for the financial support of this program, and many people who joined us in the field or contributed to solve logistic issues for their help.

References

- Akin-Oriola, G.A., 2003. On the phytoplankton of Awab reservoir, Ibadan, Nigeria. *Rev. Biol. Trop.*, **51**: 99-106
- Araoye, P.A., 2002. Man-made lakes, ecological studies and conservation needs in Nigeria *Rev. Biol. Trop.*, **50** : 857-864
- Arner, M., S. Kaivisto, J. Norberg and N. Kautsky, 1998. Trophic interactions in rockpool food webs: regulation of zooplankton and phytoplankton by *Notonecta* and *Daphnia*. *Freshwater Biology*, **39**: 79-90
- Baxter R.M. and R.B. Wood, 1965. Studies on stratification in the Bisheftu crater lakes. Journal of Applied Ecology, 2: 403-417
- Bicudo D.D.C., B.M. Fonseca, L.M. Bini, L.O. Crossetti, C.E.D.M. Bicudo, and T. Araujo-Jesus, 2007. Undesirable side-effects of water hyacinth control in a shallow tropical reservoir. *Freshwater Biology*, **52**: 1120-1133.
- Blanchet F.G., Legendre P., and D. Borcard, 2008. Forward selection fo explanatory variables. *Ecology*, **89:** 2623-2632.
- Bonnet E., and Y. Van de Peer, 2002. Zt: a software tool for simple and partial mantel tests. *J. Statist. Software*, **7**: 1-12

- Borcard, D., P. Legendre, and P. Drapeau, 1992. Partialling out the spatial component of ecological variation. *Ecology*, **73**: 1045-1055
- Borcard, D. &P. Legendre, 2002. All-scale spatial analysis of ecological data by means of principal coordinates of neighbor matrices. *Ecological Modelling*, **153**: 51-68
- Borcard, D., P. Legendre, C. Avois-Jacquet, and H. Tuomisto, 2004. Dissecting the spatial structure of ecological data at multiple spatial scales. *Ecology*, **85**: 1826-1832
- Brucet,S., D. Boix, R. López-Flores, A. Badosa, R. Moreno-Amich and X.D. Quintana, 2005.
 Zooplankton structure and dynamics in permanent and temporary Mediterranean salt marshes: taxon-based and size-based approaches. *Archiv für Hidrobiologie*, 162: 535–555
- Carpenter S.R., 1988. Complex Interactions in Lake Communities. Springer-Verlag, New York.
- Chambouvet, A., P. Morin, D. Marie, and L. Guillou, 2008. Control of marine dinoflagellate blooms by serial parasite killers. *Science*, **322**:1254-1257
- Chorus, I. and J. Bartram (eds.), 1999. Toxic cyanobacteria in water: a guide to their public health consequences, monitoring and management. St Edmundsbury Press, London.
- Codd, A, J. Lindsay, F.M. Young, L.F. Morrison and J.S. Metcalf. Harmful Cyanobacteria: from mass mortalities to management measures. In: Huisman, J., H.C.P. Matthijs, , and P.M. Visser (eds), 2005. *Harmful Cyanobacteria*, Springer, The Netherlands, pp. 1–23
- Dejenie, T., Tsehaye, A., L. De Meester, Afework, M., Abreha, G., S. Risch, A. Pals, K. van der Gucht, W. Vyverman, J. Nyssen, J. deckers, and S. Declerc, 2008. Limnological and ecological characteristics of tropical highland reservoirs in Tigray, Northern Ethiopia. *Hydrobiologia*, 610: 193 – 209

- Deng, D., P. Xie, Q. Zhou, H; Yang, L. Guo, and H. Geng, 2008. Field and experimental studies on the combined impacts of cyanobacterial blooms and small algae on crustacean zooplankton in a large, eutrophic, subtropical, Chinese lake. *Limnology*, 9:1–11
- Dodson, S.I., S.E. Arnott and C.L. Cottingham, 2000. The relationship in lake communities between primary productivity and species richness. *Ecology*, **81**: 2662–2679
- Dokulil, M.T. and K. Teubner, 2000. Cyanobacterial dominance in lakes. *Hydrobiologia.*, **438**: 1-12
- Dray, S., P. Legendre, and P. Peres-Neto, 2006. Spatial modelling: a comprehensive framework for principal coordinate analysis of neighbour matrices (PCNM). *Ecological Modelling*, **196**:483-493
- Dudgeon, D., A.H. Arthington, M.O. Gessner, Z.-I. Kawabata, D.J. Knowler, C. Leveque,
 R.J. Naiman, A.-H. Prieur-Richard, D. Soto, M.L.J. Stiassny and C.A. Sullivan, 2006.
 Freshwater biodiversity: importance, threats, status and conservation challenges. *Biological Reviews*, 81: 163-182
- Dumont, H. J., 1994a. Ancient lakes have simplified pelagic food webs. Arch. *Hydrobiolgia* (suppl.), 44: 223-234
- Dumont, H.J., 1994b. On the diversity of the cladocera in the tropics. *Hydrobiologia*, **272**, 27–38
- Elizabeth, K. and Amha, B., 1994. Species composition and phytoplankton biomass in a tropical African lake (Lake Awass, Ethiopia). *Hydrobiologia*, **288**: 1-32
- Elizabeth, K. 1996. Phytoplankton in a Salinity-Alkalinity Series of lakes in the Ethiopian Rift Valley. Ph.D. thesis, Uppsala University, Uppsala, Sweden.

- Elizabeth, K. and E. Willén, 1996. *Anabaenopsis abijatae*, a new cyanophyte from Lake Abijata, an alkaline, saline lake in the Ethiopian RiftValley. *Algol. Stud.*, **80**, 1-8
- Elizabeth, K., Getachew, T., W.D. Taylor and Zinabu G.-M., 1992. Eutrophication of Lake Hayq in the Ethiopian Highlands. *Journal of Plankton Research*, **14**: 1473-1482
- Fernando, C.H., 1994. Zooplankton, fish and fisheries in tropical freshwaters. *Hydrobiologia*,272: 105–123
- Foran, J.A., 1986. The relationship between temperature, competition and the potential for colonization of a subtropical pond by *Daphnia magna*. *Hydrobiologia*, *134*: 103–112
- Gillooly, J.F., and S.I. Dodson, 2000. Latitudinal patterns in the size distribution and seasonal dynamics of new world, freshwater cladocerans. *Limnology and Oceanography*, 45: 22–30
- Gliwiccz, Z.M., and W. Lampert, 1990. Food threshold in *Daphnia species* in the absence and presence of blue-green filaments. *Ecology*, **71**: 691-702
- Guo, N., and P. Xie, 2006. Development of tolerance against toxic *Microcystis aeruginosa* in three cladocerans and the ecological implications. *Environmental Pollution*, 143: 513-518
- Gyllstrom, M., L.-A., Hansson, E. Jeppensen, F. Garci-Criado, E. Gross, K. Irvine, T. Kairesalo, R. Kornijow, M.R. Miracle, N. Nykanea, T. Noges, S. Romo, D. Stephen, E. Van Donk and B. Moss, 2005. The role of climate in shaping zooplankton communities of shallow lakes. *Limnol. Oceanogr.*, 50: 2008-2021
- Hanson, M.A., and M.G. Butler, 1994. Responses of plankton, turbidity, and macrophytes to biomanipulation in a shallow prairie lake. *Canadian Journal of Fisheries and Aquatic Sciences*, 51: 1180–1188

Havens, K.E, 2002. Zooplankton structure and potential food web interactions in plankton of subtropical chain of Lakes. *The Scientific World Journal*, **2**, 926-942

Hebert, P.D.N., 1978. The population biology of Daphnia. Biol. Rev. 53, 387-426.

- Hecky, R.E. and H.J. Kling, 1981. The phytoplankton and protozooplankton of the euphotic zone of Lake Tanganyika: species composition, biomass, chlorophyll content, and spatio-temporel distribution. *Limnol. Oceanogr.*, **26**: 548-564
- Hillebrand, H.L.-D., D. Durselen, D. Kirschtel, U. Pollingher, and T.Zohary. 1999. Biovolume caulculation for pelagic and benthic microalgae. *J. Phycol.* **35**: 403-424
- Huisman, J., H.C.P. Matthijs, and P.M. Visser (Eds), 2005. Harmful cyanobacteria. Springer, The Netherlands, pp241.
- Humphries, S. E. & V. D. Lyne, 1988. Cyanophyte blooms: the role of cell buoyancy. *Limnol. Oceanogr.* 33: 79-91.Jayatissa L.P., Silva E.I.L., McElhiney J., Lawton L.A. 2005.
 Occurrence of toxigenic cyanobacterial blooms in freshwaters of Sri Lanka. *Systematic and Applied Microbiology* 29: 156-164.
- Huszar, V.L.M., L.H.S. Silva, P. Domingos, M. Marinho and S. Melo, 1998. Phytoplankton species composition is more sensitive than OECD criteria to the trophic status of three Brazilian tropical lakes. *Hydrobiologia*, 369/370: 59–71
- Jarvis, A.C., 1986. Zooplankton community grazing in a hypertrophic lake (Hartbeespoort Dam, South Africa). iJournal of Plankton Research. 8: 1065-1078
- Jeppensen, E., M. Sondergaard, and K. Christofferson, 1997. The structuring role of submerged water plnats in lakes. Ecological Studies Series 131. Springer-Velag, New York

- Jeppesen, E., J.P. Jensen, M. Sondergaard, T. Lauridsen and F. Landkildehus, 2000. Trophic structure, species richness, and biodiversity in Danish lakes: Changes along a phosphorus gradient. *Freshwater Biology*, **45**: 201–218
- Jurgens, K., and G. Stolpe, 1995. Seasonal dynamics of crustacean zooplankton, heterotrophic nanoflagellates and bacteria in a shallow, Eutrophic lake. *Freshwater Biol.*, **33**: 27–38.
- John, D.M., B.A. Whitton and A.J. Brook (Eds.), 2005. The freshwater algae of the British Isles: an identification guide to freshwater and terrestrial algae. Cambridge University Press.
- Kardinaal, W.E.A., and P.M. Visser. Dynamics of cyanobacterial toxins: sources of variability in microcystin concentrations. In: Huisman, J., H.C.P. Matthijs, and P.M. Visser (eds), 2005. *Harmful Cyanobacteria*, Springer, The Netherlands, pp. 41–63
- King, D.L., 1972. Carbon limitation in sewage lagoons. In: Nutrients and Eutrophication, ASLO Special Symposia, Vol. 1 (ed. G.E. Likens), pp. 98–110. Allen Press, Lawrence, Kansas.
- Komarek, J., and K. Anagnostidis, 2000. *Cyanoprokaryota: Chroococcalrs*. Spektrum Akademischer Verlag.
- Komarek, J. and K. Anagnostidis K., 2005. *Cyanoprokaryota: Oscillatoriales*. Spektrum Academischer Verlag.
- Lampert, W., W. Fleckner, H. Rai, and B.E. Taylor, 1986. Phytoplankton control by grazing zooplankton: a study on the spring clear-water phase. *Limnology and Oceanography*, 31, 478–490
- Lampert, W., 1987. Laboratory studies on zooplankton-cyanobacteria interactions. *New Zealand Journal of Marine and Freshwater Research*, **21**: 483-490

- Lazzaro, X., 1997. Do the trophic cascade hypothesis and classical biomanipulation approaches apply to tropical lakes and reservoirs? *Verh Int. Ver. Theor. Angew. Limnol.*, **26**: 719–730
- Leibold, M.A., 1999. Biodiversity and nutrient enrichment in pond plankton communities. *Evolutionary Ecology Research*, 1: 73–95
- Lo, C., D.C. Bicudo, C.E.M. Bicudo, and L.M. Bini, 2008. Phytoplankton biodiversity changes in shallow tropical reservoir during the hypertrophication process. *Braz. J. Biol.*, **68**: 1061-1067
- Lund, J.W.G., C. Kipling, and E.D. Le Cren, 1958. The inverted microscope method of estimating algal numbers and the statistical basis of estimations by counting. *Hydrobiologia*, 11:143-170
- Mantel, N., 1967. The detection of disease clustering and a generalized regression approach. *Cancer Res.*, **27:** 209-220
- Mantel, N., & R.S. Valand, 1970. A technique of nonparametric multivariate analysis Biometrics, 26: 547-558
- Margalef, R, E., 1972. Homage to E. Hutchinson, or Why is there an upper limit to diversity? *Trans. Connect. Acad. Art Sci.*, **44:** 211–235
- May, R. M., 1975. Patterns of species abundance and diversity. In: M. L. Cody and J. L. Diamond, eds. Ecology and evolution of communities. Harvard University Press, Cambridge, Mass.
- Meerhoff, M., N. Mazzeo, B. Moss and L. Rodriguez-Gallego, 2003. The structuring role of free-floating versus submerged macrophytes in a subtropical shallow lake. *Aquatic Ecology*, **37**: 377-391

- Meerhoff, M., C. Fosalba, C. Bruzzone, N. Mazzeo, W. Noordoven and E. Jeppesen, 2005.An experimental study of habitat choice of Daphnia: plants signal danger more than refuge in subtropical lakes. *Freshwater Biology*, **51**: 1320-1330
- Melaku, M., C. Tudorancea, and R. M. Baxter, 1988. Some limnological observations on two Ethiopian hydroelectric reservoirs: Koka (Shewa administrative district) and Finchaa (Welega administratice district). *Hydrobiologia*, 157: 47-55
- Menden-Duer, S. and E.J. Lessard, 2000. Carbon to volume relationships for dinoflagellates, diatoms, and other protist plankton. *Limnology and Oceanography*, 45: 569-579
- Meijer, M.-L., I. De Boois, M. Scheffer, R. Portielje, and M. Hosper, 1999. Biomanipulation in shallow lakes in The Netherlands: an evaluation of 18 case studies.
 Hydrobiologogia, 408/409: 13 30
- Muylaert, K., S. Declerck, V. Geenens, J. Van Wichelen, H. Degans, J. Vandekerkhove, K. Van der Gucht, N. Vloemans, W. Rommens, D. Rejas, R. Urrutia, K. Sabbe, M. Gillis, K. Decleer, L. De Meester, W. Vyverman, 2003. Zooplankton, phytoplankton and the microbial food web in two turbid and two clearwater shallow lakes in Belgium. *Aquat. Ecol.*, 37: 137-150
- Paerl, H.W. and J. Huisman, 2008. Blooms like it hot. Science, 320: 57-58
- Persson, A., J.M. Svensson, 2006. Effects of benthivorous fish on biogeochemical processes in lake sediments. *Freshwater Biology*, 51: 1298-1309
- Pilkaityté, R. and A. Razinkovas, 2007. Seasonal changes in phytoplankton composition and nutrient limitation in a shallow Baltic lagoon. *Boreal Environment Research*, 12: 551-559

- Reynolds, C.S., 1984. Phytoplankton periodicity: the interactions of form, function and environmental variability. *Freshwater Bioloogy*, 14: 111 142
- Reynolds, C.S., 1998. What factors influence the species composition of phytoplankton in lakes of different trophic status? *Hydrobiologia*, **369/370**: 11 26
- Robarts, R.D. and T. Zohary, 1987. Temperature effects on photosynthetic capacity, respiration, and growth rates of bloom forming cyanobacteria. *New Zealand Journal of Marine and Freshwater Research*, **21: 391-399**
- Rondel, C., R. Arfi, D. Corbin, F. Le Bihan, E.H. Ndour, and X. Lazzaro, 2008. A cyanobacterial bloom prevents fish trophic cascades. *Freshwater Biology*, 53: 637-651
- Sarnelle, O., 1993. Herbivore effects on phytoplankton succession in a eutrophic lake. *Ecological Monographs*, **63**: 129–149
- Scheffer, M., A. H. Bakema, and F. G. Wortelboer, 1993. Megaplant a simulation-model of the dynamics of submerged plants. *Aquatic Botany*, **45**: 341-56.
- Scheffer, M., S. Rinaldi, A. Gragnani, L.R. Mur, and E.H. van Nes, 1997. On the dominance of filamentous cyanobacteria in shallow, turbid lakes. *Ecology*, **78**: 272-282
- Scheffer, M., 1998. Ecology of shallow lakes: Population and Community Biology Series 22. Chapman & Hall, London: 357 pp.
- Scheffer, M., and E.H. van Nes, 2007. Shallow lakes theory revisited: various alternative regimes driven by climate, nutrients, depth and lake size. *Hydrobiologia*, **584**: 455-466
- Schreurs, H., 1992. Cyanobacterial dominance. Relations to eutrophication and lake morphology. Doctoral thesis, Univ. Amsterdam: 198 pp.

- Schriver, P., J. BØgestrand, E. Jeppesen, and M. Søndergaard, 1995. Impact of submerged macrophytes on fish-zooplankton-phytoplankton interactions large-scale enclosure experiments in a shallow eutrophic lake. *Freshwater Biology*, 33: 255-270
- Shapiro, J. 1990. Current beliefs regarding dominance by blue-greens. The case for the importance of CO₂ and pH. *Verh. Int. Verein. Limnol.*, 24: 38-54
- Shapiro, J., 1997. The role of carbon dioxide in the initiation and maintenance of blue-green dominance in lakes. *Freshwater Biology*, 37: 370-323
- Sivonen, K. and G. Jones. Toxic ctanobacterial toxins; In Chorus, I. and J. Bartram (eds.), 1999. Toxic cyanobacteria in water: a guide to their public health consequences, monitoring and management. World Health Organization.
- Sommer, U., Z.M. Gilwicz, W. Lampert, and A. Duncan, 1986. The PEG model of seasonal succession of planktonic events in fresh waters. *Archiv für Hydrobiologie*, 106: 433-471
- Soininen, J., M. Kokocinski, S. Kotanen, and J. Heino, 2007. Naturality, niches, and determinants of plankton metacommunity structure across boreal wetland ponds. *Ecoscience*, **14**, 146-154
- Talling, J.F., 1986. The seasonality of phytoplankton in African lakes. *Hydrobiologia*, **138**: 139-160
- ter Braak, C.J.F. and P. Smilauer, 1998. CANOCO Reference Manual and user's guide to Conoco for Windows: Software for Canonical Community Ordination (version 4). Microcomputer Power, Ithaca, New York, USA, pp 352
- Tilzer, M.M., 1987. Light-dependence of photosynthesis and growth in cyanobacteria: implications for their dominance in eutrophic lakes. *New Zealand Journal of Marine and Freshwater Research*, **21**: 401-412

- Tudorancea, C., Zinabu G.-M. and Elias D. Limnology in Ethiopia. In: Wetzel. R.G. and B. Gopal, 1999 (eds.). Limnology in Developing Countires, Volume 2, pp:63-118
- Van Ginkel, C.E., 2003. A National Survey of the incidence of cyanobacterial blooms and toxin production in major impoundments. Internal Report No. N/0000/00/DEQ/0503.
 Resource Quality Services, Department of Water Affairs and Forestry. Pretoria.
- van Leeuwen, E., G. Lacerot, E.H. van Nes, L. Hemerik, and M. Scheffer, 2007. Reduced topdown control of phytoplankton in warmer climates can be explained by continuous fish reproduction. *Ecological Modelling*, **206**: 205-212
- Vanormelingen, P., K. Cottenie, E. Michels, K. Muylaert, W. Vyverman, and L. DeMeester, 2008. The relative importance of dispersal and local processes in structuring phytoplankton communities in a set of highly interconnected ponds. *Freshwater Biology*, 53:2170-2183
- Wetzel RG, Likens GE 2000. Limnological analyses, 3rd ed. Springer, New York, USA.
- Wood, R. B. and J.F. Talling, 1988. Chemical and algal relationships in a salinity series of Ehtiopian inland waters. *Hydrobiologie*, **158**: 29-67
- Zohary, T. and R.R. Robarts, 1989. Diurnal mixed layers and the long term dominance of *Microcystis aeruginosa*. In: Zurawell, R.W., H.R. Chen, J.M. Burke, and E.E. Prepas, 2005. Hepatotoxic cyanobacteria: A review of the biological importance of microcystins in freshwater environments. *Journal of Toxicology and Environmentl Health-Part B-Critical Reviews*, 8: 1-37

Appendix 2.1 Taxa list

Chlorophyta, green algae

Chlamydomonas sp. Closterium sp. Coelastrum sp. Coenococcus sp. Coenocystis sp. Cosmarium sp. Dictyosphaerium sp. Crucigenella sp. Eudorina sp. Keratococcus sp. Nephrocystium sp. Oocystis sp. Pediastrum sp. Planktosphaeria sp. Scenedesmus sp. Staurastaurum sp. Tetraspora sp. Volvox sp.

Cryptomonads

Cryptomonas sp.

Dinophyceae, dinoflagellates

Peridinium sp.

Cyanophyta, blue green Anabaena sp. Anabaena circinalis Anabaenopsis sp. Aphanizomenon sp. Pseudoanabaena sp. Merismopedia sp. Microcystis aeruginos.

Diatomophyceae, diatoms

Aulacoseira sp. Fragilaria sp. Navicula sp. Nitzschia sp.

Euglenophyceae, euglenoids

Euglena sp. Phacus sp. Strombomonassp. Trachelomonas sp.

Chapter 3

Seasonal dynamics of environmental variables and associated phytoplankton community structure in tropical semi-arid highland reservoirs of Tigray, North Ethiopia

Tsehaye Asmelash¹, Tadesse Dejenie¹, Pieter Vanormelingen², Teklit Gebregiorgis¹, Abreha Gebrekidan¹, Luc ³De Meester³, Ineke Van Gremberghe², Katleen Van der Gucht², Steven Declerck³ and Wim Vyverman²

(Unpublished manuscript)

Mekelle University, PO Box 231, Mekelle, Ethiopia

Laboratory of Aquatic Ecology and Protistology, Ghent University, Krijgslaan 281 - S8, 9000 Gent, Belgium

Laboratory of Aquatic Ecology, Katholieke Universiteit Leuven, Ch De Beriotstraat 32, B-3000 Leuven,

Belgium
Abstract

In semi-arid regions, many small man-made reservoirs have been constructed to ensure water availability for local communities, but a better knowledge of the aquatic food web in such reservoirs is necessary in order to maximize their benefits. We studied the temporal dynamics of the phytoplankton communities and associated environmental variables in eight shallow, semi-arid highland reservoirs in Tigray, North Ethiopia, during one year. Inter-annual variability was assessed by a comparison with a field survey carried out one year earlier. Statistical analysis using Friedman ANOVA and Kendall Coefficient of Concordance showed a significant temporal pattern for most environmental variables. Partial RDA showed that the main limnological changes were associated with seasonal differences in rainfall, while also water temperature differed strongly between winter and the rest of the year. Variables associated with the aquatic food web, i.e. chlorophyll a concentration, Daphnia, cladoceran and copepod biomass, and Secchi disc depth showed no or a less pronounced temporal variation, in sharp contrast to the situation in temperate water bodies. Phytoplankton biomass had two minima, one at the temperature minimum in winter and a more pronounced one during August, the month with the heaviest rainfall. Cyanobacteria seemed to have two main bloom periods, one in September-October and a more pronounced one in May-June. Seasonal variation in total phytoplankton and cyanobacterial biomass was, however, not significant, probably due to the large variation in total phytoplankton biomass among reservoirs and the near-absence of cyanobacteria in some reservoirs. Also for dinoflagellates and green algae, temporal variation was insignificant, while diatoms, green algae and cryptomonads had significantly higher biomasses in autumn and winter than in the remaining period. Despite the fact that seasonal differences in local environment and associated phytoplankton communities were significant, they were generally smaller than the more pronounced changes between the investigated reservoirs and year-to-year changes.

Introduction

In tropical (semi-)arid regions, the construction of small reservoirs is of vital importance to ensure water availability for irrigation and as drinking water for both people and their cattle. The main water source for such reservoirs is precipitation, which is seasonal in nature, as a consequence of a highly variable water input and associated water level fluctuations. Next to the dynamics of inflow and variable flushing rates, also seasonal changes in wind and temperature may alter environmental conditions. Due to their high turn-over rates, phytoplankton communities may respond rapidly to such environmental changes. Phytoplankton seasonality has been observed in several large natural lakes or reservoirs in the tropics and is usually driven by changes in hydrology, including water in- and out-put or water column structure and circulation due to seasonal changes in wind strength and direction (Talling, 1986 and references therein, Hooker and Hernandez, 1991). Despite the socio-economic importance of small reservoirs, their seasonal environmental dynamics and associated changes in the phytoplankton communities are largely unstudied, although they may differ substantially from larger water bodies.

The eutrophic nature of many (sub)tropical reservoirs and their constant high water temperature render them especially vulnerable to the occurrence of cyanobacterial blooms (Bouvy *et al.*, 2000), which are notorious for producing powerful toxins (Codd *et al.*, 2005, Huisman *et al.*, 2005). Other environmental conditions favoring the dominance of cyanobacteria over other phytoplankton include a high temperature and water column stability, which is enhanced by increased temperatures but also by reduced flushing rates (Harris and Baxter, 1996; Paerl and Huisman, 2008). Such conditions are often met in the dry season in (semi)arid tropical regions, but interannual variability in rainfall may be at least as important in determining cyanobacterial bloom occurrence (Harris and Baxter, 1996; Bouvy et al., 2000).

In Tigray, North Ethiopia, more than 70 small reservoirs were constructed to reduce problems of water scarcity in an area with a dry season lasting up to eight months. A previous field survey revealed that most reservoirs were in the eutrophic state and many were heavily impacted by cyanobacterial blooms, mainly of *Microcystis*, with chlorophytes and dinophytes as co-dominants (Dejenie et al., 2008; Tsehaye et al., unpublished, Chapter 2 of this thesis). The objective of the present study was to determine whether seasonal variation occurred in (1) limnological variables and (2) the phytoplankton communities of the highland reservoirs in Tigray, independent of among-reservoir differences. Inter-annual variability was assessed by a comparison with a field survey of phytoplankton communities conducted one year earlier (Dejenie et al., 2008, Tsehaye et al., unpublished, Chapter 2 of this thesis).

Material and methods

Study site

Tigray, the northernmost regional state of Ethiopia, is located between 12° and 15° N and $37^{\circ}10'$ and $40^{\circ}10'$ E (Figure 3.1). Eight reservoirs representing the main environmental gradients present in the region were selected for the present study from a set of 32 reservoirs characterized during a previous field survey (Dejenie *et al.*, 2008; Tsehaye et al., unpublished, Chapter 2 of this thesis). The included reservoirs were permanent or temporary and varied in various ecological characteristics such as cyanobacterial bloom occurrence, macrophyte coverage, presence of fish, and zooplankton abundance (see Table 3.1. for an overview and Dejenie *et al.*, 2008). Altitude, area, depth, turbidity and nutrient load were also considered in an effort to make the recorded seasonal variation representative for the large number of reservoirs in the region.



Figure 3.1 The geographic location of the eight reservoirs in Tigray, North Ethiopia, selected for the seasonal study of environmental variables and associated phytoplankton communities.

Sampling

Field samples from the 8 reservoirs were collected every month for an entire year, from September 2005 to August 2006. The sampling procedure was as described by Dejenie et al. (2008). Water samples, except for crustacean zooplankton, were taken at a fixed location from three depths: surface, middle (intermediate depth), and near to the bottom, using a 3 liter Heart valve water sampler and pooled. The pooled samples were transported to the shores immediately after retrieval and sub-samples were taken and fixed on the shores. At each site conductivity, dissolved oxygen, pH and temperature were measured with a portable meter WTW Multi 340 I electrode. Conductivity was measured for the pooled samples, but dissolved oxygen and pH were recorded at the three depths. Transparency was determined using a Secchi disc (diameter: 30 cm). Sub-samples for nutrient analysis were transported in a

cool box to the laboratory and stored at -18°C until analysis. Sub-samples for enumeration of phytoplankton were fixed with acid Lugol's solution to 0.1 % final concentration. Crustacean zooplankton were sampled from two pelagic sites and pooled. Twelve liters of water was taken from the surface down to near the bottom of the reservoir at an interval of one meter using a Schindler-Petalas trap (64 μ m mesh size). Samples ware preserved with sugar saturated formalin. Wind speed data were obtained from the digital database of climatological and stream flow data of Geba catchment (MU – IUC, 2007).

Table 3.1 General characteristics of the eight reservoirs selected for assessing seasonal

 dynamics of environmental variables and associated phytoplankton communities.

Reservoir	Altitude	Trophic	Cyanobacteria	Fish	Macrophyte
name	(m.a.s.l.)	state	bloom*		coverage
Gereb Awso	2283	Eutrophic	None	Present	0
Gum Selasa	2116	Eutrophic	Present	Present	25%
Mai Gassa I	2131	Mesotrophic	Present	Present	75%
Mai Gassa II	2130	Eutrophic	Present	Present	50%
Mai Leba	2231	Mesotrophic	None	Absent	0
Ruba Feleg	2747	Eutrophic	None	Present	< 25%
Tegh'ane	2735	Eutrophic	None	Absent	0
Tsinkanet	2320	Mesotrophic	None	Present	25%

* Bloom defined when cyanobacteria % biomass contribution is $\geq 65\%$

Sample analysis

Total nitrogen was analyzed with the kjeldahl method and total phosphorus was analyzed following the ascorbic acid method (Anderson & Ingram, 1989). Chlorophyll a was estimated *in situ* with a portable fluorometer (Turner Aquafluor; average of three measurements) from the pooled water samples. Zooplankton was counted with a stereoscopic microscope (Dejenie et al. 2008). Individual counts were used to calculate densities (individuals per liter) and biomass. Phytoplankton was enumerated using the inverted microscope method (Lund *et al.*, 1958, Wetzel and Likens, 2000). Bengal Rose B dye solution was added to facilitate distinction between detritus particles and phytoplankton cells. Colonies (filaments, coenobia) were considered as a single 'unit' during the counts, and counting was continued until at least 300 units were counted. Biovolume calculations were made by measuring the linear dimensions of 50 units of each taxon in a sample and fitting the different taxa to geometric forms (Hillebrand et al., 1999). Phytoplankton biomass was estimated from cell biovolume measurements using previously published biovolume-to-carbon conversion data (Menden-Deuer & Lessard, 2000).

Data analysis

Temporal patterns were investigated for two sets of variables: (1) environmental variables and (2) phytoplankton taxa. The variables included in the first data set were rainfall, total nitrogen, total phosphorus, chlorophyll a, oxygen and suspended matter concentrations, Secchi disc depth, temperature, pH, conductivity, depth, *Daphnia* and copepod biomass. The second data set consisted of the biomass of the different recorded phytoplankton genera and higher taxon groupings. The relationships between temporal change and each set of variables, independent of differences between reservoirs, were investigated using partial redundancy analysis (partial RDA). Months were used as explanatory variables and reservoirs as co-variables to partial out

the effects attributable to differences between the reservoirs. Dummy variables were constructed for the months and reservoirs, and reservoirs were assigned as 'blocks' (Lepŝ and Ŝmilauer, 2003). Significance tests were based on 999 random Monte Carlo permutations, which were restricted to blocks to account for the dependency of data based on samples from the same reservoir. The partial RDA on the environmental data set was performed with standardized limnological variables.

To determine whether temporal changes for each separate environmental or phytoplankton variable were significant, we used Friedman ANOVA, a non-parametric alternative for one-way analysis of variance, and Kendall Coefficient of Concordance. The Kendall coefficient of concordance is similar to a non-parametric Spearman Rank correlation coefficient R, except that it expresses the relationships between more than two groups of data through time (Legendre, 2004). As wind speed was only measured for a single reservoir, temporal changes in wind speed were not tested statistically. To account for the large variability in phytoplankton community composition among reservoirs, we carried out the Friedman ANOVA analysis for reservoirs permanently dominated by cyanobacteria separately.

Inter-annual variation was assessed by comparing the data collected for this study in September 2005 and April 2006 with those obtained from a field survey in September 2004 and April 2005 (see Dejenie et al. 2008 and Tsehaye et al., unpublished, Chapter 2 of this thesis). Inter-annual (among year variation) and intra-annual (among months variation) of both abiotic and biotic variables were assessed using partial RDA as described above.

Prior to analysis, all abiotic and biotic variables were log(x+1) transformed. Univariate analyses were performed with the statistical program STATISTICA 7.1 (Stat Soft Inc., Tulsa, OK, U.S.A.) and multivariate analyses with CANOCO 4.5 (ter Braak & Smilauer, 1998; Lepŝ and Ŝmilauer, 2003).

Results

Environmental variation

A partial RDA shows that there was a significant temporal pattern in the environmental variation of Tigray reservoirs (Trace = 0.305, F = 7.444, p = 0.0001). Centroids of the different months displayed a marked trajectory on the RDA biplot, with centroids of the winter, summer, spring and autumn months in different quadrants of the biplot (Fig. 3.2). The first axis of the RDA explained 16.8% of the total environmental variation and was mainly associated with variation in rainfall. It was also associated with changes in temperature, total nitrogen and suspended matter concentrations. The second axis explained a much lower proportion of the environmental variation (5.6%) and was associated with changes in suspended matter and temperature. The following picture emerged: September to February are characterized by drought, low temperature (especially December and January), and low suspended matter and total nitrogen concentrations. The period March to August is associated with rainfall and high suspended matter concentrations (especially July and August) and high temperature, pH and conductivity (especially in spring). Friedman ANOVA of the limnological variables (Table 3.2) corroborated these findings as most variables displayed highly significant temporal variation. Chlorophyll a concentration (p = 0.037) and copepod (p= 0.015) and cladoceran biomass (p = 0.025) showed significant but rather weak temporal patterns, whereas we could not observe any temporal signal for Secchi disc depth (p = 0.13) and *Daphnia* biomass (p = 0.73). The low coefficients of concordance (Table 3.2) and low scores on the first two axes of the RDA (Fig. 3.2) also indicate that they fail to show a consistent temporal pattern.



Figure 3.2 First and second axis of a partial RDA on the response of environmental variables in function of month of the year, corrected for the influence of reservoir by including reservoirs as co-variables. Environmental variables are indicated by arrows, months by uptriangles, and samples by empty circles. Broken arrows indicate the sequence of sample collection. The first and second axes explain 16.8 and 5.6% of the total environmental variation, respectively.

Table 3.2 Results of Friedman ANOVA and Kendall Coefficient of Concordance (W) calculations on the temporal dynamics of environmental variables recorded for eight reservoirs in Tigray, Ethiopia, between September 2005 and August 2006.

Variable	ANOVA	Р	W
	Chi Sqr.		
Physicochemical		L	
Total nitrogen	50.43	< 0.001	0.57
Total phosphorus	74.09	< 0.001	0.85
Secchi depth	16.17	0.14	0.18
Suspended matter	34.56	< 0.001	0.39
Chlorophyll a	20.67	< 0.04	0.24
pH	32.57	< 0.001	0.37
Conductivity	67.51	< 0.001	0.77
% O ₂	25.41	0.007	0.29
Temperature	58.75	< 0.001	0.67
Rainfall	78.52	< 0.001	0.89
Depth	75.53	< 0.001	0.86
Zooplankton			
Total	26.15	0.006	0.29
Daphnia	7.85	0.730	0.09
Cladocerans	21.96	0.025	0.25
Copepods	23.59	0.015	0.27

Month-to-month changes of the limnological variables are shown in Fig. 3.3(a-d). Rain fell between March and August with July and August showing a pronounced maximum. Associated with this is a gradual increase in conductivity from September to June and a steep decrease afterwards when the reservoirs are at full bank from the heavy rains of July and August. Suspended matter concentrations increased in June and peaked in August. Water temperature ranged between 12.1°C and 22.4°C, was minimal in December and January and increased, first fast and then gradually, towards a broad maximum in spring and early summer (Figure 3d). pH peaked at the same time but declined fast afterwards. Total phosphorus levels were high until March but then dropped dramatically, stayed relatively low for the rest of the study period and increased again in August. High total nitrogen concentrations were associated with the rainy season but started to increase in winter before the first rainfall. Concentrations were somewhat lower in May-June but increased again in July-August. Dissolved oxygen was almost always relatively high and close to saturation (on average 91%). Secchi disc depth was on average low, usually not exceeding 1 m for most of the reservoirs, except Ruba Feleg.

Phytoplankton seasonality

A total of 28 phytoplankton genera were identified. The green algae were the most diverse group, accounting for 43 % of the total taxa identified. Six genera of cyanobacteria were counted, accounting for 21 % of the total number of genera. *Microcystis* was the single most abundant taxon and accounted for about 74% of the total phytoplankton biomass recorded over the year for all the reservoirs (Fig. 3.4). Other identified genera belonged to the diatoms, cryptomonads, dinoflagellates, and euglenophytes.



Figure 3.3 Changes in environmental variables during the course of one year in eight reservoirs in Tigray, North Ethiopia. a) Rain fall, conductivity and suspended matter; b) Chlorophyll a, cladoceran zooplankton biomass and Secchi depth; c) Depth, pH and wind speed; d) Temperature, total nitrogen and total phosphorus. Data for wind speed are available for only one station.



Figure 3.4 Changes in phytoplankton biomass during the course of one year in eight reservoirs in Tigray, North Ethiopia. Total phytoplankton biomass was divided into *Microcystis* (black), the single most dominant genus, the single dinoflagellate genus *Peridinium* (white), cyanobacteria other than *Microcystis* (dark grey), and biomass of all other phytoplankton (light grey).

A highly significant temporal turn-over in phytoplankton community composition was observed by partial RDA of the phytoplankton at the order level (Trace = 0.146, F = 1.846, *p*

= 0.0005). The first axis of the RDA explained 6.9% of the total variation and, similar to the partial RDA of the environmental variables, separated the period March-August from September-February (Fig. 3.5). A similar pattern emerged from a partial RDA in which the ten most abundant phytoplankton genera were used (Trace = 0.097, F = 1.517, p = 0.0020, biplot not shown). The main difference between both periods was the higher abundance of green algae, diatoms and cryptomonads in September-February. Euglenoids appeared to be mainly associated with the second axis, which explained 3.6% of the total variation, due to a peak in abundance of Trachelomonas in July. Cyanobacteria and dinophytes did not show seasonal variation, as seen by their low scores on the first two axes of the RDA. This pattern was confirmed by the Friedman ANOVAs showing significant temporal variation for diatoms, green algae, and cryptomonads, but not for cyanobacteria, dinoflagellates and euglenoids (Table 3.3). This may result from the difference in the occurrence of cyanobacterial blooms in the different reservoirs. Also partial RDA and Friedman ANOVAs run for reservoirs permanently dominated by cyanobacteria separately could not reveal temporal pattern of cyanobacteria. At the genus level, only Cryptomonas sp. showed a significant temporal variation. This may result from an inconsistent presence of some members of a genus in different reservoirs, the variation among reservoirs overwhelming the seasonal variation. But *Cryptomonas sp.* is the only recorded genus in our study reservoirs.



Figure 3.5 Partial RDA biplot on changes in phytoplankton taxon in function of time, corrected for the influence of reservoir by including reservoirs as co-variables. Phytoplankton taxa are indicated by arrows, months by up-triangles, and samples by empty circles. The first and second axes explain 6.9% and 3.9% of the total phytoplankton variation, respectively.

Table 3.3 Results of Friedman ANOVA and Kendall Coefficient of Concordance (W)calculations on the temporal dynamics of phytoplankton communities recorded for eightreservoirs in Tigray, Ethiopia, between September 2005 and August 2006.

Taxa	ANOVA Chi Sqr.	Р	W			
By division						
Diatoms	23.078	0.02	0.26			
Green algae	19.84	0.047	0.23			
Cryptomonads	27.79	0.004	0.32			
Cyanobacteria	11.21	0.43	0.13			
Dinoflagellates	12.99	0.29	0.15			
Euglenoids	19.39	0.05	0.22			
Total phytoplankton	15.77	0.15	0.18			
By Genera						
Anabaena sp.	16.44	0.13	0.19			
Aulacoseira sp.	16.08	0.14	0.18			
Closterium sp.	8.28	0.69	0.09			
Coelastrum sp.	16.43	0.13	0.19			
Cryptomonas sp.	27.79	0.003	0.23			
Euglena sp.	12.95	0.29	0.15			
Fragilaria sp.	14.34	0.22	0.16			
Microcystis	11.33	0.42	0.13			
Peridinum sp.	12.99	0.29	0.15			
Phacus sp.	10.88	0.45	0.12			
Scenedesmus sp.	17.29	0.09	0.19			
Trachelomonas sp.	18.34	0.07	0.21			

In terms of total phytoplankton biomass, two maxima can be distinguished, the first at the beginning of the dry season (October – November) and the other at the end of the dry season (May – June) (Fig. 3.6). The two maxima are separated by two minima, one in January and the other, more pronounced one, in August. The lower value of average phytoplankton biomass in January coincides with a low water temperature, while in August it coincides with high rainfall and a high amount of suspended matter. A Friedman ANOVA failed to show significant variation in phytoplankton biomass, however (Table 3.3).



Figure 3.6 Average total phytoplankton biomass in Tigray reservoirs during the course of one year

The first axis of a partial RDA of the environmental variables (Trace = 0.311, F = 5.794, p = 0.002) and phytoplankton communities (Trace = 0.124, F = 1.63, p = 0.022) in the eight studied reservoirs in both sampling years (2004/2005 and 2005/2006) showed that the main variation in both data sets was associated with among-year differences (Fig. 3.7). Differences between seasons were associated with the second axis. Moreover, intra-annual variation in the phytoplankton communities was higher in the survey campaign 2004/2005 than in this study. The year 2004/2005 was exceptionally dry year, with lower than average spring rains, a late start of the main rainy season and low overall amount of rainfall during the



rainy season. Only August was with the typical amount of rainfall (MU – IUC, 2007).

Figure 3.7 Standardized RDA plot on environmental variables (A) and phytoplankton community (B) in function of time (months), after correction for among reservoir differences. Circles and up-triangles represent centeroids of months and year. Year 2004/2005 is represented by up-triangle while 2005/2006 is represented by circles. Solid arrows represent response variables, and small circles represent individual samples.

Discussion

The abiotic variables measured in this study showed a clear temporal pattern associated with seasonal changes in rainfall and temperature. In tropical systems, marked variations in temperature and rainfall between seasons often influence the physico-chemical characteristics of water bodies (Adebisi, 1981, Chapman and Kramer, 1991). The conductivity of reservoirs and manmade lakes depends largely on that of inflowing rivers, turnover rates and the soil of the catchment area (Payne; 1986). In our study, the observed increasing trend for conductivity during the dry season (Fig. 3.3), peaking in June just before the main rain months, indicates the role of rain and the flood that enter the reservoirs and dilute the water in the reservoirs. The same pattern of conductivity has been reported for different lakes (Elizabeth & Amaha, 1994; Pointo-Coelho, 1998; Tamuka & Brain, 2007) which may be associated with the high

evaporation rates during the dry season. The draw-down of the reservoirs as water is used for irrigation during the dry season might have increased the impact of evaporation, consequently increasing conductivity. While the increase in pH in spring is probably mainly associated with the rise in conductivity, it might additionally be influenced by an increased photosynthetic productivity associated with the increasing temperature in spring. Photosynthesis removes CO₂ from the water, thereby decreasing the concentration of carbonic acid (Wetzel, 2001). Also associated with seasonal variation in rainfall is the concentration of suspended matter, showing a maximum in the main rain months July and August. Heavy rains cause a strong run-off from the catchment resulting in an accumulation of sediment particles in the reservoirs. Associated with this run-off might also be the increase in total phosphorus (TP), as TP is tightly coupled with sediment particles (Moss, 1998), and total nitrogen concentration (TN) in July and August. Total nitrogen (TN) concentration increased at the end of the winter period and remained high in spring, although there was a small decreasing. This could be due to an increased impact of trampling of the reservoirs by cattle as a result of clearing of crops from the farmlands. Average wind speed was maximal during the winter period and this may have contributed to the winter increase in nutrient concentrations, due to an increased recruitment of nutrients from the sediment in that period. This may especially be true for phosphorus concentration, which seems to be strongly correlated with wind speed. However, to better understand the nutrient cycle in these reservoirs, we recommend collecting data over several years and studying the trend in inter- and intra-annual variability of nutrients in relation to the phytoplankton community.

Despite the fact that there were highly significant seasonal changes in abiotic environmental variables, the variables indicative of aquatic food web structure (zooplankton biomass, chlorophyll a concentrations, water clarity) showed less significant or insignificant temporal variation. This is probably caused by the fact that the seasonal changes in the limnology of these reservoirs were not large enough to cause large and consistent seasonal changes in the aquatic food web structure, contrary to the situation in temperate shallow water bodies (Kalff & Watson, 1986; Sommer et al., 1986). Phytoplankton seasonal studies on other reservoirs report minimum total biomass following the rainy seasons and heights near the end of the prolonged dry season (Kalff & Watson, 1986; Hooker & Hernandez, 1991). These studies, similar to ours and unlike reports from temperate regions, report consistent dominance by cyanobacteria. The sharp decline in phytoplankton biomass observed in August my result from the combined effect of dilution by floods coming into the reservoirs and the accompanied increased suspended matter which in-turn affects the light climate.

The taxon composition of the reservoirs in Tigray was dominated by the Chlorophyta. Reports from large tropical lakes also show a dominance of Chlorophytes in terms of diversity (Kalff & Watson, 1986; Elizabeth & Amaha, 1994). Cyanobacteria, however, dominate in terms of biomass. Blue green algae have a wide geographical distribution, although some species are specific to temperate or tropical regions. *Spirulina, Anabaenopsis and Cylidrospermopsis* species, for example, occur more frequently at lower latitudes (Gibson & Smith, 1982; Bouvy et al., 2000). Planktonic cyanobacteria, like *Microcystis*, are more commonly associated with eutrophic (Gibson & Smith, 1982) and alkaline waters (Wetzel, 1983). The reservoirs in Tigray are therefore not exceptional in having one fifth of the number of taxa and 75 % of the total phytoplankton biomass composed of cyanobacteria and *Microcystis*, respectively.

Comparing the partial RDA ordinations for the environmental and phytoplankton taxa, we can draw some conclusions with respect to the pattern of dominance of different phytoplankton taxa in relation to environmental variables. From September to February, with relatively low temperature and high wind speed, probably resulting in lower water column stability and high total phosphorus concentrations, the chlorophytes, cryptomonads, and diatoms had a higher biomass, resulting in co-dominance with cyanobacteria. Cyanobacteria (dominated by *Microcystis*) and *Peridinum* (Dinoflagellates) tended to dominate from March to August, periods characterized by higher temperature, lower TP, and higher conductivity and pH. Large blooms of Microcysis came up in May in Mai Gassa I and Gum Selassa and a less pronounced maximum was observed in Tsinkanet during that same period. These blooms reached a maximum in June-July, right before they were diluted by the heavy rainfall in August. A similar pattern was observed for another cyanobacterium, Anabaenopsis, in Teghane - one of the high altitude reservoirs. A less pronounced Microcystis bloom was detected in early autumn in Mai Gassa I and II. Despite these strong Microcystis bloom dynamics, overall temporal patterns in cyanobacterial biomass were not significant, probably due to the fact that cyanobacterial blooms only developed in five of the eight reservoirs. In the other three reservoirs, mainly high altitude reservoirs, cyanobacteria blooms were not observed. Although we can see some associations between phytoplankton taxa and environmental variables, the phytoplankton assemblages could also have been influenced by more complex interactions of physical, chemical and biological factors (e.g., zooplankton biomass and community structure).

Seasonal variation in phytoplankton community structure is less important than variation among reservoirs (see Table 3.3). This may be the result of the large difference between the reservoirs in physicochemical variables like altitude and depth. Dejenie et al. (2008) reported that reservoirs in lowlands suffer more from cyanobacteria blooms compared with high altitude reservoirs.

The comparison of the April/September data set of this study with the field survey in the same months of the preceding year showed the lack of repeatability in temporal pattern among years. The limnological variables showed both inter-annual and intra-annual variation. The variation among seasons for the limnological variables and phytoplankton community was more pronounced in 2004/2005 than in 2005/2006 (Fig. 3.7). This may be because the year 2004/5 was an exceptionally dry year. For example, of the 32 reservoirs surveyed in September 2004, eight were dry or very shallow in the subsequent dry season (April 2005; Dejenie et al., 2008). Our data show that the reservoirs are dynamic in their phytoplankton community assemblage. One of the major driving factors for this variation seems to be rainfall.

Acknowledgements

This study is part of the "Aquatic Ecology" project of the VLIR IUC scientific cooperation program between several Flemish universities and Mekelle University. We thank the Flemish government for the financial support of this program, and many people who joined us in the field or contributed to solve logistic issues for their help.

References

- Adebisi, A.A., 1981. The physico-chemical hydrology of a tropical seasonal river-Upper Ogun river. *Hydrobiology*, **79**: 157–165
- Anderson, J.M. and J.S.I Ingram, 1989. TSBF: A Handbook of Methods of Analysis. CAB International.
- Bouvy, M, D. Falcao, M. Marinho, M. Pagano, A. Moura, 2000. Occurrence of *Cylindrospermopsis* (Cyanobacteria) in 39 Brazilian tropical reservoirs during the 1998 drought. *Aquat. Microb. Ecol.* 23:13-27
- Chapman, L.J. and D.L. Kramer, 1991. Limnological observations of an intermittent tropical dry forest stream. *Hydrobiol.*, **226**: 153–166

- Codd, G.A., J. Lindsay, F.M. Young, L.F. Morrison, and J.S. Metcalf, 2005. Harmful cyanobacteria – from mass mortalities to management measures. Pp. 1-23 *In* [Huisman, J, Matthijs, HCP.
- Dejenie, T., Tsehaye, A., L. De Meester, Afework, M., Abreha, G., S. Risch, A. Pals, K. van der Gucht, W. Vyverman, J. Nyssen, J. deckers, and S. Declerc, 2008. Limnological and ecological characteristics of tropical highland reservoirs in Tigray, Northern Ethiopia. *Hydrobiologia*, 610: 193 – 209.
- Elizabeth, K. and B. Amaha, 1994. Species composition and phytoplankton biomass in a tropical African lake (Lake Awassa, Ethiopia). *Hydrobiologia*, **288**: 13-32
- Gibson, C.E., and R.V. Smith, 1982. Freshwater plankton. *In:* Carr, N. G.; Whitton, B. A. *ed.*,The biology of Cyanobacteria, pp. 463-489. Oxford, Blackwell Scientific Publications.
- Harris, G.P. and G. Baxter, G., 1996. Interannual variability in phytoplankton biomass and species composition in a subtropical reservoir. *Freshwater Biology*, **35**: 545-560
- Hillebrand, H., C.-D. Dürselen, D. Kirschtel, U. Pollingher, and T. Zohary, 1999. Biovolume calculations for pelagic and benthic microalgae. *Journal of Phycology*, **35**: 403-424
- Hooker, E., and S. Hernandez, 1991. Phytoplankton biomass in Lake Xolotlan (Managua): its seasonal and horizontal distribution. *Aquatic Ecology*, **25**: 125-131
- Huisman, J., H.C.P. Matthijs, and P.M. Visser (Eds), 2005. Harmful cyanobacteria. Springer, Berlin.
- Kalff, J. and J. Watson, J. 1986. Phytoplankton and its dynamics in two tropical lakes: a tropical and temperate zone comparizone. *Hydrobiologia*, **138**: 161 176

- Legendre, P. 2004. Species associations: the Kendall Coefficient of Concordance Revisited. Journal of Agricultural, Biological and Environmental Statistics, 10: 226 - 245
- Lepŝ, J, and P. Ŝmilauer, 2003. Multivariate analysis of ecological data using CANOCO. Cambridge University Press, Cambridge.
- Lund, J.W.G., C. Kipling, and E.D. Le Cren, 1958. The inverted microscope method of estimating algal numbers and the statistical basis of estimations by counting. *Hydrobiologia*, **11**:143-170
- Menden-Deuer, S. and E.J. Lessard, 2000. Carbon to volume relationships for dinoflagellates, diatoms, and other protist plankton. *Limnology and Oceanography*, 45: 569-579
- Moss, B., 1998. Ecology of fresh waters, man and medium. Blackwell Scientific Publications, London, pp. 557
- MU IUC, 2007. Digital database of climatological and stream flow data of Geba catchment, obtained from National Meteorological Services Agency and Ministry of Water
 Resources. VLIR (Belgium) Mekelle University Institutional University Cooperation
 Programme, Mekelle (Ethiopia) and Leuven (Belgium).
- Paerl, H.W., and J. Huisman, 2008. Blooms like it hot. Science, 320:57-58

Payne, A.I., 1986. The ecology of tropical lakes and rivers. Wiley, 301 p.

- Pointo-Coelho, RM., 1998. Effect of eutrophication on seasonal pattern of mesozooplankton in tropical reservoir: a 4-year study in Pampulha Lake, Brazil. *Freshwater Biology*, **40**: 159 – 173
- Sommer, U., Z.M. Glimicz, W. Lampert, and A. Duncan, 1986. The PEG model of seasonal succession of planktonic events in fresh waters. *Archiv für Hydrobiologie*, **106**: 433-471.

- Talling, J.F., 1986. The seasonality of phytoplankton in African lakes. Hydrobiologia. **138**: 139-160.
- Tamuka, N. and M. Brain, 2007. Water quality and plankton dynamics in two small dams in Zimbabwe. African Journal of Aquatic Science, 32: 139 - 151
- ter Braak, C.J.F., and P. Smilauer, 1998. *CANOCO Reference Manual and user's guide to Canoco for Windows: Software for Canonical Community Ordination (version 4).* Microcomputer Power, Ithaca, New York, USA, pp 352.
- Wetzel, R. G., 1983. Review of "Hypertrophic Ecosystems." Developments in Hydrobiology, Vol. 2. *Limnol. Oceanogr.*, 28:607-608
- Wetzel, R.G., 2001. Limnology. Lake and river ecosystems. Third edition, Academic Press, London.
- Wetzel, R.G., and G.E. Likens, 2000. Limnological analyses, 3rd ed. Springer, New York, USA.

Chapter 4

Bacterial community composition in semi-arid highland reservoirs of Tigray, North Ethiopia - impact of environmental factors and Microcystis blooms

Tsehaye Asmelash^{1,2}, Tadesse Dejenie³, Ineke van Gremberghe², Teklit Gebregiorgis⁵, Belay

Gebreyohannes⁶, Steven Declerck⁴, Luc De Meester⁴, Katleen Van der Gucht², Wim Vyverman²

(Unpublished manuscript)

¹ Department of Microbiology, Mekelle University, P.O Box 231, Mekelle, Ethiopia

² Laboratory of Aquatic Ecology and Protistology, Gent University, Krijgslaan 281-S8, B-9000 Gent, Belgium ³ Department of Biology, Mekelle University, P.O Box 231, Mekelle, Ethiopia

⁴ Laboratory of Aquatic Ecology and Evolutionary Biology, KULeuven, Ch. Deberiotstraat 32, 3000 Leuven, Belgium⁵ Department of Chemistry, Mekelle University, P.O Box 231, Mekelle, Ethiopia

⁶Department of biochemistry, Mekelle University, P. O. Box 231, Mekelle, Ethiopia

Abstract

We studied the taxon composition of bacterioplankton communities in semi-arid highland reservoirs in Tigray, northern Ethiopia using molecular fingerprinting (DGGE). A total of 26 reservoirs were sampled during both the wet and dry season. Abiotic factors as well as the composition and standing stock biomass of phytoplankton, zooplankton and fish were measured and were used to interpret variation in bacterial community composition. Multivariate analysis revealed that bacterial community composition was significantly explained by *Microcystis* and zooplankton (copepod) biomass in the wet season, and by nutrient concentrations, dissolved oxygen concentration, cattle and fish biomass in the dry season. Reservoirs characterised by a *Microcystis* bloom had a lower bacterial taxon richness as revealed by DGGE than reservoirs that did not harbour a *Microcystis* bloom.

Introduction

To date studies of bacterial communities in lakes and their relationship with biotic and abiotic environmental factors largely focused on temperate freshwater bodies (Eiler & Bertilsson, 2004; Allgair & Grossart, 2006; Shade et al., 2007; Van der Gucht et al., 2007; Dimitriu et al., 2008). A few studies in tropical and sub-tropical Africa focused on bacterioplankton ecology in relatively large lakes. Zenabu and Taylor (1989) reported vertical heterogeneity of heterotrophic bacterial activity during long stratification period from the Ethiopian rift-valley lake Awasa. De Wever *et al.* (2005) also studied the vertical and horizontal heterogeneity of bacterial community composition of Lake Tanganyika. Heterotrophic bacterioplankton production and mortality rates by grazing were also reported from East African lakes (Zenabu & Taylor, 1990; Pirlot et al., 2007). Bacterial grazing by predators (mainly protozoa and ciliates) was reported as the major cause of bacterial mortality in lakes Awassa and Tangnyka (Zenabu & Taylor, 1990; Pirlot et al., 2007). So far, however, there is little known on the bacterial communities inhabiting tropical and subtropical freshwater ecosystems, including the many artificial reservoirs that are constructed for irrigation in semi-arid areas.

Many freshwater lakes all over the world are affected by eutrophication. This is largely the result of high external nutrient loading associated with sewage influx and an intensification of agriculture. Especially shallow lakes are very sensitive to eutrophication, due to the intensive exchange of nutrients between the sediment and the water column (Moss, 1998; Scheffer, 1998). Because of eutrophication, blooms of cyanobacteria have become a common and recurrent phenomenon (Akin-Oriola, 2003; Chorus and Bartram, 1999; Huisman *et al.*, 2005; Zurawell *et al.*, 2005; Jayatissa *et al.*, 2006). Also in the man-made reservoirs in northern Ethiopia, cyanobacterial blooms are common. Tsehaye et al. (unpublished ms) detected dense *Miccrocystis* blooms in 10 out of 32 reservoirs in Tigray.

Cyanobacterial blooms might influence community composition of heterotrophic bacterial communities, as they are likely to affect the amount and quality of dissolved organic matter, the primary food source of bacterial communities. In addition, they often produce elevated levels of cyanotoxins and noxious allelopathic compounds which might have an inhibitory effect on the growth of bacteria (Gorham & Carmichael, 1988; Østensvik et al, 1998; Valdor & Aboal, 2007). Finally, there may also be direct competition for nutrients or trace elements between cyanobacteria and heterotrophic bacteria (Mure et al., 1999). The use of 16S rRNA based finger-printing methods has revealed shifts in bacterial community composition during the course of a phytoplankton bloom (Reimann & Winding, 2001) and also revealed dramatic changes in the bacterial community upon viral-induced termination of a cyanobacterial bloom (Van Hannen et al., 1999a). Aboal et al. (2000, 2002) showed that the production of microcystins was associated with a reduction in animal and algal community diversity, but did not study bacterial communities. Eiler & Bertilsson (2004), however, found that bacterial communities associated with massive blooms of cyanobacteria were very different in different habitats, suggesting that the cyanobacteria blooms did not result in a strong convergence in taxon composition of bacterial communities. They also found that bacterial communities accompanying cyanobacterial blooms are as diverse as non-bloom communities. Still, some clusters of bacterial sequences that may be characteristic for cyanobacterial blooms were detected.

The aim of the present study was to characterize the taxon composition of bacterial communities in reservoirs in the semi-arid highlands of northern Ethiopia in relation to environmental conditions, and with special emphasis on the occurrence of cyanobacteria (*Microcystis*) blooms. We investigated the taxon composition of bacterial communities in 26 eutrophic reservoirs using denaturing gradient gel electrophoresis (DGGE), and related this to environmental variables. The focus of this research was to investigate to what extent there is a causal relationship between the presence of a *Microcystis* bloom and the bacterial community composition. A comparison is made between bacterial community diversity in reservoirs with and without a *Microcystis* bloom.

Materials and Methods

Study sites and sample collection

Samples were collected from 32 shallow reservoirs in Tigray, the northernmost regional state of Ethiopia (between 12° and 15° N and between 37°10' and 40°10' E). Samples were taken once in September - October 2004 (end of the rainy season; "wet" season) and once in April - May 2005 (end of the dry season, "dry season") to explore seasonal differences. In addition to the sampling of bacterial communities a whole suite of morphometric, abiotic and biotic variables were quantified (Table 4.1). For details on sampling and analysis procedures of all variables, we refer to Dejenie *et al.* (2008). Fig. 4.1 shows the geographic location of the 32 reservoirs in Tigray. Only 26 reservoirs are included in this study because six reservoirs had dried/almost dried out at the moment of the second sampling, truncating the data-set for that season. Therefore, our study focused

on the characteristics of the permanent reservoirs in the region. To sample bacterial communities, water samples were taken at a fixed location from three depths: surface, middle , and near the bottom, using a 3 L Heart valve sampler. These samples were pooled for each reservoir and sampling date and the pooled sample was transported to the shore immediately after retrieval, where sub-samples were taken and fixed. For analysis of bacterial community composition (BCC), water was filtered over a 25mm 0.2 μ m GSWP filter (Millipore) until the filter was clogged. The filter was folded twice and packed in sterile aluminium foil. Filters were transported on ice packs to the laboratory and stored at -20 °C until further processing.



Fig. 4.1 Map with geographic location of the studied reservoirs in Tigray, North Ethiopia

DNA extraction and **PCR**

Genomic DNA from the natural bacterial communities was extracted following the protocol described by Zwart et al. (1998), which includes the bead-beating method concomitant with phenol extraction and ethanol precipitation. Bead beating is a method of disrupting cells using grinding small balls (200 micron) and a shaking homogenizer to smash, rip, and tear samples. After extraction, the DNA was purified on a Wizard column (Promega, Madison, WI) according to the manufacturer's recommendations. For DGGE analysis, a small rDNA fragment was amplified with primers 357F-GC-clamp (5'-AGCAG-3') and 518R (5'-ATTACCGCGGCTGCTGG-3'). Using these specific primers, our analysis is restricted to eunbacteria. The PCR was carried out in a Genius temperature cycler. After an incubation for 5 min at 94°C, a touchdown PCR was performed using 20 cycles consisting of denaturation at 94°C for 1 min, annealing at 65°C (the temperature was decreased by 0.5°C every cycle until the touchdown temperature of 56°C was reached) for 1 min and primer extension at 72°C for 1 min. Five additional cycles were carried out at an annealing temperature of 55°C. The tubes were then incubated for 10 min at 72°C. To determine the presence and concentration of PCR products, 5 µl of the product was put on 1% (wt/vol) agarose gels, stained with ethidium bromide, and compared to a molecular weight marker (Smartladder; Eurogentec).

Table 4.1 Summary statistics of limnological variables in the 32 studied reservoirs during the wet and dry season sampling campaign (2004 and 2005, respectively). Data on nutrient concentrations of the dry season were accidently lost; data of the wet season were used for all analyses. For more detailed information on the various variables, see Dejenie *et al.* (2008).

	Wet season (2004)			Dry season (2005)		
Variable	Mean	Min	Max	Mean	Min	Max
Abiotic variables						
pH	8.46	7.47	9.30	8.66	7.45	9.58
Conductivity (μ S cm ⁻¹)	203	75	471	283	135	824
Secchi disc transparency (m)	0.46	0.05	1.50	0.44	0.12	2.00
Suspended matter $(mg l^{-1})$	42	6	737	61	6	596
Total phosphorus ($\mu g l^{-1}$)	52	11	221			
Total nitrogen (µg l ⁻¹)	850	232	2411			
Phytoplankton (pgC l ⁻¹)	$20*10^5$	4000	$20*10^{6}$	$10*10^{6}$	9000	$10*10^{7}$
% Microcystis	34.8	0	99.7	5.1	0	98.2
Zooplankton						
(μ g dry weight l ⁻¹)	295	9	1603	722	8	4137
Rotifers (µg dry weight						
l^{-1})	16	0	135	177	0	4127
Copepods			(0.0	100	•	
$(\mu g dry weight \Gamma^{+})$	144	2	683	120	2	755
Cladocerans	105		1100	200		2065
(µg dry weight l ⁻)	135	1	1128	308	1	2065
$Daphnia$ (µg dry weight l^{-1})	18	0	254	117	0	1597

DGGE analysis

Denaturing gradient gel electrophoresis (DGGE) is a molecular fingerprinting method that separates polymerase chain reaction (PCR)-generated DNA products (Muyzer et al., 1993). DGGE can separate PCR products based on sequence differences that results in differential denaturing characteristics of the DNA. During DGGE, PCR products encounter increasingly higher concentrations of chemical denaturant as they migrate through a polyacrylamide gel. Upon reaching a threshold denaturant concentration, the weaker melting domains of the double-stranded PCR product will begin to denature at which time migration slows dramatically. Differing sequences of DNA (from different bacteria) will denature at different denaturant concentrations resulting in a pattern of bands; each band theoretically representing a different bacterial population present in the community (referred and the Operational Taxonomic Unit, OTU).

Equal amounts of PCR products were analyzed on a 35 to 70% denaturant DGGE gel as described by Van der Gucht et al. (2001). DGGE gels were stained with ethidium bromide and photographed on a UV transillumination table (302 nm) with a CCD camera. As standard, a mixture of DNA from nine clones was used, which were obtained from a clone library of the 16S rRNA genes from a small eutrophic lake. On every gel, three or four standard lanes were analyzed in parallel to the samples. Since these bands are expected to form at the same denaturant concentration in the gel, their position was used to compare the patterns formed in different gels. Digitized DGGE images were analyzed using the software package Bionumerics 5.1 (Applied Maths BVBA, Kortrijk, Belgium).

The software performs a density profile through each lane, detects the bands and

calculates the relative contribution of each band to the total band signal in the lane after applying a rolling disk as background substraction. Bands occupying the same position in the different lanes of the gel were identified by visual inspection, and were subsequently analysed as operational taxonomic units (OTUs). Cyanobacteria were excluded during the analysis of the OTUs. Sequence information of a subset of the bands (see further) was used to check the grouping of bands into band classes. A matrix was compiled based upon the relative constribution of individual bands (representing OTUs) to the total band signal in each lane.

A number of bands with more than 40% relative band intensity in at least two samples were selected for sequencing. These bands were excised and sequenced after re-extraction and amplification. Sequencing was performed with an ABI-Prism sequencing kit (PE Biosystems) using primer Stef1Tex (5_- GCGTTCATCGTTGCGAG-3_) and an automated sequencer (ABI-Prism 377). A nucleotide BLAST search (http://www.ncbi.nlm.nih.gov/BLAST/) was performed in order to find the closest match and identity the taxa to which a sequence belongs.

Detection of toxin genes

Potentially toxic species of cyanobacteria were detected using PCR. We searched for two genes coding for different variants of the hepatotoxine microcystin: (1) *mcy A*, present in different toxin producing cyanobacteria such as *Microcystis*, *Anabaena* and *Planktothrix* (Hisbergues *et al.* 2003) and (2) *mcy E*, characteristic for *Microcystis* (Vaitooma *et al.* 2003).

Data analysis

DGGE banding data were used to determine bacterial community composition, using band intensity as a measure for the relative abundances of OTUs (see also Van der Gucht et al., 2007, Muylaert et al., 2003). We estimated bacterial richness by treating each band as an individual OTU and using the number of bands as an indicator of richness.

Multivariate statistical analysis on relative abundances of OTUs as determined by relative band intensity was used to explore relationships between BCC and environmental conditions as well as spatial variables. To evaluate the association between BCC and environmental variables, a principal component analysis PCA was performed followed by a redundancy analysis (RDA). Prior to the analyses, OTU's occurring in only one reservoir were omitted. All phytoplankton and environmental variables, except pH, were log(x+1) transformed. In RDA, a forward selection procedure was used to add significant explanatory variables to the model. They were added in the order of largest additional contribution to the total variance explained. To avoid inflation of the Type I error due to a large number environmental variables, an *a priori* selection of environmental variables was made, and the significance of a global model including all variables checked before applying the forward selection procedure (Blanchet et al., 2008). We additionally checked whether the R^2 adjusted of the forward selection model did not exceed that of the full model. The statistical significance in RDA was assessed by Monte-Carlo permutation tests (999 permutations). We also assessed the unique contribution of variables through partial RDA analysis. The computer program CANOCO version 4.5 for Windows (center for Biometry - Plant Research International, Wageningen; the Netherlands) was used to perform the PCA and RDA. Correlations were carried out with Statistica version 6.0 for
Windows (StatSoft Inc., Tulsa, USA). Data on environmental variables were taken from Dejenie et al. (2008).

To investigate the relationship between BCC and the occurrence of *Microcystis* blooms, the occurrence of a *Microcystis* bloom was identified from phytoplankton counts (Tsehaye et al., unpublished manuscript, chapter 2 of this thesis). Examination of the phytoplankton community in the reservoirs revealed three genera of cyanobacteria that are potentially able to produce toxins (see Paerl *et al.*, 2001): *Anabaena, Aphanizomenon* and *Microcystis*. *Microcystis* was the most abundant cyanobacterium in both seasons. Reservoirs were considered to harbour a *Microcystis* bloom when *Microcystis* represented 65% or more of the total phytoplankton biomass. Bacterial OTU richness of the reservoirs in which a *Microcystis* bloom occurred and those without a *Microcystis* bloom were compared by unpaired Student's t test, using the software package Statistica 6.0 (Statsoft, Inc. 1995).

Results

Microcystis blooms in relation to environmental variables

Microcystis was observed to form blooms in four reservoirs in the dry season and seven reservoirs in the wet season (Table 4.2). *Microcystis* abundance in the 26 reservoirs was correlated over the two seasons (r = 0.156, p = 0.02). Microcystin synthetase genes (*mcyE and mcyA*) were detected in all reservoirs where *Microcystis* was detected. In the wet season, *Microcystis* was positively correlated with pH (r = 0.4429, p = 0.002) and fish (*Garra* species) biomass (r = 0.467, p = 0.016). In the dry season, *Microcystis* biomass was positively correlated with conductivity (r = 0.5246, p = 0.06) and rotifers (r = 0.427, p = 0.03), and negatively with *Daphnia* biomass (r = -0.4384, p = 0.025) and altitude of the

reservoirs (r =-0.394, p=0.046). The biomass of *Daphnia* was significantly higher in the reservoirs without than with a *Microcystis* bloom (*t*-test p = 0.039).

Table 4.2 Occurrence of *Microcystis* blooms in highland reservoirs in Tigray during the wet and dry season. Only reservoirs are mentioned in which a *Microcystis* bloom was observed during at least one season. A bloom is defined by *Microcystis* accounting for 65% or more of total phytoplankton biomass.

Reservoir	% Microcystis biomass				
	Wet season	Dry season			
Adi Amharay	0	65.5			
Betequa	76.4	5.7			
Dur Anbassa	99.7	49.8			
Era Quihila	80.8	0			
Mai Dele	0	96.2			
Mai Gassa I	56.7	67.5			
Mai Gassa II	99.6	98.2			
Mai Nigus	73.1	0.1			
Meala	69.3	4.5			
Shilanat IV	97.6	0			

Bacterial communities in relation to environmental variables

In total, 38 different bacterial OTUs were detected, of which 38 were observed in the wet season and 35 in the dry season. Bacterial richness for the individual reservoirs varied between 2 to 15 (average 7.2) and 3 to 13 (average 8.4) OTUs in the wet and dry season, respectively. OTU richness in the reservoirs was significantly different between the two seasons (Mann-Whitney U Test, p < 0.05, N = 26). We observed a positive correlation between bacterial richness and submerged vegetation (r = 0.4789, p = 0.013).

Mantel test (Zt, Bonnet and Van de Peer, 2002) revealed no correlation between the bacterial community composition (BCC) of the two seasons (r = 0.009; p = 0.450). Redundancy analysis (RDA) with forward selection and unrestricted Monte Carlo permutation tests revealed that the percentage of *Microcystis* in the phytoplankton community (% *Microcystis*) and copepod biomass (CopB) significantly explain 18% of the variation in BCC of the wet season samples (Fig. 4.2a). The effect of *Microcystis* on the bacterial community diversity in the wet season remained significant after we controlled for the contribution of copepods in a variation partitioning analysis (Fig. 4.3). For the samples taken in the dry season, redundancy analysis (RDA) with forward selection and unrestricted Monte Carlo permutation tests revealed that TN and TP (both positively correlated with altitude), oxygen concentration (O₂; positively correlated to pH), the amount of cattle (as derived from the number of cattle dung along the shoreline, see Dejenie et al., 208) and fish biomass (fish B) significantly explain a total of 32% of the variation in BCC (Fig. 4.2b).



Fig. 4.2 Principal component analysis (PCA) of bacterial community composition (BCC) in the studied reservoirs in the wet (a) and dry (b) seasons, showing the structuring role of copepod biomass (CopB) and % Microcystis (accounting for 18% of the variation in the dataset of the wet season) and TN, TP, oxygen concentration (O2), cattle and fish biomass (fishB) (accounting for 32% of the variation in the dataset of the dry season)



Fig. 4.3 Variance partitioning of the BCC in the wet season. Both % *Microcystis* and Copepod biomass explained variation undependently, there were no common effects.

Bacterioplankton richness in relation to the occurrence of Microcystis blooms

OTU richness in reservoirs with *Microcystis* bloom ranged from 2 to 9 (average 6.2), whereas it ranged from 3 to 14 (average 8.3) in reservoirs without a *Microcystis* bloom. This difference was statistically significant (*t*-test, p = 0.042). No overall correlation was, however, observed between OTU richness and *Microcystis* biomass or percentage contribution of *Microcystis* to total phytoplankton biomass.

Discussion

Bacterial taxon richness as detected by DGGE analysis in individual reservoirs ranged from 2 to 15, which is somewhat lower than the bacterial richness observed in most other lakes (Konopka et al., 1999; Van der Gucht et al. 2001, 2007). Van der Gucht et al. (2007) reported 11 to 37 bacterial OTUs from temperate lakes. Our results suggest that bacterial communities accompanying *Microcystis* blooms are characterised by lower

taxon richness than bacterial communities in reservoirs without *Microcystis* blooms. This is in contrast with the results of Eiler & Bertilsson (2004), who found no indications that the environmental conditions created by cyanobacterial blooms lowered the community richness or decreased the evenness of OTUs in a clone library study. We feel our observation should be considered with caution, because there is a possibility that the lower taxon richness observed in the reservoirs with a *Microcystis* bloom is the result of an artefact in the PCR reactions when there is a strong dominance of one taxon (Acinas et al., 2005).

Our data provide evidence that the relative abundance of *Microcystis* is significantly associated with taxon composition of the bacterial community, at least in the wet season. Cyanobacteria, including *Microcystis* strains, are known to produce secondary metabolites that have antimicrobial and toxic effects (Carmichael, 1992; Patterson *et al.* 1994). Valdor & Aboal (2007) demonstrated an inhibitory effect of both cyanobacterial extracts and pure microcystin on the growth of microalgae and bacteria. These inhibitory effects were even longer lasting on bacteria than on microalgae. Østensvik et al. (1998) reported antibacterial activity of secondary metabolites from cyanobacteria, including *Microcystis aeruginosa*. Similarly, Casamatta & Wickstrom (2000) showed that *Microcystis* has toxic properties for some bacterioplankton strains.

In addition to the percentage of *Microcystis*, bacterial community composition in the reservoirs during the wet season was also significantly impacted by copepod biomass. Grazing by zooplankton, both protozoans as well as crustacean zooplankton, has been

136

identified as a force driving changes in bacterial community composition (Langenheder & Jürgens, 2001; van Hannen *et al.* 1999b). Jürgens et al. (1993) observed an association between copepod biomass and bacterial abundances in an enclosure study (Jürgens et al., 1993). Several studies have reported that the highest bacterial growth and abundances occur in the presence of copepods (Jürgens et al., 1993; Roman et al., 1998). This is suggested to result from a combined effect of cascading predation effects on protozoans that feed on bacterioplankton and copepod excretion stimulating the growth of bacteria.

In the dry season, the relationships of bacterial community composition with environmental factors are different from those observed in the wet season. We observed significant associations of bacterial community composition with total nitrogen (TN), total phosphorus (TP), oxygen, the number of cattle frequenting the reservoir, and fish biomass. Muylaert et al. (2002) reported significant associations of bacterial community composition of shallow lakes with both bottom-up factors (temperature, phytoplankton biomass, nitrogen concentration, phosphorus concentration, and pH) as well as top-down factors (biomass of oligotrich and non-oligotrich ciliates, Daphnia and Ceriodaphnia), but with a stronger impact of bottom-up factors in turbid lakes. Other studies too have reported a joint influence of bottom-up and top-down regulation of bacterioplankton (Pace & Cole, 1996), whereas Langenheder & Jürgens (2001) reported regulation of bacterial community composition by Daphnia grazing. Our data suggest that BCC in the studied reservoirs in the dry season were largely regulated by bottom-up factors, which is broadly in agreement with Muylaert et al. (2002), as the reservoirs in the dry season are characterized by higher conductivities, phytoplankton biomass and lower amounts of suspended matter than in the wet season. The amount of cattle frequenting the reservoirs

impacts bottom-up factors by influencing nutrient concentrations both directly by their excretion products as well as indirectly by trampling and thus enhancing sediment resuspension. Cattle may also change the quality of organic matter available for bacteria to grow upon. Although the impact of fish may reflect a top-down control, the biomass of fish may also strongly impact the ecosystem of the reservoirs through a bottom-up effect, both by excreting nutrients as well as by enhancing sediment resuspension.

We conclude that bacterioplankton community structure of Ethiopian highland reservoirs is significantly related to both bottom-up as well as top-down factors, but that the relative importance of these factors changed strongly among the wet and dry season in the study period. The relative abundance of the cyanobacterium *Microcystis* seems to have a significant influence on bacterial community composition during the wet season.

Acknowledgements

This study is part of the "Aquatic Ecology" project of the VLIR IUC scientific cooperation program between Flemish universities and Mekelle University. We thank the Flemish government for financial support to this program, and the many people who joined us in the field or contributed to solve logistic issues for their help.

References

- Aboal, M., M.A. Puig, H. Rios and E. Lopez-Jimenez, 2000. Relationship between macroinvertebrate diversity and toxicity of cyanophyceae (Cyanobacteria) in some streams from Eastern Spain. *Int. Assoc. Theoretical Applied Limnol. Proc.*, 27: 550-559
- Aboal, M., M.A. Puig, P. Mateo and E. Perona, 2002. Implications of cyanophyte toxicity on biological monitoring of calcareous streams in North-East Spain. J. Applied Phycol., 14: 49-56
- Acinas, S.G.; R. Sarma-Rupavtarm; V. Klepac-Ceraj and M.F. Polz, 2005. PCR induced sequence artefacts and biases: insights from comparison of two 16S rRNA clone libraries constructed from the same sample. *Applied and Environmental Microbiology*, **71**: 8966-8969
- Akin-Oriola, G. A., 2003. On the phytoplankton of Awab reservoir, Ibadan, Nigeria. *Rev. Biol. Trop.*, **51**: 99-106
- Allgaier, M., and H.P. Grossart, 2006. Seasonal dynamics and phylogenetic diversity of free-living and particle-associated bacterial communities in four lakes in northeastern Germany. *Aquatic Microbial Ecology*, **45**: 115-128
- Blanchet F.G., Legendre P., and D. Borcard, 2008. Forward selection fo explanatory variables. *Ecology*, **89:** 2623-2632.
- Bonnet, E. and Y. Van de Peer, 2002. Zt: a software tool for simple and partial Mantel tests. *Journal of Statistical software*, **7**: 1-12
- Brett, M.T., F.S. Lubnow, M. Villar-Argaiz, A. Muller-Solger and C.R. Goldman, 1999.

Nutrient control of bacterioplankton and phytoplankton dynamics. *Aquatic Ecology*, **33**: 135-145

- Carmichael, W.W., 1992. Cyanobacteria secondary metabolites the cyanotoxins. Journal of Applied Bacteriology, 72: 445 – 459
- Caron, D.A., 1994. Inorganic nutrients, bacteria, and the microbial loop. *Microbial Ecology*, **28**: 295 – 298
- Casamatta, D.A. and C.E. Wickstrom, 2000. Sensitivity of two disjunct bacterioplankton communities to exudates from cyanobacterim *Microcystis aeruginosa* Küutzing. *Microbial Ecology*, **41**: 64-73
- Cole, J.J. and D.L. Strayer, 1982. Photosynthetically produced organic carbon: an important carbon source for planktonic bacteria. *Limnology & Oceanography*, 27: 1080 – 1090
- Currie, D.J., 1990. Large-scale variability and interactions among phytoplankton, bacterioplankton and phosphorus. *Limnology & Oceanography*, **35**: 1437–1455
- Dejenie, T., Tsehaye, A., L. De Meester, Afework M., Abreha, G., A., S. Risch, A. Pals,
 K. van der Gucht, W. Vyverman, J. Nysen, J. Deckers, and S. Decklerck 2008.
 Limnological and ecological characteristics of tropical semi-arid highland
 reservoirs in Tigray, Northern Ethiopia. *Hydrobiologia*, 610: 193-209
- De Wever, A., K. Muylaert, K. Van der Gucht, S. Pirlot, C. Cocquyt, J.-P. Descy, P.-D. Plisnier, 2005. Bacterial community composition in Lake Tangnyika: vertical and horizontal heterogeneity. *Applied & Environmental Microbiology*, **71**: 5029 – 5037
- Dimitriu, P.A., H.C. Pinkart, B.M. Peyton, and M.R. Mormile, 2008. Spatial and temporal patterns in the microbial diversity of a meromictic Soda Lake in

Washington State. Applied and Environmental Microbiology, 74: 4877–4888

- Elier, A., and S. Bertilsson. 2004, Composition of freshwater bacterial communities associated with cyanobacterial blooms in four Swedish lakes. *Environmental Microbiology*, **6**: 1228 1243
- Forsberg, C. and S.O. Ryding, 1980. Eutrophication parameters and trophic state indices in 30 Swedish waste-receiving lakes. *Archive fur Hydrobiologie*, **89:** 189 – 2007
- Gorham, P.R., and W.W. Carmichael, 1988. Hazards of freshwater blue-green algae (cyanobacteria). In *Algae and Human Affairs*. Lembe, C.A., and J.R. Waaland, (eds). Cambridge, UK: University Press, pp. 403–431
- Gross, E.M. 2003, Allelopathy of aquatic autotrophs. *Critical review in plant science*, **22**: 313 339
- Hisbergues, M., G. Christiansen, L. Rouhiainen, K. Sivonen, and T. Borner. 2003, PCRbased identification of microcystin-producing genotypes of different cyanobacterial genera. *Archives of Microbiology*, **180**: 402-410
- Jürgens, K., H. Arndt, and K.O. Rothhaupt, 1993. Zooplankton-mediated changes of bacterial community structure. *Microbial Ecology*, 27: 27 – 42
- Konopka, A., T. Bercot, and C. Nakatsu, 1999. Bacterioplankton community diversity in a series of thermally stratified lakes. *Microb Ecol*, **38**: 126–135
- Kristiansen, K., H. Nielsen, B. Riemann, and J.A. Fuhrman, 1992. Growth efficiencies of freshwater bacterioplankton. *Microbial Ecology*, 24: 145 – 160
- Langenheder, S., and K. Jürgens, 2001. Regulation of bacterial biomass and community structure by metazoan and protozoan predation. *Limnol. Oceanogr.*, **46**:121–134
- Menden-Deuer, S., and E.J. Lessard, 2000. Carbon to volume relationships for

dinoflagellates, diatoms, and other protist plankton. *Limnol. Oceanogr.*, **45:**569–579

- Moss, B., 1998. Ecology of fresh waters, man and medium, past to future. Blackwell Scientific Publicaionns, London, pp. 557.
- Mur, L.R., O.M. Skulberg, and H. Utkilen, 1999. Cyanobacteria in the environment. In *Toxic Cyanobacteria in Water: a Guide to Their Public Health Consequences, Monitoring and Management*. Chorus, I., and J. Bartram, (eds). London: E & FN Spon, pp. 15–40.
- Muylaert, K., K. Van der Gucht, N. Vloemans, L. De Meester, M. Gillis, and W. Vyverman, 2002. Relationship between bacterial community composition and bottom-up versus top-down variables in four eutrophic shallow lakes. *Applied and Environmental Microbiology*, 68: 4740 4750
- Muyzer, G., E.C. Dewaal, and A.G. Uitterlinden, 1993. Profiling of complex microbial populations by denaturing gradient gel-electrophoresis analysis of polymerase chain reaction-amplified genes-coding for 16s ribosomal RNA. *Applied and Environmental Microbiology*, **59**: 695 – 700
- Pace, M.L. and J.J. Cole, 1994. Comparative and experimental approaches to top-down and bottom-up regulation of bacteria. *Microb. Ecol.*, **28**: 181 193
- Paerl, H.W., R. Fulton, P.H. Moisander, and J. Dyble, 2001. Harmful freshwater algal blooms, with an emphasis on cyanobacteria. *The Scientific World*, **1**: 76 113
- Patterson, G.M.L., L.K. Larsen, and R.E. Moore, 1994. Bioactive natural products from blue-green algae. *Journal of Applied Phycology*, **6**: 151 157
- Pirlot, S.,F. Unrein, J.-P. Descy & P. Servais, 2007. Fate of heterotrophic bacteria in Lake

Tanganyika (East Africa). FEMS Microbial Ecology, 62: 354364

- Østensvik Ø., O.M. Skulberg, , B. Underdal, , and V. Hormazabal, 1998. Antibacterial properties of extracts from selected planktonic freshwater Cyanobacteria a comparative study of bacterial bioassays. *J. Appl. Microbiol.*, **84:**1117–1124
- Rejas, D., K. Muylaert, and L. De Meester, 2005. Phytoplankton-bacterioplankton interactions in a neotropical floodplain lake (Laguna Bufeos, Bolivia). *Hydrobiologia*, 543: 91–99
- Riemann, L., and A. Winding, 2001. Community dynamics of free-living and particleassociated bacterial assemblages during a freshwater phytoplankton bloom. *Microb Ecol*, **42**: 274–285
- Roman, H.R., H.W. Ducklow, J.A. Fuhrman, C. Garside, P.M. Gilbert, T.C. Malone, T.C. McManus, 1988. Production, consumption and nutrient cycling in a laboratory mesocosm. *Mar. Ecol. Prog. Ser.*, **42**: 39 52
- ShadeA., Kent A.D., Jones S.E., Newton R.J., Triplett E.W. and McMahon K.D. 2007. Interannual dynamics and phenology of bacterial communities in a eutrophic lake. Limnology and Oceanography, 52: 487–494
- Scheffer, M., 1988. Ecology of shallow lakes. Chapman ang Hall, London, pp. 357.
- Schweitzer, B. & Simon, M. 1995. Growth limitation of planktonic bacteria in a large mesotrophic lake. *Microbial Ecology* **30**: 89–104
- Thingstad, F.T., U.L., Zweifel, and F. Rassoulzadegan, 1998. P limitation of heterotrophic bacteria and phytoplankton in the northwest Mediterranean. *Limnology & Oceanography*, 43: 88–94
- Toolan, T., J.D. Wehr and S. Findlay, 1991. Inorganic phosphorus stimulation of

bacterioplankton production in a meso-eutrophic lake. *Applied Environmental Microbiology*, **57**: 2074–2078.

- Van der Gucht, K., K. Sabbe, L. De Meester, N. Vloemans, G. Zwart, M. Gillis, and W. Vyverman, 2001. Contrasting bacterioplankton community composition and seasonal dynamics in two neighbouring hypertrophic freshwater lakes. *Environ. Microbiol.*, 3: 680–690
- Van der Gucht, K., K. Cottenie, K. Muylaert, N. Vloemans, S. Cousin, S. Declerck, E. Jeppesen, J.-M. Conde-Porcuna, K. Schwenk, G. Zwart, H. Degans, W. Vyverman.,
 L. De Meester, 2007. The power of species sorting: Local factors drive bacterial community composition over a wide range of spatial scales. *PNAS*, 104, 20404-20409
- Van Hannen, E.J., G. Zwart, M.P. Van Agterveld, H.J. Gons, J. Ebert, and H.J. Laanbroek, 1999a. Changes in bacterial and eukaryotic community structure after mass lysis of filamentous cyanobacteria associated with viruses. *Appl Environ Microbiol*, 65: 795–801
- Van Hannen, E.J., M. Veninga, J. Bloem, H.J. Gons, and H.J. Laanbroek, 1999b. Genetic changes in the bacterial community structure associated with protistan grazers. *Arch. Hydrobiol.*, 145:25–38
- Vaitooma, J., A. Rantala, K. Halinen, L. Rouhiainan, P. Tallberg, L. Mokelke, and K. Sivonen, 2003. Quantitative real-time PCR for determination of Microcystin synthetase E copy numbers for *Microcystis* and *Anabaena* in lakes. *Applied and environmental Microbiology*, 69: 7289-7297
- Vrede, K., T. Vrede, A. Isaksson and Karlsson, 1999. Effects of nutrients (phosphorus,

nitrogen, and carbon,) and zooplankton on bacterioplankton and phytoplankton – a seasonal study. *Limnology & Oceonography*, **44**: 1616 – 1624

- Wu S.K., P. Xie, G.D. Liang, S.B. Wang, and X.M. Liang, 2006. Relationships between microcystins and environmental parameters in 30 subtropical shallow lakes along the Yangtze River, China. *Freshwater Biology*, **51**: 2309 – 2319
- Zinabu, G.-M. and W.D. Taylor, 1989. Seasonality and spatial variation in abundance, and activity of heterotrophic bacterioplankton in relation to some biotic and abiotic variables in an Ethiopian riftvalley lake (Awassa). *Freshwater Biology*, **22:** 155-368
- Zinabu, G.-M. and W.D. Taylor, 1990. Heterotrophic bacterioplankton production and grazing mortality rates in an Ethiopian rift-valley lake (Awasa). *Freshwater Biology*, **22**: 369 381
- Zwart, G., R. Huismansvan, M.P. van Agterveld, Y. Van de Peer, P. De Rijk, H. Eenhoorn, G. Muyzer, E.J. van Hannen, H.J. Gons, and H.J. Laanbroek, 1998.
 Divergent members of the bacterial division *Verrucomicrobiales* in a temperate freshwater lake. *FEMS Microbial. Ecol.*, 25: 159-169

Chapter 5

Impact of the fish Garra on the ecology of reservoirs and the occurrence of *Microcystis* blooms in semi-arid tropical highlands: an experimental assessment using enclosures*

Tadesse Dejenie¹, Tsehaye Asmelash², Sarah Rousseaux³, Teklit Gebregiorgis⁴, Abreha Gebrekidan⁴, Mekonnen Teferi¹, Jan Nyssen⁵, Josef Decers⁶, Katleen Van Der Gucht⁷, Wim Vyverman⁷, Luc De Meester³, and Steven A.J., Declerck³.

Manuscript published online in Freshwater Biology, 2009 (http://www3.interscience.wiley.com/cgi-bin/fulltext/122260892/PDFSTART)

¹Department of Biology, Mekelle University, Mekelle, Ethiopia

- ²Department of Microbiology, Mekelle University, Mekelle, Ethiopia
- ⁷Laboratory of Aquatic Ecology and Protistology, Ghent University, Ghent, Belgium
- ⁴Department of Chemistry, Mekelle University, Mekelle, Ethiopia
- ⁵Geography Department, Ghent University, Ghent, Belgium

^{*} Refer appendix 5.1 for the analysis of the phytoplankton community composition

³Laboratory of Aquatic Ecology and Evolutionary Biology, Katholieke Universiteit Leuven, Leuven, Belgium

⁶Division Forest, Nature and Landscape, Katholieke Universiteit Leuven, Geo-instituut, Leuven, Belgium

Summary

Many man-made reservoirs in the semi-arid highlands of Northern Ethiopia (Tigray) are characterised by the occurrence of intensive blooms of cyanobacteria and a dominance of small riverine fishes belonging to the genus Garra. We carried out enclosure experiments to test for the effect of these small fish on abiotic characteristics, phytoplankton biomass and zooplankton community structure in the pelagic of two reservoirs (Gereb Awso and Tsinkanet). Two experiments were carried out in each of the reservoirs, one at the end of the rainy season (highest water level) and one at the end of the dry season (lowest water level). The presence of Garra in general increased the amount of suspended matter, nutrient concentrations (total nitrogen and total phosphorus), phytoplankton and *Microcystis* biomass (including the proportion of Microcystis in the phytoplankton community), and reduced water transparency. The positive effect of the presence of *Garra* on nutrient concentrations and phytoplankton productivity indicate that *Garra* has the potential to affect food web functioning indirectly through bottom-up effects, by enhancing nutrient concentrations through sediment resuspension and excretion of nutrients. Indeed, population densities of the cladoceran zooplankton taxa Ceriodaphnia and Diaphanosoma also showed an overall increase in enclosures with Garra. However, our data also provide some evidence for a potential of Garra to exert top-down control on large bodied daphnids (Daphnia carinata, D. barbata), although such effect varied among experiments. The limited capability of Garra to control zooplankton communities mainly reflects the low efficiency of these small, riverine and benthosoriented fish in foraging on zooplankton and suggests the existence of an unoccupied niche for zooplanktivorous fish in the majority of the reservoirs. Although the main effects of Garra on the pelagic food web seemed to be mediated by bottom-up

mechanisms, our results also indicate that one of the key variables, the relative abundance of *Microcystis*, was impacted by *Daphnia*-mediated trophic cascade effects.

Introduction

Man-made reservoirs represent a considerable fraction of standing waters across the world (Cooke et al., 1993; Kalff, 2002). In semi-arid regions, reservoirs play an important role in providing access to water for irrigation and watering cattle. Often, water quality in such reservoirs is poor and is characterised by high nutrient levels, high turbidity and phytoplankton blooms. In the semi-arid highlands of Tigray, Northern Ethiopia, more than 70 small (c. 18 000–454 000 m²) reservoirs have been constructed for the purpose of irrigation and watering livestock. These reservoirs were often constructed in valleys with rivulets, and many (c. 65%) became colonised by small riverine fishes of the genus *Garra*. Piscivorous fish are regionally absent from the network of rivulets in the Ethiopian highlands. Because of the high productivity of some of the reservoirs and the absence of predatory fish, *Garra* populations often reach high densities (Dejenie et al., 2008).

A survey of 32 reservoirs of Tigray revealed the presence of cyanobacteria in high relative (>20% of algal biomass) and absolute biomass in more than 65% of the studied reservoirs (Dejenie et al., 2008). The data from that survey also suggested a strong bottom-up effect, with the biomass of phytoplankton and fish (mainly *Garra*) being positively correlated with an altitude-related gradient of total phosphorus and with a positive correlation between the biomass of fish and cladocerans. Yet, the biomass of cyanobacteria was negatively related to the biomass of cladocerans (*Daphnia*), which may reflect the potential of daphnids to exert a top-down control on cyanobacteria. Overall, the survey suggested only limited top-down control by *Garra* on zooplankton. Representatives of the genus *Garra* are present in Asia and Africa (Getahun & Stiassny, 1998; Golubtsov, Degbuadze & Mina, 2002). About 60% of the African species are found in Ethiopia (Getahun & Stiassny, 1998) and many of them are endemic to the region (Getahun, 2000). *Garra* are often found in large numbers in streams. The genus is not found higher than 3000 m a.s.l. (Getahun & Stiassny, 1998). Although very common in Tigray and some other regions of Eastern Africa, it is unclear to what extent this fish taxon can impact the pelagic food web of standing waters through planktivory. This study had two major objectives: (i) to assess experimentally the potential for top-down control of zooplankton by *Garra*, and (ii) to test for a causal relationship between the presence of *Garra* and the occurrence of cyanobacteria blooms. For this, we carried out in situ enclosure experiments in two ecologically contrasting reservoirs that are both characterised by the presence of *Garra*, albeit at different densities.

Methods

Study area and design of enclosure experiment

We performed two enclosure experiments in each of the reservoirs Gereb Awso (surface: 2.4 ha; depth: 3.5 m; 39°33'17"N; 13°25'59'E) and Tsinkanet (surface: 7 ha; depth: 4.3 m; 39°32'40"N; 14°54"E), two reservoirs located in the semi-arid highlands of Tigray, Northern Ethiopia (Dejenie et al., 2008). The two study reservoirs are ecologically quite distinct. Although they are both characterised by high nutrient concentrations (>0.5 mg TP L⁻¹ and 50 mg TN L⁻¹), the turbidity is much higher in Gereb Awso (Secchi depth <20 cm; amount of suspended matter c. 40–130 mg L⁻¹) than in Tsinkanet (Secchi depth >50 cm in September; amount of suspended matter <20 mg L⁻¹). Population densities of *Garra* are much higher in Gereb Awso [catch per unit of effort (CPUE) of 922 individuals, with effort equalling two multimesh gill nets of 28 m total length during one night] than in Tsinkanet (CPUE of 65 individuals). Whereas Gereb Awso (GA) is located centrally in a small rural village,

Tsinkanet is located in farmland area. Probably related to this quieter location, Tsinkanet (TS) hosts a small group of *Pelecanus onocrotalus* L. (5–15 birds) and other fish-eating birds such as *Podiceps cristatus* L. and *Sterna albifrons* Pallas.

In each reservoir, we performed one enclosure experiment during September (GA1 and TS1, starting dates at 14 and 15 September 2005) and one enclosure experiment during April (GA2 and TS2, start at 26 and 27 April 2006). The month of September follows shortly after the rainy season, and in that period the reservoirs are filled to full capacity. The month of April is typically at the end of a prolonged dry period, and water levels tend to reach their annual minima.

For each experiment, we filled six floating enclosures (1000 L polypropylene tanks: diameter: 1 m, depth: 1.3 m) in the pelagic zone of the reservoirs with 64 μ m mesh filtered water from the reservoir. Subsequently, we inoculated the enclosures with pelagic zooplankton of the reservoir. We collected this zooplankton by making vertical hauls with a conical zooplankton net (64 μ m mesh), integrating the entire water column. The volume of lake water filtered by this method was equivalent to the volume of the enclosures. We applied this procedure to best represent the entire pelagic zooplankton community in the enclosures by accounting for the heterogeneity in the vertical distribution of the zooplankton.

We applied the same manipulation in all experiments. We added four equally sized *Garra* individuals (*Garra ignestii* Gianferrari; average length: 117 mm; average weight: 28 g) to three randomly selected enclosures (F), whereas the remaining enclosures were kept as control (no fish; NF). To prevent predation by birds, all enclosures were covered with nylon netting (mesh: 1 mm²). We choose to use *Garra ignestii* for the fish treatment because *Garra* is the only dominant fish species found

in the reservoirs (Tsinkanet also hosts a small population of stocked *Oreochromis niloticus* L.).

Sampling and sample analysis

The enclosures were sampled at 4-day intervals during a period of 4 weeks. We measured dissolved oxygen, pH, temperature and conductivity in situ with a WTW Multi 340 I electrode. We collected depth integrated water samples from the central part of the enclosures with a tube sampler (diameter: 670 mm). These samples were taken for the quantification of phytoplankton, nutrients and suspended matter. The concentration of chlorophyll a was measured in the field with a fluorometer (Turner Aquafluor; average of three measurements). Phytoplankton samples were preserved with acid lugol solution. Suspended matter was determined by filtering 200 mL on a pre-weighed Whatman GE/C filter in the field and quantifying dry weight in the laboratory. The remainder of the samples were taken to the laboratory in a refrigerator box and frozen at -18 °C until further analysis. The concentrations of total phosphorus and total nitrogen were measured following the ascorbic acid method and the Kjeldahl method respectively (Anderson & Ingram, 1989). Unfortunately, the samples from the first enclosure experiments (GA1 and TS1) were lost in transport, so only the samples from GA2 and TS2 could be analysed for nutrient concentrations.

We collected zooplankton samples with a Schindler-Patalas plankton trap (12 L). We always took samples in the central zone of the enclosures, two at the surface and two at a depth of 1 m. Samples were fixed with sucrose-saturated formalin solution (4% final concentration) to prevent egg loss from cladoceran carapaces (Haney & Hall, 1973).

153

Phytoplankton was counted and identified to genus level and cyanobacteria to species level, following published manuals and identification guides (Whitford & Schumacher, 1973; Komarek & Anagnostidis, 2000, 2005; John, 2002). Biomass in terms of carbon of cyanobacteria was obtained through biovolume to biomass conversions following counting (Hillebrand et al., 1999; Menden-Deuer & Lessard, 2000). Zooplankton samples were counted using a stereomicroscope. Subsamples of known volumes were taken; for each sample at least 150 individuals of cladocerans and copepods were counted and identified. Cladocerans were determined to species level, except for *Diaphanosoma* and *Ceriodaphnia*. Copepods were determined to suborder level. Determinations of cladocerans were based on Benzie (2005) and Flossner (2000). In addition to the crustacean zooplankton, we also counted and measured larvae of the phantom midge *Chaoborus* (50 individuals per sample, if available).

Statistical analyses

We started the analyses with a multivariate approach to make an overall evaluation of the effect of *Garra* on pelagic communities and to avoid problems related to multiple testing. In these analyses, we distinguished between two groups of response variables: (i) abiotic variables (i.e. suspended matter, water transparency, pH, and oxygen saturation), chlorophyll a and the proportion of *Microcystis* biomass to the total biomass of the phytoplankton community, and (ii) the zooplankton density matrix. Prior to analysis, the response variables were averaged over time, log-transformed and abiotic variables, chlorophyll a and percentage *Microcystis* were also standardised to account for differences in units. We performed principal component analyses (PCA) to evaluate qualitative differences among experiments according to lake and season and to visually assess the magnitude and direction of the shifts in response variables to the fish treatment. We performed redundancy analysis (RDA) to formally test the overall effect of fish on each of both sets of response variables for the four experiments simultaneously. Fish presence was here coded as a nominal predictor variable and the four experiments were specified as co-variables using dummy variables to account for variation among lakes and seasons. Significance testing for the effect of fish was done by 499 Monte Carlo permutations. Permutations were random but restricted to individual experiments (permutation restriction to blocks; Leps' & S'milauer, 2003). The RDA's were performed in CANOCO v4.5 (Ter Braak & S'milauer, 2002). Significant fish effects were further explored in more detail by applying additional tests to each of the response variables separately. Finally, because species of the genus *Daphnia* are potential key players in trophic cascade interactions, we wanted to obtain a more detailed insight in their dynamics. We therefore executed repeated measures ANOVA on the logarithmically transformed density data of the most abundant *Daphnia* taxa (*Daphnia barbata* Weltner and *Daphnia carinata* King) for each of the enclosure experiments, separately. These analyses were performed with the software package STATISTICA 7.0.

Results

Effect of Garra on physicochemical characteristics and phytoplankton

Conditions in non-fish enclosures differed among seasons more than among lakes, with less chlorophyll a but a higher relative amount of *Microcystis* at the end than at the beginning of the dry season (Fig. 5.1a). In each experiment, enclosures with *Garra* exhibited a substantial shift in abiotic conditions compared to the non-fish enclosures, and these shifts occurred in a similar direction along PCA1 (Fig. 5.1a).



Fig. 5.1 PCA and RDA ordination plots, showing the response of limnological variables and zooplankton communities to the presence of fish (*Garra*) in enclosures: (a) PCA-plot of abiotic variables, chlorophyll a, and the relative biomass of Microcystis (percentage of total phytoplankton biomass), (b) RDA-plot of abiotic variables, chlorophyll a, and the relative biomass of *Microcystis* (percentage of total phytoplankton biomass), (c) PCA-plot of population densities of crustacean zooplankton taxa, (d) RDA-plot of population densities of crustacean zooplankton taxa. The analyses were performed on the entire dataset of the four enclosure experiments, but individual experiments were specified as co-variables in the RDAanalysis (see Methods for details). Squares, fishless enclosures; circles, enclosures with fish; filled triangles, centroids that show the average position of replicate enclosures for all the possible combinations of treatments, seasons and lakes (a and c) or for treatments (b and d); NF, no fish; F, with fish; GA1, Gereb Awso rainy season experiment; GA2, Gereb Awso dry season experiment; TS1, Tsinkanet rainy season experiment; TS2, Tsinkanet dry season experiment. Plot axes are scaled to sample distances.

Redundancy analysis on the entire dataset (four enclosure experiments, taking the individual experiments as co-variables) indeed indicated an overall significant effect of fish on this variable set [total fraction of explained variation (λ) = 0.28, F = 19.993, p = 0.002]. Analyses performed on individual variables indicated that Garra significantly decreased water transparency ($\lambda = 0.59$) and increased the amount of suspended matter ($\lambda = 0.35$), nutrient concentrations (λ -values >0.43), pH ($\lambda = 0.31$) and the amount of dissolved oxygen ($\lambda = 0.23$, Figs 5.1b & 5.2). Overall, the presence of *Garra* also tended to increase phytoplankton chlorophyll a ($\lambda = 0.115$) and the relative proportion of Microcystis to total estimated phytoplankton biomass ($\lambda =$ 0.068; Figs 5.1b & 5.2).



Fig. 5.2 The response of abiotic and phytoplankton variables to the presence of fish (*Garra*) in the enclosure experiments in Gereb Awso (GA1: rainy season; GA2: dry season) and Tsinkanet (TS1: rainy season; TS2: dry season): (a–d) suspended matter (SM), (e–h) water transparency (Snell's reading), (i–l) chlorophyll a; (m–p) percentage biomass of *Microcystis*; (q–r) total nitrogen (TN) and (s–t) total phosphorus (TP). F, enclosures with fish; NF, enclosures without fish; Lake, data from the pelagic zone of the lake. Note that no TN or TP data are available for the rainy season experiments (GA1 and TS1).

Effect of Garra on zooplankton communities

The species composition of the zooplankton communities differed considerably among lakes (Fig. 5.1c). The composition of the zooplankton communities in Gereb

Awso were relatively similar among experiments and were mainly composed of *D. carinata, Diaphanosoma, Ceriodaphnia* and cyclopoid copepods (Fig. 5.3). Zooplankton communities in the Tsinkanet enclosures showed pronounced differences among experiments. Tsinkanet zooplankton enclosure communities at the end of the rainy season were mainly dominated by *D. carinata* and *Ceriodaphnia*, whereas enclosure communities at the end of the dry season consisted mainly of *Diaphanosoma* and *Ceriodaphnia* (Figs 5.1c & 5.3).



Fig. 5.3 Population densities of cladoceran taxa in response to the presence of fish (*Garra*) in the enclosure experiments in Gereb Awso (GA1: rainy season; GA2: dry season) and Tsinkanet (TS1: rainy season; TS2: dry season). F, enclosures with fish; NF, enclosures without fish; Lake, data from the pelagic zone of the lake. Note differences in abundance scale between panels.

There was an overall significant effect of *Garra* on cladoceran communities in the four enclosure experiments ($\lambda = 0.026$, F = 2.629, P = 0.012) (Fig. 5.1d).

According to analyses per individual zooplankton taxon, population densities of *Ceriodaphnia* ($\lambda = 0.138$, p = 0.046) and *Diaphanosoma* ($\lambda = 0.120$, p = 0.002) increased upon addition of fish in comparison to non-fish enclosures. Overall treatment effects were not significant for the *Daphnia* species. Apparently, this was because the strength of the response of these species to the fish treatments differed considerably among experiments. Repeated measures ANOVA on the density of *D. carinata* for each of the experiments, separately, revealed negative effects of *Garra* on this species during the rainy season experiment in Tsinkanet and the dry season experiment in Gereb Awso, but not in the rainy season experiment in Gereb Awso (Table 5.1, Fig. 5.3). Furthermore, *Garra* had a significant negative impact on *D. barbata* only in the rainy season experiment in Gereb Awso (Table 5.1).

Overall, there tended to be a negative association between densities of *D. carinata* and the proportion of *Microcystis* relative to total phytoplankton biomass, especially in the enclosures without fish (Fig. 5.4). During the rainy season experiment in Gereb Awso and Tsinkanet, the initial relative abundance of Microcystis dropped along with an increase in the densities of *D. carinata*, except for the enclosures with fish in Tsinkanet, where *D. carinata* densities remained low and *Microcystis* abundance remained relatively high during the entire experiment (averages ranging between 15% and 50% of total phytoplankton biomass). During the first half of the dry season experiment in Gereb Awso, both densities of *D. carinata* and the relative biomass of *Microcystis* increased concomitantly. However, during the second half of the experiment, the share of *Microcystis* in the total phytoplankton biomass declined to less than 5% in the enclosures without fish, whereas it remained at high levels in the enclosures with fish (ranging between 40% and 70%). During the

dry season experiment of Tsinkanet, *Microcystis* dominated the phytoplankton communities heavily in the absence of *D. carinata* (ranging between 95% and 99%).

Table 5.1 Results of repeated measures ANOVA testing for the effects of fish (F), time (T) and the interaction on densities of *Daphnia carinata* and *D. barbata* during experiments at the end of the rainy season (14 September 2005 to 15 October 2005) in Gereb Awso (GA) and Tsinkanet (TS) and the dry season (26 April 2006 to 25 May 2006) in Gereb Awso. No *D. carinata* or *D. barbata* were detected in Tsinkanet during the dry season

Daphnia carinata					Daphnia barbata							
	d.f.	MS	MS	F	р		d.f.	MS	MS	F	Р	
		effect	error					effect	error			
Gereb Awso – rainy season												
F	1,4	13545	21282	0.64	0.470		1,4	7.05	0.822	8.58	0.43	F <nf< td=""></nf<>
Т	7,28	39379	6142	0.64	0.470		6,24	1.00	1.136	0.88	0.524	
F*T	7,28	9762	6142	1.59	0.180		6,24	0.65	1.13	0.57	0.746	
Gereb Awso – dry season												
F	1,4	17216	2773	6.20	0.067	F <nf< td=""><td>1,4</td><td>3.99</td><td>1.26</td><td>3.17</td><td>0.149</td><td></td></nf<>	1,4	3.99	1.26	3.17	0.149	
Т	7,28	8405	823	10.2	0.000		5,20	1.19	0.806	1.48	0.240	
F*T	7,28	2194	823	2.66	0.030	F <nf< td=""><td>5,20</td><td>0.49</td><td>0.806</td><td>0.60</td><td>0.694</td><td></td></nf<>	5,20	0.49	0.806	0.60	0.694	
Tsinkanet – rainy season												
F	1,4	2228	246	9.05	0.040	F <nf< td=""><td>1,4</td><td>20.4</td><td>30.14</td><td>0.68</td><td>0.457</td><td></td></nf<>	1,4	20.4	30.14	0.68	0.457	
Т	7,28	198	41	4.76	0.001		7,28	1385	49.6	27.9	0.000	
F*T	7,28	135	41	3.25	0.012	F <nf< td=""><td>7,28</td><td>49.54</td><td>49.6</td><td>1</td><td>0.453</td><td></td></nf<>	7,28	49.54	49.6	1	0.453	

F<NF refers to significant treatments or treatment x time interaction effects with fish affected population densities negatively.

Chaoborus were observed in most samples of the rainy season (average density in Tsinkanet and Gereb Awso enclosures: 2.7 and 0.6 individuals L^{-1}), but not during the dry season. One-way ANOVA indicated no significant differences in Chaoborus densities between treatments.



Fig. 5.4 The association between Daphnia carinata abundance and the relative biomass of Microcystis in the experiments in Gereb Awso (GA1: rainy season; GA2: dry season) and Tsinkanet (TS1: rainy season; TS2: dry season): (a–d) enclosures without fish, (e–h) enclosures with fish.

Discussion

Garra had a large impact on both biotic and abiotic water column variables in the enclosures. Higher concentrations of macronutrients (nitrogen and phosphorus) in the enclosures with *Garra* than in the fishless enclosures indicate an enhancement of nutrient recycling by fish (Vanni & Layne, 1997; Vanni, Layne & Arnott, 1997). Higher chlorophyll a concentrations and oxygen levels in the enclosures with *Garra* also indicate that higher nutrient availability had stimulated phytoplankton productivity, a response that was also associated with an increase in the relative abundance of the cyanobacterium *Microcystis*. Transparency was also lower in enclosures with *Garra* compared to fishless enclosures, due to the combination of enhanced phytoplankton growth and an increase in the amount of suspended matter, probably originating from the resuspension of sediment particulate matter by the foraging and swimming activity of the fish. Overall, the differences in abiotic conditions and phytoplankton biomass between enclosures with and without fish in our study indicate that *Garra* can potentially have an impact on important ecosystem

characteristics of the reservoirs (water transparency, nutrient availability, phytoplankton primary productivity).

Our results indicate that *Garra* have affected the density and composition of the zooplankton communities mainly indirectly through bottom-up mechanisms, by enhancing nutrient availability and stimulating phytoplankton growth, rather than through a direct top-down control. Abundant zooplankton species, such as Diaphanosoma and Ceriodaphnia, were promoted by the presence of Garra and the positive response of these species to Garra was probably the result of increased food availability. Similar results were obtained by Rejas et al. (2005), who reported increased zooplankton densities in enclosures with fish (Moenkhausia dichroura) along with increased levels of oxygen and phytoplankton biomass in a tropical varzea lake. In our experiments, the lack of a consistent, significant enhancement of Daphnia densities in the fish enclosures, despite increased phytoplankton productivity, may indicate a mild top-down control by Garra. This suggestion is consistent with our observations that brood sizes of *D. carinata* were higher in the presence than in the absence of fish (data not shown). Thus higher fecundity of this species may have been compensated by higher death rates, which would suggest an indirect bottom-up effect of Garra (increased fecundity because of higher food concentration) combined with a moderate top-down control.

In addition to indirect positive effects of fish on zooplankton via the enhancement of phytoplankton productivity, two other potential mechanisms may have contributed to the increased population densities of zooplankton in the fish enclosures: (i) reduced predation by macro-invertebrate predators (Sorano, Carpenter & He, 1993), and (ii) competitive release of competitively inferior taxa because of selective predation on strong competitors (cf. keystone predation). The only potential

162

macro-invertebrate predators we observed in the enclosures of Gereb Awso and Tsinkanet were larvae of the phantom midge *Chaoborus*. These larvae were only present during the rainy season experiment and there were no significant differences in their densities between enclosures with different fish treatments. Reduced predation by macro-invertebrate predators is therefore not a very probable explanation for the observed effects of *Garra*. The second mechanism may have played some role during the dry season in Gereb Awso, but there is no consistent evidence for this in the other experiments.

The higher nutrient concentrations in the enclosures with fish were not only related to higher phytoplankton biomass, but also to a shift in taxon composition within the phytoplankton towards more cyanobacteria (Microcystis). This is in agreement with other studies showing that increased nutrient concentrations are often associated with cyanobacteria blooms (Elser, 1999). The increase in relative importance of *Microcystis* in enclosures with fish may, however, also be partly related to a shift in zooplankton community towards smaller zooplankton at the expense of Daphnia. From our experimental results it is impossible to say whether the negative association between densities of D. carinata and the percentage of Microcystis in the phytoplankton in the reservoirs is due to a toxic effect of Microcystis on Daphnia or rather due to a top-down effect of Daphnia grazing on Microcystis (Reinikainen, Ketola & Walls, 1994). Close inspection of the timing of events, however, provides more support for a top-down than for a bottom-up impact. The observations in the dry season experiment in Gereb Awso (Fig. 5.4b,f), for instance, suggest a top-down effect, because both taxa first show an overall increase during the first 10 days of the experiment, indicating that Daphnia was able to grow to higher densities in the presence of an actively growing *Microcystis* population. Then, as *Daphnia* continued to increase in densities, the *Microcystis* population started to crash, but only in the absence of fish (Fig. 4b). In the presence of fish (Fig. 5.4f), the growth of *Daphnia* levelled off at approximately 60 individuals L^{-1} , whereas *Microcystis* abundance remained high. In the other experiments, reductions in *Microcystis* dominance and increases in *Daphnia* densities occurred concomitantly, especially in the absence of fish. This is also more in agreement with a top-down control (grazing immediately impacting algae densities) than with a toxicity effect of *Microcystis* on *Daphnia*. If *Microcystis* toxins impacted *Daphnia* populations negatively by affecting their fertility and mortality rates, we would expect a recovery of *Daphnia* populations following a *Microcystis* decline only after some time lag.

The enclosures used in our experiment were self contained and did not allow interaction of the fish with lake sediments. This may have resulted both in an underand overestimation of the effect of *Garra* on the pelagic food web. The absence of real lake sediments on the floor of the enclosures may have resulted in lower rates of sediment re-suspension and associated nutrient recycling by the foraging and swimming activity of fish compared to the fishless enclosures and the reservoirs. Despite this potential limitation, we were still able to show convincingly the effects of *Garra* on water column concentrations of nutrients, suspended matter and chlorophyll a. Conversely, the absence of lake sediments in the enclosures may have deprived the fish of access to benthic food resources, thereby forcing them to forage on zooplankton. This could have resulted in an overestimation of the top-down control of *D. carinata* and resulting trophic cascade effects on *Microcystis*. Our results are, nevertheless, overall very consistent with patterns that were detected during a survey study of 32 reservoirs (Dejenie et al., 2008). In that survey, we recorded a positive association between *Garra* biomass and total phosphorus, an association that is undoubtedly due to the higher carrying capacity of phosphorus-rich reservoirs for *Garra* populations, but which may also reflect a higher recycling of phosphorus by high densities of *Garra* (see also this study). Furthermore, the survey yielded a positive association of *Garra* with the estimated biomass of cladocerans, which is in agreement with the observation of enhanced densities of *Diaphanosoma* and *Ceriodaphnia* in enclosures with *Garra*. The survey study also suggested a moderate top-down impact of *Garra* on cladocerans due to the lack of an association between phytoplankton and cladoceran biomass (see Dejenie et al., 2008), which is in line with the present findings of a subtle top-down effect of *Garra* on *Daphnia*. Furthermore, the field survey (Dejenie et al., 2008) also revealed a negative association between *Daphnia* and cyanobacteria biomass, suggesting a *Daphnia*-mediated trophic cascade effect.

The reason why the fish in our experiments and in the reservoirs exert only a limited top-down effect is likely related to the fact that they are opportunistic, largely benthivorous riverine fish that are not very well adapted to feed on zooplankton. This also indicates the existence of an unoccupied niche for zooplanktivorous fish in the majority of the reservoirs. Detailed studies on the diet of *Garra* and whole lake manipulation experiments are presently being undertaken to further broaden our knowledge on the ecology of *Garra* and its impact on reservoir food web structure and dynamics (Mekonen et al., unpubl. data).

Acknowledgments

This study was financially supported by the Flemish Institutional University Cooperation (VLIR), through the VLIR inter-institutional cooperation between Mekelle University and Flemish universities. We thank Dr Nurhusen Teha and Dr Mulu Bayray, the program leader and program manager at Mekelle University, and the MU-IUC staff for their continuous support to this study. We especially thank villagers of the Gereb Awso and Tsinkanet areas, as well as local authorities for their support. During the fieldwork period, J.N. was at K.U.Leuven (Belgium) with working place Mekelle University, Ethiopia. S.A.J.D. is a postdoctoral researcher with the National Fund for Scientific Research, Flanders (FWO-Vlaanderen).

References

- Anderson, J.M. and J.S.I. Ingram, 1989. TSBF: A Handbook of Methods of Analysis. CAB International, Wallingford, U.K.
- Benzie, J.A.H., 2005. Cladocera: the genus Daphnia (including Daphniopsis). In: Guides to the Identification of the Microinvertebrates of the Continental Waters of the World (Ed H.J.F. Dumont), pp. 376. Backhuys Publishers, Leiden.
- Cooke, R.G., B.E. Welch, A.S. Peterson and R.P. Newroth, 1993. Restoration and Management of Lakes and Reservoirs, 2nd edn. Lewis Publishers, New York.
- Dejenie, T., Tsehaye, A., L. De Meester, Afework, M., Abreha, G., G.S. Risch, A.
 Pals, K. van der Gucht, W. Vyverman, J. Nyssen, J. Deckers, and S. Declerc, 2008. Limnological and ecological characteristics of tropical highland reservoirs in Tigray, Northern Ethiopia. *Hydrobiologia*, 610: 193 209
- Elser, J.J., 1999. The pathway to noxious cyanobacteria blooms in lakes: the food web as the final turn. *Freshwater Biology*, **42**: 537–543

- Flossner, D., 2000. Die Haplopoda und Cladocera (ohne Bosminidae) mitteleuropas. Backhuys Publishers, Leiden, 428 pp.
- Getahun, A., 2000. Systematic Studies of the African species of the Genus Garra (Pisces: Cyprinidae). PhD Thesis, CUNY, New York.
- Getahun A. & Stiassny M.L.J. (1998) The freshwater biodiversity crisis: the case of the Ethiopian fish fauna. SINET: Ethiopian Journal of Sciences, 21, 207–230.
- Golubtsov, A.S., Y. Degbuadze and M.V. Mina, 2002. Fishes of the Ethiopian rift valley lakes. In: Ethiopian Rift Valley Lakes (Eds. C. Tudorancea & W.D. Taylor), pp. 167–258. Backhuys Publishers, Leiden.
- Haney, J.F. and D.J. Hall, 1973. Sugar-coated Daphnia: a preservation technique for Cladocera. *Limnology and Oceanography*, 18: 331–333
- Hillebrand, H., L.D. Durselen, D. Kirschtel, U. Pollingher and T. Zohary, 1999.
 Biovolume calculation for pelagic and benthic microalgae. *Journal of Phycology*, 35: 403–424
- John, D.M., 2002. Order Cladophorales (=Siphonocladales). In: The Freshwater Algal Flora of the British Isles. An Identification Guide to Freshwater and Terrestrial Algae (Eds. D.M. John, B.A. Whitton & A.J. Brook), pp. 468–470. Cambridge University Press, Cambridge.
- Kalff, J., 2002. Limnology Inland Water Ecosystem. Prentice Hall, New Jersey.
- Komarek, J. and K. Anagnostidis, 2000. Cyanoprokaryota: Chroococcals. Spektrum Akademischer Verlag, Heidelberg-Berlin, Germany.
- Komarek, J. and K. Anagnostidis, 2005. Cyanoprokaryota: Oscillatorials. Spektrum Akademischer Verlag, Heidelberg-Berlin, Germany.
- Leps^{*}, J. and P. S^{*}milauer, 2003, Multivariate Analysis of Ecological Data Using CANOCO. Cambridge University Press, Cambridge.
- Menden-Deuer, S. and E.J. Lessard, 2000. Carbon to volume relationships for dinoflagellates, diatoms, and other protist plankton. *Limnology and Oceanography*, 45: 569–579
- Reinikainen, M., M. Ketola and M. Walls, 1994. Effects of the concentrations of toxic *Microcystis aeruginosa* and an alternative food on survival of *Daphnia pulex*. *Limnology and Oceanography*, **39**: 424–432
- Rejas, D., S. Declerck, J. Auwerkerken, P. Tak and L. De Meester, 2005. Plankton dynamics in a tropical floodplain lake: fish, nutrients, and the relative importance of bottom-up and top-down control. *Freshwater Biology*, **50**: 52–69
- Sorano, P.A., S.R. Carpenter and X. He, 1993. Zooplankton biomass and body size.
 In: The Trophic Cascade in Lakes (Eds. S.R. Carpenter & J.F. Kitchell), pp. 172–188. Cambridge University Press, New York.
- Ter Braak C.J.F. and P. S^{*}milauer, 2002. Canoco Reference Manual and CanoDraw for Windows User's Guide: Soft-ware for Canonical Community Ordination (version 4.5). Microcomputer Power, Ithaca, NY.
- Vanni, M.J. and C.D. Layne, 1997. Nutrient recycling and herbivory as mechanisms in the 'top-down' effect of fish on algae in lakes. *Ecology*, **78**: 21–40
- Vanni, M.J., C.D. Layne, and S.E. Arnott, 1997. Top-down trophic interactions in lakes: effects of fish on nutrient dynamics. *Ecology*, 78: 1–20
- Whitford, L.A. and G.J. Schumacher, 1973. A Manual of Freshwater Algae. Sparks Press,

Appendix 5.1

The impact of fish on phytoplankton and cyanobacterial blooms in semi-arid tropical highland reservoirs in Tigray: analysis of phytoplankton community taxa

Introduction

Chapter 5 of this thesis deals with the results of the field enclosure experiments carried out to assess the impact of fish on the ecology of reservoirs and the occurrence of *Microcystis* blooms in semi-arid highland reservoirs in Tigray, Ethiopia. This manuscript is published online in the *Freshwater Biology*. In this appendix, we present part of the data and results of the enclosure experiments which were not included in this manuscript. We hear by present in this appendix the results of analysis for the phytoplankton community composition *Microcystis* biomass and genetic structure. We also present the methods for the DGGE profiling of the cyanobacteria. The experimental design, collection and analysis of samples, and statistical analysis are described in chapter 5. Finally we will also present some additional discussion on the effect of Garra on the phytoplankton composition communities.

Materials and methods

To study the *Microcystis* ITS rDNA population structure, water samples were manually filtered over a 25mm 0.2 μ m GSWP filter (Millipore) until the filter was clogged. The filter was folded two times and packed with sterile aluminium foil. Filters were transported on ice packs to the laboratory and stored at -20 °C until further processing. Genomic DNA from the GSWP filters containing field samples was extracted following the protocol described by Zwart et al. (1998), which includes beat-beating with phenol extraction and ethanol precipitation. After extraction, the DNA was purified on a Wizard column (Promega).

A specific nested-PCR protocol (Van Gremberghe et al. 2009) was used to specifically amplify only *Microcystis* ITS sequences from the water samples. In a first

PCR, a specific 16S rDNA primer for *Microcystis* (CH) described by Rudi *et al.* (1997) was used as forward primer and a general 23S rDNA primer (ULR) was used as reverse primer (Janse *et al.*, 2003). For the second PCR the cyanobacterium-specific 16S rDNA primer (GC)-CSIF (with GC-clamp for DGGE-analysis) in combination with primer ULR (Janse *et al.*, 2003) was used. For the composition of the reaction mixtures of the PCR's we refer to Van Gremberghe et al. (2009).

DGGE profiling

DGGE was essentially performed as described by Muyzer *et al.* (1993). Equal amounts of PCR products were loaded onto 8% (w/v) polyacrylamide gels (1 mm thick, in 1× TAE [20 mM Tris acetate (pH 7.4), 10 mM acetate, 0.5 mM disodium EDTA]). The denaturing gradient contained 35-40% denaturant [100% denaturant corresponded to 7 M urea and 40% (v/v) formamide]. Electrophoresis was performed for 16 h at 75 V and the temperature was set at 60°C. Finally, the gels were stained with ethidium bromide and photographed on a UV transillumination table with a CCD camera. Furthermore, a small piece of gel from the middle of the target band was excised from the DGGE gel for sequencing and incubated in 50 µl of sterile TE buffer (10 mM Tris, pH 7.6, 1 mM EDTA) for 24 h at 4 °C. The eluent was then reamplified and purified on DGGE one or two times. The resulting PCR products were purified using a QiaQuick PCR purification kit (QiaGen). Finally, sequencing was performed with the ABI-Prism sequencing kit and the resulting sequencing reaction products were analysed on an automatic sequence (ABI-Prism 3100).

On every gel, three standard lanes were analysed in parallel to the samples. These standard lanes are composed of several known bands positioned distinctly in the gel. As these bands should always be formed at the same denaturant concentration in the gel, their position was used to compare the patterns formed in different gels. Digitalized DGGE images were analysed using the software package BioNumerics 4.5 (Applied Maths BVBA). The program detects the bands and groups the bands into band classes, based on their position in the gel. Sequence information of the bands was used to manually check the grouping of bands into band classes. Further, a matrix was compiled, based upon the band intensity or the presence or absence of bands in band classes. The band intensities were then converted into relative intensities (the relative contribution of each band to the total band signal in the lane).

Statistical analyses

We performed a Principal Component Analyses (PCA) and redundancy analysis (RDA) on the matrix with log transformed biomasses of the different phytoplankton taxa and the ITS types of *Microcystis* following the procedure described in methods of this chapter. For details of the statistical analysis, we refer to the methods section of this chapter (chapter 5). Finally, we executed repeated measures ANOVA in Statistica 7.0 to test for the effect of fish on the biomasses of the different taxa in all enclosure experiments separately.

Results

1. Effect of fish on phytoplankton community structure

The biomass of the different phytoplankton taxa in the two experiments carried out in Gereb Awso and Tsinkanet is shown in Fig. 5.5. The abundance of phytoplankton taxa differed considerably among lakes and among enclosure experiments (Fig. 5.6a). An overall increase in the biomass of phytoplankton in general and the biomass of *Microcystis* in particular was observed in the April experiment (Fig. 5.5).



Fig. 5.5 Biomass of the different phytoplankton taxa in Gereb Awso (GA) and Tsinkanet (TS) experiments 1 and 2.

We observed an overall significant effect of *Garra* on the phytoplankton communities in the four enclosure experiments (Trace= 0.08, F= 4.3, p= 0.032) (Figure app. 5. 6b). Analyses performed on individual phytoplankton taxa indicated that *Garra* significantly increase diatom (T= 0.238, F = 14.778, p = 0.002), *Microcystis* (T = 0.249, F = 12.791, p = 0.002), and total phytoplankton biomass (T = 0.271, F = 11.229, p = 0.004). Fish treatment effects were not significant for Chlorophytes, Cryptomonas and Euglenoids. According to the repeated measures ANOVA there was a significant effect of *Garra* on *Microcystis* biomasses in experiments of the rainy season carried out in both Gereb Awso and Tsinkanet (Table 5.2). There was also a significant effect of fish on diatoms and total phytoplankton for the rainy season experiment in Tsinkanet.



Fig. app. 5.6 (a)Triplot of PCA results performed on biomass of phytoplankton taxa in the four enclosure experiments. (b) Triplots of RDA results, showing the response to the presence of fish (Garra) of the biomass of phytoplankton taxa in the four enclosure experiments. Individual experiments were specified as co-variables. The nominal variables Fish and Non Fish in the RDA plot are represented by centroids. Circles represent enclosure with fish, squares represent fishless enclosures. F: with fish, NF: no fish; GA1and TS1 Gereb Awso and Tsinkanet September experiment respectively; GA2 and TS2: Gereb Awso and Tsinkanet April experiment respectively. Filled triangles represent centroids and show the average position of replicate enclosures in each treatment level.

Table 5.2 Results of repeated measures ANOVA on log transformed biomass date testing for the interaction effect of 'Garra' (Fish = F) and 'time' (T) on total phytoplankton biomass, biomass of Diatoms, and Microcystis during enclosure experiment 1 (September/October, 2005) and experiment 2 (April/May, 2006) in Gereb Awso (GA) and Tsinkanet (TS). F= fish, T= time and F*T= fish and time interaction.

Microcystis					Diatoms							
	d.f.	MS	MS	F	р		d.f.	MS	MS	F	р	
		effect	error					effect	error			
Gereb Awso – rainy season												
F	1,2	32.391	0.702	46.081	0.02	F>NF	1,4	5.734	1.431	4.007	0.11	
Т	5	7.471		2.705	0.08		6	9.625		19.33	< 0.01	
F*T	5,10	2.884	2.761	1.044	0.44		6,24	0.830	0.497	1.667	0.17	
Gereb Awso – dry season												
F	1,4	7.218	3.168	2.278	0.27		1,2	0.404	0.353	1.146	0.396	
Т	6	2.443		1.753	0.19		5	0.548		4.535	0.020	
F*T	6,24	0.761	1.393	0.546	0.76		5,10	0.140	0.121	1.163	0.391	
Tsinkanet – rainy season												
F	1,4	5.297	4.603	1.150	0.34		1,4	4.004	0.339	11.799	0.026	F>NF
Т	6	7.729		2.521	0.04		6	12.01		47.529	< 0.01	
F*T	6,24	10.85	3.065	3.540	0.01	F>NF	6,24	0.361	0.252	1.432	0.244	
Tsinkanet – dry season												
F	1,4	0.364	0.184	1.978	0.23		1,4	3.223	0.919	3.506	0.134	
Т	6	3.934		71.057	< 0.01		6	4.915		2.495	0.051	
F*T	6,24	0.104	0.055	1.874	0.13		6,24	2.014	1.969	1.023	0.434	
Total phytoplankton Total phytoplankton												
Gereb Awso – rainy season Tsinkanet – rainy season												
F	1,4	2.052	1.283	1.599	0.27		1,4	1.171	0.121	9.66	0.034	F>NF
Т	6	2.278		2.522	0.05		6	1.890		59.71	< 0.01	
F*T	6,24	0.937	0.903	1.037	0.43		6,24	0.295	0.032	9.32	< 0.01	F>NF
Gereb Awso – dry season Tsinkanet – dry season												
F	1,2	1.535	0793	1.934	0.30		1,4	0.355	0.197	1.801	0.25	
Т	6	0.996		3.644	0.03		6	3.789		75.10	< 0.01	
F*T	2,12	0.06	0.273	1.123	0.41		6,24	0.097	0.050	1.916	0.11	
F>NF	refers	to sign	ificant	treatments	or tre	eatment	x tim	e intera	ction eff	fects wit	h fish a	affected

population biomasses positively

2) Effect of fish on Microcystis ITS types

Twenty band classes (equivalent to different ITS types) were detected with the PCR amplified DGGE profiles of the 16S rDNA. Most of the band classes were, however encountered in one or two enclosures only once. We used, therefore, only the four band classes which were relatively abundant in all four enclosures. We used sequence information to verify whether the bands in a band class represent the same ITS

sequence. The ITS types tended to differ among lakes and among enclosures with no pattern in enclosures with fish and fishless enclosures showed among the September and April experiments (Fig. 5.7a). Overall the RDA (Fig. 5.7b) of the effect of *Garra* on the relative abundance of the ITS types was not significant (Trace = 0.058; F = 1.115; p = 0.338). This may result from the inconsistent presence of the different ITS types in the different enclosures and lakes. We further carried out RDA separately for the experiments in Gereb Awso and Tsinkanet. We observed an increase in the p-value (not significant) of the Tsinkanet experiment by 0.0198 (Trace 0.172, F 2.468, p 0.884). ITS type 1 seem to increase with the addition of fish. We also performed Mann-Whitney U tests for effects of *Garra* on the four different ITS types. We observed a significant effect of *Garra* on ITS type 1 (U=0.0000, Z=1.964, p=0.0496) in the dry season (April) experiment in Tsinkanet.



Fig. 5.7 Triplot of PCA results on the Microcystis ITS in the four enclosure experiments (GA1, GA2, TS1, and TS2) (a) and RDA triplots on Microcystis ITS (b-c): (b)the four enclosure experiments, (c) Gereb Awso experiments (GA1 and GA2) and (d) Tsinkanet experiments (TS1 and TS2). Circles represent enclosure with fish; squares represent fishless enclosures. F: with fish, NF: no fish; GA1and TS1 Gereb Awso and Tsinkanet rainy season experiment respectively; GA2 and TS2: Gereb Awso and Tsinkanet dry season experiment respectively. Filled triangles represent centroids and show the average position of replicate enclosures in each treatment level.

Discussion

Our results indicate that the presence of *Garra* markedly influenced phytoplankton biomass and *Microcystis biomass*. But no significant consistent influence was observed for the ITS types of *Microcystis*. We have reported (this chapter) an increase

in nutrient concentration and suspended matter and a decrease in transparency after addition of fish. Increased levels of dissolved oxygen and pH were also measured in the fish enclosures. This is an indication for a higher phytoplankton primary productivity in the presence of fish.

The increase in phytoplankton primary productivity associated with *Garra* is probably the result of increased nutrient concentrations (total phosphorus and total nitrogen) caused by fish. Reports indicate that phytoplankton community structure is frequently affected by the presence of fish (Vanni & Layne, 1997), resulting from a consumer mediated nutrient recycling. The higher nutrient concentrations in the fish compared to the fishless enclosures was not only related to a higher phytoplankton towards more *Microcystis* and diatoms. This is in agreement with other studies showing that increased nutrient concentrations are often associated with cyanobacterial blooms (Elser, 1999). *Microcystis* was the dominant phytoplankton taxon in the experiments, with an increase in biomass due to the addition of fish. The fish-zooplankton-*Microcystis* interactions are discussed in chapter 5.

We could not establish clearly the effect of fish on different strains of *Microcystis* represented by ITS types. It seemed that different strains (species) of *Microcystis* respond differently to the different treatments especially in Tsinkanet.

- Elser, J.J., 1999. The pathway to noxious cyanobacteria blooms in lakes: the food web as the final turn. Freshwater Biology, 1999: 537 543
- Janse, I., M. Meima, W.E.A. Kardinaal, and G. Zwart, 2003. High resolution differentiation of cyanobacteria using rRNA-ITS DGGE. *Appl Environ Microbioi.*, 69: 6634–6643
- Muyzer, G., E. C. De Waal, and A. G. Uittierlinden, 1993. Profiling of complex microbial populations by Denaturing Gradient gel Electrophoresis analysis of Polymerase Chain Reaction-amplified genes coding for 16S rRNA. *Applied and Environmental Microbiology*, **59**: 695-700
- Rudi, K. O. M. Skulberg, F. Larsen, and K. S. Jakobseni, 1997. Strain characterization and classification of Oxyphotobacteria in clone cultures on the basis of 16S rRNA sequences from the variable regions V6, V7, and V8. *Applied and Environmental Microbiology*, 63/ 2593–2599
- Van Gremberghe, I., J. Van Wichelen, K. Van der Gucht, P. Vanormelingen, S. D'hondt, C. Boutte, A. Wilmotte, and W. Vyverman, 2009. Co-variation between zooplankton community composition and cyanobacterial community dynamics in Lake Blaarmeersen (Belgium). *FEMS Microbiol. Ecol.*, 63: 222-237
- Vanni, M.J., and C.D. Layne, 1997. Nutrient recycling and herbivory as mechanisms in the top-down effect of fish on algae in lakes. *Ecology*: **78**: 21-40
- Zwart G., R. Huismans, M.P. van Agterveld, Y. van de Peer, P. De Rijk, H. Eenhoorn,
 G. Muyzer, E.J. van Hannen, and H.J. Laanbroek, 1998. Divergent members of
 the bacterial division *Verrucomicrobiales* in a temperate freshwater lake. *FEMS Microbial. Ecol.*, 25: 159-169

Chapter 6

An experimental assessment of potential top-down effects of zooplankton on the phytoplankton communities, including toxic cyanobacteria, of eutrophic shallow reservoirs in Tigray, North Ethiopia

Tsehaye Asmelash¹, Ineke van Gremberghe², Tadesse Dejenie¹, Pieter Vanormelingen², Teklit Gebregiorgis¹, Belay Gebreyohannes¹, Steven Declerck⁻³, Luc De Meester³, Katleen Van der Gucht², Wim Vyverman²

(Unpublished manuscript)

¹ Mekelle University, P.O Box 231, Mekelle, Ethiopia

² Laboratory of Aquatic Ecology and Protistology, Gent University, Krijgslaan 281-S8, B-9000 Gent, Belgium

³ Laboratory of Aquatic Ecology and Evolutionary Biology, KULeuven, Ch. Deberiotstraat 32, 3000 Leuven, Belgium

Abstract

An enclosure experiment was carried out to study the effect of zooplankton grazing on phytoplankton community structure, including the relative abundance of toxic cyanobacteria, in a semi-arid shallow reservoir in Tigray, North Ethiopia. Twelve 80 L buckets, suspended in the lake, were assigned to four treatments that represented factorial combinations of source water (34 μ m filtered water from two reservoirs) and presence/absence of zooplankton. Plankton net concentrated phytoplankton was inoculated in equal densities to all buckets. Top-down regulation by zooplankton was observed for some of the phytoplankton taxa, including Anabaena, Euglenoids, Diatoms, Chlorophytes and Cryptomonads, whereas the impact of the presence of zooplankton on *Microcystis* and *Peridinum* biomass was limited. Microcystin was detected from all experimental units, and concentrations ranged from 0.082 to 1.268 μ g L⁻¹. Higher microcystin concentrations were detected in the treatment with than without zooplankton. A positive association between zooplankton and microcystin concentration per unit biomass of *Microcystis* was observed. Overall, our results suggest that the presence of zooplankton stimulates toxin production.

Introduction

Efforts to understand the complex relationships between algae and herbivorous zooplankton have been a central focus of aquatic ecology. In recent years, the response of algal community to herbivorous zooplankton is considered to result from the net effect of ingestion by zooplankton and growth stimulation due to nutrient recycling (Carpenter & Kitchell, 1984, Lehman & Sandgren, 1985, Bergquist et al., 1986, Sterner, 1986). Zooplankton community composition is an important factor influencing algal grazing losses. The nature and intensity of grazing pressure on the phytoplankton community depend on numerous aspects of the zooplankton community. Elser et al. (1987) reported that algae respond more strongly to changes in zooplankton composition than to changes in zooplankton biomass. Different studies have demonstrated that zooplankton selective feeding can impose differential mortality on algal taxa (Bogdan and McNaught, 1975, Lampert and Taylor, 1985), and that algal taxa respond differentially to grazing in ways not strictly related to cell size (Porter, 1975, Havens and DeCosta, 1985, Bergquist et al., 1986, Elser, 1999). Opposite shifts of phytoplankton size distribution were observed by Bergquist et al. (1985) from grazing by two zooplankton assemblages with different size distribution. Density of small phytoplankton taxa increase when grazed by small zooplankton, but decreased when grazed by large zooplankton. Conversely, large phytoplankton became less abundant in the presence of small zooplankton, but increased in density in the presence of large zooplankton. This is explanation for the contrasting size shifts of the phytoplankton are differences in grazing selectivity (Peters & Downing, 1984) and nutrient recycling rate (Peters, 1983). Consequently, the nature of the interaction between phytoplankton and their grazers is dependent on the composition of the algal community as well as on overall nutrient availability and the composition of the zooplankton. *Daphnia* are automatic, relatively unselective filter feeding crustaceans (MacMahon & Reeve, 1965); but copepods feed discontinuously and show considerable discrimination when presented with a choice of several organisms. Selective feeding of copepods may be based on quality of food but they also apparently tend to pick large sized particles when given a choice (Mullin, 1963, Richman and Rogers, 1969). The difference in feeding behaviour between cladocerans and copepods is reported to cause complementary impact on phytoplankton (Sommer et al., 2001, 2003); cladocerans suppress small phytoplankton, while copepods suppress large phytoplankton. Nutrients released by grazing zooplankton are available for algal uptake (Lehman and Scavia, 1982) and, in some situations, can provide sufficient nutrients to satisfy a substantial proportion of the growth requirement of the algal community (Lehman, 1980, Axler *et al.*, 1981, Sterner, 1986, Elser, 1999). Nutrient recycling by herbivores is also size- and speciesdependent (Lehman, 1980, Elser, 1999).

Cyanobacterial blooms are increasingly frequent in aquatic ecosystems around the world as a result of eutrophication (Chorus & Bartram, 1999, Huisman et al., 2005) and are likely to increase with global warming. Factors such as nutrient concentrations (nitrogen and phosphorus), temperature, light, pH and alkalinity, buoyancy, selective zooplankton grazing, hydrologic and meteorological conditions, and the morphology of the impoundment have all been implicated (Shapiro, 1984, Chorus & Bartram, 1999) as factors contributing to cyanobacterial bloom formation. Interactions between bloom-forming cyanobacteria and zooplankton are important to study considering the potential of top-down control of cyanobacterial blooms (Burns, 1987, Lampert, 1987, Sommer, 1989). Consequently, a large body of research has been directed at understanding the mechanisms by which cyanobacteria affect zooplankton and vice versa. This information has important implications for understanding planktonic community structure and function, as well as water quality in lakes. For example, the filamentous morphology of certain cyanobacteria has been shown to negatively affect large cladocerans more than small cladocerans through reduced fecundity (Webster and Peters, 1978, Gilbert, 1990), a mechanism that may help to explain declines in the dominance of large cladocerans with lake eutrophication (Jeppesen *et al.*, 2000). Two major mechanisms have been proposed for the greater susceptibility of large cladocerans to colonial or filamentous blue-green algae (Gliwicz, 1977, Webster & Peters 1978, Richman & Dodson, 1983, Porter & McDonough, 1984). First, colonial or filamentous algae clog the filtering appendages of larger cladocerans, reducing their feeding rates on co-occurring nutritious food sources or increasing their respiration rates or both. And second, larger cladocerans more readily ingest colonial or filamentous blue-greens and thereby are more strongly affected by any toxic chemicals that these algae may possess.

Bloom-forming cyanobacteria are in general considered a poor food source for herbivorous zooplankton (Porter and Orcutt, 1980). Three properties of cyanobacteria have been proposed to account for this poor quality: filamentous/colonial morphologies, production of intracellular secondary metabolites with toxic properties, and deficiencies in essential nutrients (Porter & Orcutt, 1980, Lampert, 1987, DeMott, 1989). Of these, morphology and toxins have attracted most interest (Lampert, 1982, Fulton & Paerl, 1987b, Lürling, 2003). In addition, a number of cyanobacterial genera produce a wide range of toxins, causing a potential risk to livestock and human health (Resson et al., 1994, Chorus & Bartram, 1999, Carmichael et al., 2001). Several reports suggest that cyanobacterial toxins act as a defence against zooplankton grazing by reducing the feeding rate of zooplankton (Lampert, 1981, Nizan *et al.*, 1986, Fulton & Paerl, 1987a, DeMott, 1999). An in increase toxin production by *Microcystis aeruginosa* in response to direct and indirect exposure to herbivorous zooplankton of several species has been reported by Jang et al. (2003); in support to the hypothesis that this response is an induced defence mediated by the release of chemicals called "infochemicals" from zooplankton. These are chemical signals released by predators that can induce defences in aquatic prey (Bronmark & Hansson, 2000). Defence mechanisms such as colony formation, regulation of growth rate and toxin production could be induced in *M. aeruginosa* by one or more infochemicals released by zooplankton (Larsson & Dodson, 1993, Jang et al., 2003, 2008).

Different strains of the same phytoplankton species may coexist in the phytoplankton community (Kirk and Gilbert, 1992; Barreiro et al., 2007). These strains may differ in their toxic properties, a case documented for toxic cyanobacteria (Laamanen et al., 2001). It is also reported that some zooplankton organisms are able to select between algae differing in toxicity (Huntley et al., 1986, DeMott, 1989, Gilbert, 1990, DeMott and Moxter, 1991, Kirk and Gilbert, 1992, Teegarden, 1999). The ability of grazers to select non-toxic food is a mechanism which favours toxic algae against non-toxic algae, and is possibly one of the mechanisms favouring bloom initiation in toxic phytoplankton species (Gilbert, 1990, Guisande et al., 2002). However, zooplankton grazers could be less able to select between coexisting toxic and non toxic strains of the same species. In this aspect, the benefits of producing toxic compounds in order to avoid grazing may be minimal.

In this study, a short-term enclosure experiment was carried out to study the effect of zooplankton on the phytoplankton community, including toxic cyanobacteria, in a semi-arid shallow reservoir in Tigray, North Ethiopia. The central question addressed was to what extent a top-down effect of zooplankton grazing on cyanobacteria can regulate bloom formation and to what extent zooplankton influences microcystin production. In addition, we wondered if the reservoir origin of the water used in the experiment has implications for the phytoplankton community composition and how this interacts with the presence of zooplankton. Therefore, phytoplankton was collected from two different reservoirs, mixed, and grown in the two water sources of which phytoplankton was collected, in the presence or absence of zooplankton.

Materials and methods

Experimental design

An enclosure experiment was carried out in April 2007 in Mai Gassa I reservoir, (13° 17' 13.5" N, and 39° 29' 25" E, Tigray, North Ethiopia), a shallow reservoir characterized by both the presence of algal blooms and submerged vegetation (see Dejenie et al., 2008). The enclosures consisted of synthetic polyethylene plastic cone shaped buckets (top diameter: 40 cm, bottom diameter: 30 cm, depth: 80 cm, volume: 80 L) filled with plankton net (34 μ m) filtered lake water. The enclosure buckets were reinforced, fixed in a wooden frame, and positioned at the centre of the lake with anchors. Before filling, the twelve bucket enclosures were randomly assigned to four treatments (two source waters and presence/absence of zooplankton) according to a factorial design: 1) three enclosures received filtered Mai Gassa I water without

zooplankton (MGZ-), 3) three received filtered water from a second reservoir Era Quihila water and zooplankton (EQZ+), and 4) three received filtered Era Quihila water without zooplankton (EQZ-). The enclosures were set up on April 16, 2007. The experiment lasted for 16 days. The two source reservoirs for filtered water, phytoplankton and zooplankton, Mai Gassa I and Era Quihila, are both characterized by cyanobacterial blooms, but the blooms are more pronounced in Era Quihila than in Mai Gassa I. Mai Gassa I is also characterized by the presence of submerged vegetation. All buckets were inoculated with a mixture of phytoplankton collected with a plankton net (34 µm mesh size) from Mai Gassa I and Era Quihila reservoirs. Equal amounts of phytoplankton from the reservoirs were added to the mixture prepared as inocula by measuring chlorophyll a with a fluorometer (Turner Aquafluor) set on the chlorophyll mode. For the treatments with zooplankton, Daphnia from a third reservoir, Adi Amharay, were collected with a zooplankton net and introduced into the zooplankton treatments. Adi Amharay was selected as a source of the zooplankton inocula because *Daphnia*, the zooplankton grazer with potentially the strongest top-down influence on phytoplankton communities was exceptionally dominant in this reservoir. These Daphnia represent enrichment to the zooplankton that was introduced in the buckets together with the phytoplankton inoculum. The zooplankton inoculum was equal across all experimental units of the zooplankton treatment, and represented a mixture of zooplankton of Mai Gassa I and Era Quihila supplemented with *Daphnia* from Adi Amharay reservoir.

Sampling and analysis of samples

Phytoplankton, zooplankton, and water samples for quantification of toxin concentrations were collected from the enclosures on days 0, 2, 4, 8 and 16. Samples

were collected with a tube sampler. Water samples (250 ml) for phytoplankton counting were preserved with acid lugol's solution, while zooplankton samples (2 L filtered over 34 μ m) were immediately fixed with a formalin-sucrose solution (4% final concentration) to prevent egg loss from cladoceran carapaces (Haney & Hall 1973). Phytoplankton biomass was estimated from cell bio-volume measurements and previously published biovolume-to-carbon conversion data (Menden-Deuer & Lessard, 2000). Zooplankton was counted at the start of the experiment (d 0) and on days 4 and 8. The counts were used to calculate densities (individuals per litre) and biomass. Biomass estimates were obtained using published length-weight regressions (Bottrell et al., 1976).

Detection of toxins

The total microcystin concentration (μ g ml⁻¹) in the samples was determined by ELISA (Enzyme Linked Immunosorbent Assay). For the extraction of microcystins, the cells were boiled for 20 minutes (van der Oost, 2007), and then centrifuged for 10 minutes at 14,000 rpm after which the supernatant was used for ELISA. The ELISAtest was done according to the manufacturer's instructions (SDI - EnviroGard Microcystin ELISA plate kit). The microcystin concentration was recalculated into concentration per biomass (μ g microcystin per mg C *Microcystis* biomass). Due to accidental loss of part of the samples, toxin concentrations could only be measured on day 4 of the experiment.

Data analysis

Prior to statistical analysis, all variables were Log (X+1) transformed. Multivariate analyses were used to investigate the relation between phytoplankton taxon

composition, with emphasis on potentially toxic cyanobacteria, and experimental treatment (source water and zooplankton). We performed redundancy analysis (RDA) to test the overall effect of zooplankton and source water on the phytoplankton community composition. The RDA was performed in CANOCO version 4.5 for Windows (Ter Braak & S'milauer, 2003). Treatments effects were further explored in more detail by applying additional tests to each of the phytoplankton genera separately. We executed two way analysis of variance (ANOVA) on log transformed biomasses of different phytoplankton taxa for those we observed significant effect by the regression analysis. Finally, spearman rank correlation coefficients were calculated between the biomass of different phytoplankton taxa and the different zooplankton species to check for specific associations. The two way ANOVA was also used to investigate the significance of the differences in microcystin concentration (mg/L) between the zooplankton and water source treatments. All univariate analyses were carried out with the software Statistica version 6.0 for Windows (StatSoft Inc., Tulsa, USA).

Results

The filtered water that was used to fill the enclosures was not completely free from phytoplankton $(61 \times 10^3 \text{ pg C ml}^{-1} \text{ in Mai Gassa I filtrate and } 246 \times 10^3 \text{ pg C ml}^{-1} \text{ in Era}$ Quihila filtrate), but the filtration procedure enabled us to start with comparable phytoplankton biomass and taxon composition in the two treatments, as the inoculum that was added to all experimental units was much more dense than the remaining phytoplankton in the filtrate. Total phytoplankton biomass was similar in the two source reservoirs, but the taxon composition differed. Fig. 6.1 shows the taxon composition of the experimental units on day 0. The communities were dominated by

Microcystis (derived from both Mai Gassa I and Era Quihila at a ratio of approximately 1:2) and *Peridinium* (mainly derived from Mai Gassa I); some diatoms, chlorophytes, cryptomonads and *Anabaena* were also encountered, mainly in Era Quihila water.



Fig. 6.1 Biomass of major phytoplankton taxa in the experimental units on the starting day (day 0) of the experiment. Error bars represent one standard error. MG I, Mai Gassa I; EQ, Era Quihila; Z+, Zooplankton treatment; Z-, non-zooplankton treatment.

Overall, the algal biomass tended to increase during the course of the experiment in all treatments (Fig. 6.2a). In contrast, we observed a major decline in zooplankton biomass in the zooplankton treatment (Fig. 6.3). The biomass of *Diaphanosoma* and *Daphnia* rapidly declined and became similar across all treatments from day 4 onwards. The biomass of copepods, although declining, remained higher in the zooplankton than in the non-zooplankton treatments.

Most phytoplankton groups tended to show lower biomasses in the presence

than in the absence of zooplankton in our experiment, but the extent by which phytoplankton growth was reduced differed among taxa (Fig. 6.2). Of the cyanobacteria, *Anabaena* was strongly suppressed in the presence of zooplankton, whereas the impact of zooplankton on *Microcystis* was much less pronounced. *Microcystis* reached higher biomass in Mai Gassa I water both in the absence and presence of zooplankton compared to the respective treatments with Era Quihila water (Fig. 6.2). Chlorophytes seem to be strongly suppressed by zooplankton except in the Mai Gassa I treatment at the end of the experiment. *Peridinium* biomass tends to be reduced in the presence of zooplankton, although only to a small extent.



Fig.6.2 Changes in total biomass (pg C/ml; panel a) and biomass of the major phytoplankton taxa (panels b-h) in the different treatments during the course of the experiment.



Fig. 6.3 Changes in biomass (mg C/L) of the major zooplankton taxa during the course of the enclosure experiment. (a - d, biomass of each genera over the course of time; e - g, % contribution of each genera at each day and treatment).

Partial RDA analysis was performed to investigate the pattern of phytoplankton taxon composition in function of source water and the presence/absence of zooplankton, correcting for the impact of time (Fig. 6.4a). It appears that the differences between the presence and absence of zooplankton (along first axis, with Eigen value of 28.5%) (Trace = 0.363, F = 12.261, p = 0.002) were far more important than the variation caused by source water (Mai Gassa I or Era Quhila water; second axis, with Eigen value of 6.6%). To zoom in on the response of phytoplankton to the treatments, we ran the RDA for the different sources of water separately (Fig. 6.4b & c). We observed similar pattern of impact of the zooplankton on Anabaena, chlorophytes, cryptomonads and to a lesser extent on diatoms, euglenoids, Peridinium and Microcystis for Mai Gassa I (Trace = 0.441, F = 22.267, p = 0.0001) and Era Quihila (Trace = 0.252, F = 13.138, p = 0.002). These analyses suggest an overall tendency for biomass of most phytoplankton groups to be higher in the treatment without than in the treatment with zooplankton, but the impact differs among taxa. According to analyses per individual phytoplankton taxon, significant effect of zooplankton and source water was observed for chlorophytes, cryptomonads, euglenoids, Anabaena, and Microcystis (Table 6.1). Overall treatment effects were not significant for Peridinium and diatoms. Repeated measures analysis of variance (ANOVA) confirmed these results, with a significant effect of zooplankton on the biomass of chlorophytes, cryptomonads, and Anabaena and chlorophytes but not on Microcystis and euglenoids (Table 6.2). A significant effect of source water was observed for chlorophytes, cryptomonads, Anabaena and Microcystis, but not for euglenoids. The Spearman rank correlation coefficient calculations also showed significant associations (negative) of Anabaena with several zooplankton taxa (Diaphanosoma sp., R=0.684, p=0.0141; Daphnia carinata, R = -0.861, p = 0.0003;



Ceriodaphnia cornuta, R=0.998, p =0.0018; and the calanoid copepods, R=-0.853, p =0.00042).

Fig. 6.4. Partial RDA biplot showing the response of phytoplankton taxon composition to the source water and zooplankton treatments, correcting for impact of time (time represented as co-variable). (a) Mai Gassa I and Era Quihila waters together, (b) Mai Gassa I water, (c) Era Quihila water. Arrows represent phytoplankton taxa; upwards pointing triangles are centeroids of the treatments, and empty circles represent samples.

Table 6.1 Results of regression analysis testing for the effect of zooplankton and source water on phytoplankton taxa biomass in the enclosure experiment (April 2007)

Taxon	Trace	F	p
Diatoms	0.358	4.452	0.0500
Chlorophytes	0.697	41.698	0.0020
Cryptomonads	0.540	11.787	0.0040
Euglenoids	0.504	7.567	0.0100
Anabaena	0.394	15.894	0.0020
Microcystis	0.214	6.134	0.0020
Peridinium	0.137	2.864	0.0540

Table 6.2 Results of two way ANOVAs testing for the effect of zooplankton, water source and their interaction on log transformed biomasses of different phytoplankton taxa in the enclosure experiment (April 2007). Z = zooplankton treatment, W = water source, Z*W = interaction between both treatments.

Chlorophytes			Anabaena		
Source	F	р	F	р	
Ζ	11.38	0.009	112.43	<.001	
W	74.06	< 0.001	50.48	< 0.001	
Z*W	4.25	0.073	13.74	< 0.005	
Cryptomonads			Microcystis		
Ζ	6.59	0.033	0.01	0.93	
W	49.87	< 0.001	10.75	0.01	
Z*W	0.77	0.41	0.74	0.41	
Euglenoids					
Ζ	1.26	0.29	-		
W	3.18	0.11			
Z*W	0.54	0.48			

To investigate changes in the dominance of phytoplankton taxon composition during the course of the experiment, we performed partial RDA controlling for differences among treatments (Fig. 6.5). We detected a significant overall temporal variation in the phytoplankton taxa (Trace = 0.124; F = 2.998; p < 0.002). Centeroids of days displayed a circular trajectory on the RDA-plot (Fig. 6.5), indicating that the

dominant phytoplankton taxa shifted with time. *Anabaena*, cryptomonads, euglenoids, diatoms and chlorophytes tended to dominate early in the experiment, while *Microcystis* and *Peridinum* dominated later in the experiment.



Fig. 6.5 Partial RDA biplot on the response of phytoplankton taxon composition in function of time, correcting for among treatment differences (treatments specified as co-variables). Arrows represent phytoplankton taxa; upwards pointing triangles are centeroids (days), and empty circles are samples (enclosures).

Microcystin was detected in all the enclosures, and concentrations ranged from 0.082 to 1.268 µg L⁻¹. We observed significantly higher microcystin production in the treatments with zooplankton (range: 0.22 - 1.27 µg L⁻¹) than the treatments without zooplankton (range: 0.08 – 0.13 µg L⁻¹) (F = 7.76, p = 0.014) (Fig. 6.6). The microcystin concentration per unit biomass of *Microcystis* (micricystin, µg/mgC) was also significantly higher in the zooplankton treatments (f = 7.33, p = 0.027). But no significant effect of water source on microcystin production was observed (F = 0.876, p = 0.377).



Fig. 6.6 Plots representing associations between microcystin production and zooplankton treatments in the enclosure experiments. a) microcystin concentration $(\mu g/L)$ and b) microcystin concentration per unit biomass of *Microcystis* ($\mu g/mgC$).

Discussion

The main observations from our experimental study on the effect of zooplankton on phytoplankton biomass and taxon composition in a highland reservoir in semi-arid Northern Ethiopia are that (1) zooplankton exerts a top-down control on phytoplankton; (2) but this top-down effect is taxon-specific; (3) the toxic cyanobacterium *Microcystis* is not significantly affected by zooplankton in our experiment; (4) rather, our results indicate that survival of zooplankton in our experiment was quite low, which might be related to the presence of toxic cyanobacteria; and (5) the concentration of the toxin microcystin was significantly higher in the presence of zooplankton. In the following paragraphs, we discuss these observations.

Our results indicate that the presence of zooplankton has a more important impact on phytoplankton composition than water source. Multivariate analysis shows that the presence/absence of zooplankton results in a stronger differentiation of phytoplankton community structure than whether medium of Mai Gassa I or Era Quihila was used (Fig. 4a). This plot also suggests that the presence of zooplankton tends to make the taxon composition more similar across waters. In both media, the impact of zooplankton was similar, as is shown in Fig. 4 a & b. Overall, zooplankton tends to suppress the abundance of phytoplankton in our experimental units, but the impact differs strongly depending on the taxon considered. The top-down impact of zooplankton was strong for Anabaena, chlorophytes, and cryptomonads, but was not significant for Microcystis and Peridinum. It is well known that phytoplankton taxa are differentially grazed upon by zooplankton (Burns, 1987, Hartmann, 1985, Bern, 1994), and our results are generally in line with expectations, except for Anabaena. No changes in abundance of large cyanobacteria (Anabaena affinis, A. flos-aquae) were reported in another short-term enclosure experiments (Porter, 1973, Lehman & Sandgren 1985), while the response of Anabaena circinalis in yet another experiment was equivocal (Bergquist & Carpenter, 1986). Geller (1984) reported a high grazing rate of the zooplankton community on Anabaena flos-aquae. This difference may probably be a result of difference in both the cyanobacteria and the grazer zooplankton. Fore example, calanoid copepods are reported to be adapted to graze on large cyanobacteria (Haney, 1987). Haney and Traut (1985) also reported the preference of adult calanoid copepods to feed on large food dominated by filamentous cyanobacteria. Large dinoflagellates like Peridinum species are known not to be grazed efficiently by filter feeding zooplankton (Sandgren, 1992). Similarly, Microcystis is known to be well-protected against zooplankton grazing, by both the formation of colonies as well as by producing toxins (Lampert, 1981, Nizan et al., 1986, Fulton & Paerl, 1987b, Kirk & Gilbert, 1992, DeMott, 1999).

We did not observe top-down regulation of *Microcystis* by zooplankton in our experiment, even though the inoculated densities of zooplankton were quite high. We observed a massive decline of most zooplankton species during the first days of the

200

experiment, including a complete collapse of the Daphnia. The only exceptions were the calanoid copepods and the small-bodied cladoceran Ceriodaphnia. This likely reflects *Microcystis* impacting zooplankton rather than the reverse. We observed a higher concentration of the toxin microcystin per unit biomass of Microcystis in the zooplankton treatments, which suggests that increased toxin production by the Microcystis may have resulted in the collapse of zooplankton. The increased levels of toxin may either reflect an induced change or selection favouring more toxic strains. Jang et al. (2003) reported that several strains of Microcystis aeruginosa increased toxin production in response to direct and indirect exposure to herbivorous zooplankton. Toxin production by different species of cyanobacteria has indeed been considered a defence mechanism against zooplankton grazing (Lampert, 1981, Lampert, 1987, Ghadouani et al. al, 2003, Jang et al., 2003). Laboratory experiments showed that Daphnia died considerably faster than starving controls when offered pure cultures of toxic strains of Microcystis (Lampert, 1981, Nizan et al. 1986). Reduction of population growth rate and reproduction abilities of cladoceran zooplankton by Microcystis strains has been reported from many freshwater systems in tropical and temperate regions (DeMott, 1999, Ferrao-Filho et al., 2000, 2003). Many studies have, however, also reported that cladoceran zooplankton can develop tolerance to cyanobacterial toxins upon exposure (Fulton and Paerl, 1987a, Gustafsson and Hansson, 2004). Daphnia magna populations that were repeatedly exposed to toxic cyanobacteria in their natural habitat were indeed less affected by the toxin than populations lacking exposure (Gustafsson & Hansson, 2004). In addition, it has been shown that Daphnia show genetic differences in tolerance to microcystin and Hairston et al. (1999) have reported evolutionary adaptation of a local Daphnia population to increased tolerance of Microcystis. The Daphnia we used in our experiment were obtained from a reservoir where the genes for microcystin synthesis were not detected (Tsehaye Asmelash, unpubl. data). The animals we used in our experiment were thus not acclimatized to the presence of microcystin, nor were they derived from a population that had been exposed to selection by microcystins. Although not tested and thus speculative, it is possible that our observations would have been different if we would have used *Daphnia* that had been derived from a habitat that is characterized by the regular occurrence of toxic cyanobacteria blooms. In an earlier enclosure experiment, we observed that resident *Daphnia* in two other reservoirs (Tsinkanet and Gereb Awso) were able to reduce dominance of *Microcystis* (Dejenie et al., 2009). Dejenie et al. (2008) also reported a negative association between *Microcystis* and *Daphnia* from a survey of reservoirs in Tigray, North Ethiopia.

As mentioned, there was a major decline in zooplankton biomass during the first days of our experiment. This collapse was pronounced for all cladoceran species except for *Ceriodaphnia*, a species that, however, occurred in low frequencies. In contrast to the cladoceran zooplankton, we recorded higher copepod biomass in the treatments with than without zooplankton. Copepods are selective feeders and have been reported not to be directly affected by toxins of cyanobacteria as long as an alternative food source is available (Kozlowsky-Suzuki et al, 2003). Fulton and Paerl (1988) reported from a laboratory competition experiment that blooms of *M. aeruginosa* can alter zooplankton competitive relations, favouring small-bodied cladocerans and copepods at the expense of large-bodied cladocerans. Small-sized cladocerans generally are characterized by stronger tolerance against *Microcystis* than large-sized species, which may in part be related to the fact that they are too small for

the filaments to interfere with their feeding (Jarvis, 1986, Fulton and Paerl, 1987a, Guo and Xie, 2006).

Overall, the decline of the zooplankton community is remarkable, but our experiment does not allow us to give a definitive conclusion on the causes for the decline because we did not have controls with out *Microcystis*. It could be the *Microcystis*, but could also be something else as well. We suggest repeating similar experiments in the future where zooplankton presence/absence is crossed with *Microcystis* presence/absence.

Reference

- Axler, R.P., G.W. Redfiekl, and C.R. Goldman, 1981. The importance of regenerated nitrogen to phytoplankton productivity in a subalpine lake. *Ecology*, **62**: 345-354
- Bern, L., 1994. Particle selection over a broad size range by crustacean zooplankton. *Freshwater Biology.* 32: 105-112
- Bergquist, A.M., S.R. Carpenter, and J.L. Latino, 1985. Shifts in phytoplankton size structure and community composition during grazing by contrasting zooplankton assemblages. *Limnol. Oceanogr.*, **30**: 1037 -1045
- Bergquist, A.M., and S.R. Carpenter, 1986. Limnetic herbivory: effects on phytoplankton populations and primary production. *Ecology* 67: 1351-1360
- Bogdan, K.G. & D. C. McNaught, 1975. Selective feeding by *Diaptomus* and *Daphnia. Verh. Int. Verein. Limnol.*, **19**: 2935-2942
- Botes, D.P., H. Kruger, and C.C. Viljoen, 1982. Isolation and characterization of four toxins from the blue-green alga *Microcystis aeruginosa*. *Toxicon*, **20**:945-954
- Bottrell, H.H., A. Duncan, Z. Gilwicz, E. Grgyierek, A. Herzig, A. Hillbricht-Ilkowska, H. Kurasawa, P. Larsson, and T. Weglenska, 1976. A review of some problems in zooplankton production studies. *Norwegian Journal of Zoology*, 24: 419-456
- Burns, C.W., 1987. Insights into zooplankton-cyanobacteria interactions derived from enclosure studies. *New Zealand Journal of Marine and Freshwater Research*, 21: 477-482
- Carmichael, W.W., 1994. The toxins of cyanobacteria. Scientific America. 270: 78-86
- Carmichael, W.W., S.M.F.O., Azevedo, J.S. An, R.J.R. Molica, E.M. Jochimsen, S. Lau, K.L. Rinehart, G.R. Shaw, and G.K. Eaglesham, 2001. Human fatalities from Cyanobacteria: Chemical and biological evidence for cyanotoxins. *Environmental Health Perspectives*, **109**: 663-668
- Carpenter, S.R. & J.F. Kitchell, 1984. Plankton community structure and limnetic primary production. *Am. Nat.*, **124**: 159-172
- Chan, F., M.L. Pace, R.W. Howarth, and R.M. Marino, 2004. Bloom formation in heterocystic nitrogen-fixing cyanobacteria: the dependence on colony size and zooplankton grazing. *Limnology and Oceanography*, **49**: 2171-2178
- Chorus, I. and J. Bartram (eds.), 1999. Toxic cyanobacteria in water: a guide to their public health consequences, monitoring and management. St Edmundsbury Press, London.
- Dejenie, T., Tsehaye A., L. De Meester, Afework M., Abreha G., S. Risch, A. Pals, K.
 Van der Gucht, W. Vyverman, J. Nysen, S. Deckers, and J. Decklerck, 2008,
 Limnological and ecological characteristics of tropical highland reservoirs in
 Tigray, Northern Ethiopia. *Hydrobiologia*, 610:193–209

- Dejenie, T., Tsehaye, A., S. Rousseaux, Tteklit, G.-G., Aabreha, G.-K., Mekonnen T., J. Nyssen, J. Deckers, K. Van Der Gucht, WW. Vyverman, L. De Meester, and S.A.J. Declerck, 2009. Impact of the fish Garra on the ecology of reservoirs and the occurrence of Microcystis blooms in semi-arid tropical highlands: an experimental assessment using enclosures. *Freshwater Biology* (in press).
- DeMott, W.R. & F. Moxter, 1991. Foraging on cyanobacteria by copepods: responses to chemical defences and resource abundance. *Ecology*, **72**: 1820-1834
- DeMott, W.R., and S. Dhawale, 1995. Inhibition of in vitro protein phosphatase activity in three zooplankton species by microcystin-LR, a toxin from cyanobacteria. *Arch. Hydrobiol.*, **134**:417-424
- DeMott W.R., 1989. The role of competition in zooplankton succession. In: SommerU. (Ed.). Plankton ecology: succession in plankton communities. Springer-Verlag.
- DeMott, W.R., 1999. Foraging strategies and growth inhibition in five daphnids feeding on mixture of a toxic cyanobacterium and a green algae. *Freshwater Biology*, **42:** 263-274
- Elder, G.H., P.R. Hunter, and G.A. Codd, 1993. Hazardous freshwater cyanobacteria (blue green algae). *Lancet*, **341**:1519-1520
- Elser, J.J., N.C. Goff, N.A. MacKay, A.L. Amand, M.M. Elser and S.R. Carpenter, 1987. Species-specific algal responses to zooplankton: experimental and field observations in three nutrient-limited lakes. *Journal of Plankton Research*, 9: 699-717.
- Elser, J.J., 1999. The pathway of noxious cyanobacteria blooms in lakes: the food web as a final destination. *Freshwater Biology*, **42**: 537-543

- Eriksson, J. E., D. Toivola, J.A. Meriluoto, H. Karaki, Y.G. Han, and D. Hartshorne,
 1990. Hepatocyte deformation induced by cyanobacterial toxins reflects
 inhibition of protein phosphatases. *Biochem. Biophys. Res. Commun.*, 173:13471353
- Ferrao-Filho, A.S., S.H.F.O., Azevedo, and W.R. DeMott, 200. Effects of toxic and non-toxic cyanobacteria on the life history of tropical and temperate cladocerans. *Freshwater Biology*, **45**: 1-19
- Ferrao-Filho, A.S., and S.H.F.O., Azevedo, 2003. Effects of unicellular and colonial forms of toxic *Microcystis aeruginoa* from laboratory cultures and natural populations on tropical cladocerans. *Aquatic Ecology*, **37**: 23-35
- Fulton, R.S., and H.W. Paerl, 1987a. Toxic and inhibitory effects of the blue green algae *Microcystis aeruginosa* on herbivorous zooplankton. *Journal of Plankton Research*, 9: 837-855
- Fulton, R.S., and H.W. Paerl, 1987b. Effects of colonial morphology on zooplankton utilization of algal resources during blue-green algal (*Microcystis aeruginosa*) blooms. *Limnology and Oceanography*, 32: 634-644
- Fulton, R.S., and H.W. Paerl, 1988. Effects of the blue-green alga *Microcystis aeruginosa* on zooplankton competitive relations. *Oecologia*, **76**: 383-389
- Ghadouani, A., B. Pinel-Alloul, and E. Prepas, 2003. Effects of experimentally induced cyanobacterial blooms on crustacean zooplankton communities. *Freshwater Biology*, 48: 363–381
- Geller, W. 1984. A device for the *in situ* measurement of zooplankton food selection, grazing and respiration rates. *Verhandlungen der internationalen Vereinigung fur theoretische und angewandte Limnologie*, **22**:1425-2431

- Gilbert, J.J., 1990. Differential effects of *Anabaena affinis* on cladocerans and rotifers: mechanisms and implications. *Ecology*, **71**: 1727-1740
- Guo, N., and P. Xie, 2006. Development of tolerance against toxic *Microcystis aeruginosa* in three cladocerans and the ecological implications. *Environmental Pollution*, **143**: 513-518
- Gustafsson, S. and L.-A. Hansson, 2004. Development of tolerance against toxic cyanobacteria in *Daphnia*. *Aquatic Ecology*, **38**: 37-44
- Hairston, N.G. Jr, W. Lampert, C.E. Cáceres, C.L. Holtmeier, L.J. Weider, U. Gaedke, J. M. Fischer, J.A. Fox, and D. M. Post, 1999. Rapid evolution revealed by dormant eggs. *Nature*, 401: 446
- Haney, J.F, and D.J. Hall, 1973. Sugar-coated *Daphnia*: a preservation technique for Cladocera. *Limnology and Oceanography*, 18: 331-333
- Haney, J.F., and M.A. Traut, 1985. Size selective grazing by zooplankton in Lake
 Titicaca. Archiv fiir Hydrobiologie, Beiheft Ergebnisse der Limnologie, 21: 141160
- Haney, J.F., 1987. Field studies on zooplankton-cyanobacteria interactions. *Neaw Zealand Journal of Marine and Freshwater Research*. **21:** 467-475
- Hartmann, H.J., 1985. Feeding of *Daphnia pulicaria* and *Diaptomus ashlandi* on mixtures of unicellular and filamentous algae. *Verhandlungen Internationale Vereinigung für Theoretische und Angewandte Limnologie*, 22: 3178-3183
- Havens, K. & J. DeCosta, 1985. An analysis of selective herbivory in an acid lake and its importance in controlling phytoplankton community structure. *J. Plankton Res.*, 7: 207–222
- Huisman, J., H.C.P Matthijs, and P.M. Visser (Eds), 2005. Harmful cyanobacteria. Springer, Berlin.

- Humpage, A.R, and I.R. Falconer, 1999. Microcystin-LR and liver tumour promotion: effects on cytokinesis, ploidy, and apoptosis in cultured hepatocytes. *Env. Toxicol.*, **14**:61-75
- Jang, M.-H., K. Ha, G.-J. Joo, and N. Takamura, 2003. Toxin production of cyanobacteria is increased by exposure to zooplankton. *Freshwater Biology*. 48: 1540-1550
- Jarvis, A.C., 1986. Zooplankton community grazing in a hypertrophic lake (Hartbeespoort Dam, South Africa). *Journal of Plankton Research*, **8**: 1065-1078
- Jayatissa, L.P., E.I.L. Silva, J. McElhiney, and L.A. Lawton, 2005. Occurrence of toxigenic cyanobacterial blooms in freshwaters of Sri Lanka. Systematic and Applied Microbiology, 29: 156-164
- Jeppesen, E., J.P. Jensen, M. Søndergaard, T. Lauridsen, and F. Landkildehus, 2000. Trophic structure, species richness and biodiversity in Danish lakes: changes along a phosphorus gradient. *Freshwater Biology*, **45**: 201-218
- Joshimsen, E.M., W.W. Carmichael, J. An, D.M. Cardo, S.T. Cookson, C.E.M. Holmes, M.B. de C. Antunes, D.A. de M. Filho, T.M. Lyra, V.S.T. Barreto, S.M.F.O. Azevedo, and W.R. Jarvis, 1998. Liver failure in death after exposure to microcystins at a haemodialysis centre in Brazil. *N. Eng. J. Med.*, 338: 873-878
- Kirk, L.K., and J.J. Gilbert, 1992. Variation in herbivore response to chemical defences: zooplankton foraging on toxic cyanobacteria. *Ecology*, **73**: 2208–2217
- Kozlowsky-Suzuki B., M. Karjalainen, M. Lehtiniemi, J. Engström-Öst, M. Koski, and P. Carlsson, 2003. Feeding, reproduction and toxin accumulation by the

copepods *Acartia bifilosa* and *Eurytemora affinis* in the presence of the toxic cyanobacterium *Nodularia spumigena*. *Mar. Ecol. Prog. Ser.*, **249**: 237–249

- Kreintz, L., A. Ballot, K. Kotut, C. Wiegand, S. Pütz, J.S. Metcalf, G.A. Codd, and S. Pflugmacher, 2003. Contribution of hot spring cyanobacteria to the mysterious deaths of Lesser Flamingos at Lake Bogoria, Kenya. *FEMS Microbiol Ecol.*, 43, 141-148
- Lampert, W. 1981. Inhibitory and toxic effects of blue-green algae on *Daphnia*. *Internationale Revue der gesamten Hydrobiologie*, **66**: 285-298
- Lampert, W., 1982. Further studies on the inhibitory effect of the toxic blue-green Microcystis aeruginosa on the filtering rate of zooplankton. Archiv für Hydrobiologie, **94**: 207-220
- Lampert, W. & B.E. Taylor, 1985. Zooplankton grazing in a eutrophic lake: implications of diel vertical migration. *Ecology:* **66**, 68 82.
- Lampert, W., 1987. Laboratory studies on zooplankton-cyanobacteria interactions. New Zealand Journal of Marine and Freshwater Research, **21**: 483-490
- Lehman, J.T., 1980. Release and cycling of nutrients between planktooic algae and herbivores. *Limnol.Oceanogr.*, 21: 501-516
- Lehman, J.T., and C.D. Sandgren, 1985. Species-specific rates of growth and grazing loss among freshwater algae. *Limnology and oceanography* **30**: 34-46
- Lürling, M., 2003. *Daphnia* growth on microcystin-producing and microcystin-free *Microcystis aeruginosa* in different mixtures with the green alga *Scenedemsus obliquus*. *Limnology and Oceanography*, **48**: 2214-2220
- Mackintosh C, Beattie KA, Klumpp S, Cohen P, and Codd GA., 1990. Cyanobacterial microcystin-LR is a potent and specific inhibitor of protein phosphatases 1 and 2A from both mammals and higher plants. *FEBS Lett.*, **264**:189–192

- Menden-Deuer, S., and Lessard E.J., 2000. Carbon to volume relationships for dinoflagellates, diatoms, and other protist plankton. *Limnology and Oceanography*, 45: 569-579
- Peters, R.H., 1975. Phosphorus regeneration by natural populations of limnetic zooplankton. *Verh. Int. Verein.Limnol.*, 19: 273-279
- Resson, R., F.S. Soong, J. Fitzgerald, L. Turczynowicz, O. El Saadi, D. Roder, and I.Falconer, 1994. Health Effects of Toxic Cyanobacteria (Blue Green Algae).Australian Government Publishing Service, Canberra, Australia.
- Nizan, S., C. Dimentman, M. Shilo, 1986. Acute toxic effects of the cyanobacterium Microcystis aeruginosa on Daphnia magna. Limnology and oceanography **31**: 497-502
- Ohtani, L., R.E. Moore, and M.T.C. Runnegar, 1992. Cylindrospermopsin: a potent hepatotoxin from the blue green alga *Cylindrospermopsis raciborskii. J. Am. Chem. Soc.*, **114**:7941-7942
- Penaloza, R., M. Rojas, I. Vila, and F. Zambrano, 1990. Toxicity of a soluble peptide from Microcystis sp. to zooplankton and fish. *Freshwater Biology*. **24**: 233-240
- Porter, K.G. 1973. Selective grazing and differential digestion of algae by zooplankton. *Nature 244*: 179-180
- Porter, K.G., and J.D. Orcutt, 1980. Nutritional adequacy, manageability, and toxicity as factors that determine the food quality of green and blue-green algae for *Daphnia*. In Kerfoot W. C. (ed.), Evolution and ecology of zooplankton communities. University Press of New England.
- Repavich, W., W.C. Sonzogni, J.H. Standridge, R.E. Wedepohl, and L.E. Meisner, 1990. Cyanobacteria (blue green algae) in Wisconsin waters: acute and chronic toxicity. *Water Res.*, 24:225-231

- Rinehart, K.L., M. Namikoshi, and B.W. Choi, 1994. Stucture and biosynthesis of toxins from blue-green algae (cyanobacteria). J. Appl. Phycol. 6:159-176
- Sandgren, C.D., 1992 (Ed.). Growth and reproduction strategies of freshwater phytoplankton. Cambridge University Press, 452pp.
- Shapiro, J., 1984. Blue-green algal dominance in lakes: the role and management significance of pH and CO₂. *Int. Revue ges. Hydrobiol*, **69**: 765-80

Sivonen, K, 1996. Cyanobacterial toxins and toxin production. *Phycologia*, 35:12-24

- Sommer, U., 1989. Plankton Ecology. Springer-Verlag.
- Sterner, R.W, 1986. Herbivores' direct and indirect effects on algal populations. *Science*, 231: 230-232
- van der Oost, R., 2007. Cyanotoxine monitoring: standaardisering en validatie van methoden voor de Nederlandse waterkwaliteitsbeheerders. – Rapport STOWA onderzoek 2005 and 2006. Waternet/Waterproef. Amsterdam.
- Van Ginkel, C.E., 2003. A National Survey of the incidence of cyanobacterial blooms and toxin production in major impoundments. Internal Report No. N/0000/00/DEQ/0503. Resource Quality Services, Department of Water Affairs and Forestry. Pretoria.
- Watanabe, M.F., K. Tsuji, Y. Watanabe, K. Harada and M. Suzuki, 1992. Release of heptapeptide toxin during decomposition process of *Microcystis aeruginosa*. *Natural toxins*, 1: 48-53
- Webster, K.E., and R.H. Peters, 1978. Some size-dependent inhibitions of larger cladoceran filterers in filamentous suspensions. *Limnology and Oceanography* 23: 1238-1245.
- Zurawell, R.W., H.R. Chen, J.M. Burke, and E.E. Prepas, 2005. Hepatotoxic cyanobacteria: A review of the biological importance of microcystins in

freshwater environments. Journal of Toxicology and Environmental Health-Part

B-Critical Reviews, **8**: 1-37.

Chapter 7

General discussion

The Ethiopian highlands support about 88% of the population of the country (MNRDEP, 1994; Mohamed-Saleem, 1995), more than 70% of the livestock population, and 90% of the economic activities of the country (Constable, 1984; FAO, 1986). The Northern parts of the highlands are semi-arid and are among the most ancient settled areas in the world (Huffnagel, 1961; McCann, 1995). The region suffered from a long history of land degradation and loss of vegetation (El-Swaify and Hurni, 1996) which resulted in episodes of recurrent droughts (Haile, 1988). Currently, there is a trend to collect rain water in reservoirs for irrigation based agriculture, electric power generation, and drinking water production. To this end, numerous small reservoirs have been constructed in the Tigray Regional State, North Ethiopia, in the last two decades (Hurni, 1993; De Wit, 2003; Nigussie et al., 2006; Tsehaye et al. 2007). The reservoirs suffer from excessive nutrient loading associated with sediment influx (Nigussie et al. 2006) which resulted from massive erosion linked to land degradation (Nyssen et al., 2005). Many of the reservoirs, therefore, suffer from phytoplankton blooms, including cyanobacterial blooms (Dejenie et al. 2008).

This study was initiated with the aim of getting a better understanding of the conditions that favor the formation of cyanobacterial blooms in the reservoirs in Tigray, North Ethiopia. We studied spatial and temporal patterns in phytoplankton community composition in a set of reservoirs in the region, with an emphasis on potentially toxic cyanobacteria and the conditions and processes that may lead to bloom formation. In addition, the bacterial community composition and its

relationship with abiotic and biotic components of the ecosystem were investigated. Finally, we also tested the impact of fish and zooplankton on phytoplankton community composition in enclosure experiments.

In the following, we discuss our major results with respect to the different objectives outlined in the introduction. Finally, we formulate some recommendations for the management of the reservoirs.

7.1. Phytoplankton community composition of the reservoirs

While the limnological characteristics of the reservoirs studied in this thesis have been discussed by Dejenie et al. (2008), we here describe the phytoplankton species composition, with an emphasis on the occurrence of bloom-forming cyanobacteria in 32 small reservoirs which were sampled twice, once in the wet season and once in the dry season. Overall, phytoplankton richness was low, with most reservoirs being dominated by a single genus of cyanobacteria (mostly *Microcystis*), chlorophytes, euglenophytes, cryptophytes or dinophytes. We report low phytoplankton richness compared to reports from sub-tropical and tropical lakes and reservoirs (Elizabeth and Amha, 1994, Ariyadej et al., 2004; Nabout et al., 2007). This probably results from the eutrophic nature of the system (Huszar et al., 1998, Leibold, 1999, Dudgeon et al., 2006, Lo et al., 2008). Year round dominance of cyanobacteria (mainly *Microcystis*) is common in eutrophic lakes (Sommer et al., 1986, Tilzer, 1987, Zohary & Robarts, 1989; Reynolds 1998, Sivonen & Jones, 1999; Akin-Oriola, 2003).

The phytoplankton community using the biomass data showed associations only with local environmental variables; despite the importance of regional variables in explaining the spatial configuration of communities (e. eg. Soininen et al., 2007,

Vanormelingen et al., 2009). We observed dynamic phytoplankton community and environmental variables in the reservoirs in Tigray. The phytoplankton community compositions of the two seasons were not correlated to each other. This may result from the variations between the reservoirs and the high turnover of the phytoplankton communities. Different reservoirs respond differently to the environmental changes they face during the dry season (Dejenie et al., 2008). We documented a tendency of shift of dominance from cyanobacteria (mainly Microcystis) to dinoflagellates (mainly Peridinium) in the dry season. Environmental factors explaining the phytoplankton community composition differ depending on season. Zooplankton community structure, and pH showed significant association with phytoplankton community composition in the wet season, whereas the variations in the phytoplankton community composition was associated with reservoir depth and conductivity in the dry season. We reported significant temporal variation in phytoplankton community assemblages (chapter 3). We were able to draw some pattern of dominance of different phytoplankton taxa corresponding to the prevailing environmental variables. From September to February, the chlorophytes, cryptomonads, and diatoms co-dominate with cyanobacteria. Cyanobacteria (dominated by Microcystis) and Peridinum (Dinoflagellates) dominate from March to August. But, no apparent seasonal variation was reported for cyanobacteria, dinophytes and euglenoids. We observed higher inter-annual variation (variation among years) in the abiotic variables compared with the intra-annual variation (variation among different months). We also observed temporal turn-over in the phytoplankton and zooplankton community composition. The variability in the abiotic and biotic components of the reservoirs revealed the dynamic nature of the systems and absence of repeatability. The intra-annual variation was mainly associated with

variation in rainfall and temperature. In tropical systems, marked variations in temperature and rainfall between seasons often influence the physico-chemical characteristics of water bodies (Adebisis, 1981, Chapman & Kramer, 1991). Seasonal changes in phytoplankton community structure associated with droughts and water level declines and periods of inflow have been reported from other sub-tropical reservoirs (Horris et al., 1996).

7.2. The bacterial communities of the reservoirs

In chapter 4 we reported bacterial community composition using DGGE band intensity as a measure of relative abundance of operational taxonomic units (OTUs). We estimated bacterial richness by treating each band as an individual OTU and using the number of bands as an indicator of richness. We reported low bacterial richness (range between 2 and 15) compared to reports from other lakes (Konopka et al., 1999, Van der Gucht et al. 2001, 2007). In contrast to other reports (Eiler & Bertilsson, 2004), we also reported a lower bacterial taxon richness in reservoirs accompanying *Microcystis* blooms than bacterial communities in reservoirs without *Microcystis* blooms. Our data can not explain this difference in taxon richness, but we can speculate it is caused by the toxicity of *Microcystis*.

Similar to the response of the phytoplankton communities in the seasonal survey (chapter 2), different environmental factors explained the bacterial community composition in different seasons. Percentage contribution of *Microcystis* to the total phytoplankton biomass and copepod biomass showed significant association with the bacterial community composition in the wet season whereas variation in bacterial community composition was associated with total nitrogen (TN), total phosphorus (TP), oxygen, the number of cattle frequenting the reservoir, and fish biomass in the

dry season. Both top-down and bottom-up factors are reported to regulate bacterial communities (Pace & Cole, 1996; Langenheder & Jürgens, 2001, Muylaert et al., 2002). In most situations bacteria are limited primarily by resources (Pace & Cole, 1996). Predation by zooplankton has also been reported to be an important structuring factor for planktonic bacterial communities (Pace & Cole, 1996, Langenheder & Jürgens, 2001). Seasonality of bacterial community structure in eutrophic lakes has been reported to depend on substrate sources as well as on the food web structure (Muylaert et al., 2002). *Microcystis* may also impact the bacterial community with their secondary metabolites which are reported to have antimicrobial and toxic effects (Carmichael, 1992, Patterson *et al.* 1994, Casamatta & Wickstrom, 2000, Valdor & Aboal, 2007).

7.3. Structuring factors

In addition to the field work, we carried out two different experiments with an aim to 1) directly assess the impact of fish on the abiotic conditions and the different trophic levels in the reservoirs and 2) asses the potential top-down effects of zooplankton on the phytoplankton communities, including toxic cyanobacteria.

In the following, we try to present our observations of the surveys and the two experiments in an effort to explain the relationships of phytoplankton communities in the reservoirs with the abiotic and biotic components of the system. We present our explanations in the framework of bottom-up and top-down control and we also try to elaborate the role of cyanobacteria in the trophic structure.

7.3.1. Bottom-up control as a major structuring factor

The regulation of the bacterioplankton by environmental variables has been discussed in section 7.2 of this chapter. Both bottoms-up and top-down factors are important as regulators of the bacterioplankton in our study system. We have also reported the importance of *Microcystis* to structure the bacterioplankton. We report lower richness of the bacterial community in reservoirs with *Microcystis blooms*.

Strong bottom-up regulation of the phytoplankton community was not observed from our survey study based on the phytoplankton biomass estimates. We observed, however, an indirect bottom-up influence of Garra on the phytoplankton community by positively influencing the nutrient (chapter 5) concentrations (total phosphorus and total nitrogen) in the water column. Our results are consistent with patterns that were detected during a survey study of 32 reservoirs (Dejenie et al., 2008). In that survey, a positive association between Garra biomass and total phosphorus was recorded. The association of fish with bacterial community composition (chapter 4) may also reflect increased nutrient availability in the water column either through direct excretion and recycling of benthic detritus or via sediment re-suspension. Positive influences by fish on nitrogen and phosphorus have been reported from other enclosure experiments and whole lake manipulations (Northcote, 1988; Vanni et al., 1997, Vanni & Layne, 1997, Scheffer et al., 2003, Persson & Sevensson, 2006). Overall our results of the enclosure experiment (chapter 5) indicate that the zooplankton community composition has been affected by *Garra* through indirect bottom-up mechanisms. Abundant zooplankton species, such as Diaphanosoma and Ceriodaphnia, were promoted by the presence of Garra. Our observations could not, however, clearly demonstrate whether the effect on the larger bodied zooplankton (mainly Daphnia) is from direct top-down regulation of

zooplankton by fish or an indirect effect of fish by affecting the phytoplankton community composition. Two alternative explanations are possible 1) Garra, with a preference for larger zooplankton, causes the observed shift to the small-sized zooplankton by selective feeding on the large zooplankton or; 2) Garra by influencing the nutrient concentration in the water column indirectly influence the phytoplankton community and toxins form cyanobacteria (mainly Microcystis) regulate zooplankton. Cyanobacteria are known to produce a wide range of toxins which act as defence against zooplankton (Lampert, 1981, Nizan et al., 1986, Fulton & Paerl, 1987, DeMott, 1999), resulting in reduced population growth rate and reproduction of cladoceran zooplankton (Gliwicz & Lampert, 1990, DeMott, 1999, Ferrao-Filho et al., 2000, 2003, Deng, 2008). Furthermore, we observed a shift in zooplankton community composition towards smaller bodied cladocerans and copepods in our second enclosure experiment (chapter 6), with no apparent effect of zooplankton on Microcystis. Jang et al. (2003) reported an increase in toxin production by *Microcystis* in response to direct and indirect exposure to herbivorous zooplankton of several species. We could not, however, reach a firm conclusion on the mechanisms of Garra – zooplankton (mainly Daphnia) – Microcystis interactions. Our data could not clearly demonstrate whether the effect on Daphnia is from topdown regulation by fish or an indirect effect of fish by affecting the phytoplankton community composition. We could report a tendency for Daphnia to regulate Microcystis from one of our experiment (chapter 5), but Daphnia rapidly declined in the second experiment (chapter 5) which may result from toxic effect of Microcystis.

The main purpose of the reservoirs in Tigray is to supply water for irrigation, but they are also used as sources of water for domestic animals and different household purposed. Our report on the presence of toxic strains of *Microcystis*

(detection of the genes responsible for microcystin production and the toxins), therefore, is of public health concern. Cyanobacterial toxins are known to affect animals including humans (Lampert, 1981, Nizan *et al.*, 1986, Fulton & Paerl, 1987, Resson et al., 1994, Chorus & Bartram, 1999, DeMott, 1999, Carmichael et al., 2001). Incidents of animal deaths and human illnesses attributed to toxic cyanobacteria are reported (Kuiper-Goodman et al., 1999). The World Health Organization (WHO) has set a provisional guidline value for microcystin-LR in drinking water (0.001 mg L⁻¹ for total microcystin-LR, free plus cell-bound) (Chorus & Bartram, 1999, WHO, 2006).

7.3.2. Top-down control as a major structuring factor

In our seasonal survey (chapter 2) we reported association of phytoplankton community with zooplankton (top-down regulators). The chances for top-down regulation of phytoplankton are lower in tropical lakes (Fernando, 1994; Lazzaro, 1997) because *Daphnia* seem to be rare or absent at lower latitudes (Hebert, 1978, Foran, 1986, Dumont, 1994b, Gillooly and Dodson, 2000, Brucet et al., 2005; Gyllstrom et al., 2005). In tropical lakes, fish species reproduce throughout the year, and this can reduce the chances for top-down control of phytoplankton by *Daphnia*. Our study reservoirs mainly harbour the riverine fish, *Garra*. Dejenie et al. (2008) observed a positive association for specific taxa of zooplankton, such as *Daphnia barbata*, small sized cladocera with *Garra*. This, together with our results of the enclosure experiment (chapter 5) indicates that the ability of *Garra* to control zooplankton is limited. This may be due to the fact that *Garra* is a riverine fish with a ventrally oriented mouth that presumably feeds mainly on detritus, benthos or larger invertebrates (Mekonen et al., unpublished).

We were able to see an apparent top-down regulation of phytoplankton by the zooplankton community in the second experiment (chapter 6). We observed top-down regulation by zooplankton on some of the phytoplankton taxa including Anabaena, chlorophytes, Cryptomonads, but the impact on Microcystis and Peridinum was limited. Reports on differential grazing by zooplankton on different sizes of phytoplankton support this observation (Burns, 1987, Hartmann, 1985, Bern, 1994). The difference in response to grazing zooplankton by the two dominant taxa of cyanobacteria in the reservoirs, Microcysytis and Anabaena, is interesting. Anabaena tended to be negatively affected, while Microcystis increased in the presence of zooplankton. Contrasting reports on the ability of zooplankton to regulate Anabaena exist (Geller, 1984, Haney and Traut, 1985, Lehman & Sandgren 1985, Bergquist & Carpenter, 1986). This probably resulted from differences in the sizes of different Anabaena taxa and the grazer zooplankton. Calanoid copepods are adapted to utilize large cyanobacteria (Haney, 1987). We reported a negative correlation of Anabaena with calanoid copepods and D. carinata. Preferential feeding of adult calanoid copepods on large food dominated by filamentous cyanobacteria has been reported by Haney and Traut (1985).

We reported a negative association of *Microcystis* with *Daphnia* and a tendency for a positive relationship with fish in chapter 2. But from the enclosure experiment (chapter 6) we reported lower ability of zooplankton to regulate *Microcystis;* and an over-all effect of fish on *Microcystis* is reported in chapter 5. We also reported a negative effect of *Microcystis* on the bacterial community richness in the reservoirs. All these results show the importance of *Microcystis* in our system and its relationship with the biotic components is central. These results also indicate an inter-play between bottom-up (possibility of *Microcystis* affecting the zooplankton

composition) and top-down (zooplankton grazing) regulation. We have detected toxins from our experimental units in one of the experiments we carried out and higher levels of toxins were produced in the presence of zooplankton than in the absence. These observations support the hypothesis that *Microcystis* regulate zooplankton. On the other hand, negative association of *Microcystis* with *Daphnia* has been suggested mainly from the field survey. Dejenie et al. (2008) also reported negative association of the percentage contribution of *Microcystis* to the total phytoplankton with *Daphnia*. From our field observations and the experiments, we can conclude that the zooplankton grazing can not regulate *Microcystis* blooms in these reservoirs and that zooplankton can increase the microcystis blooms and *Microcystis* to that *Daphnia* may control *Microcystis* at lower biomasses until it reaches a certain biomass and *Microcystis* "escapes" potential control by zooplankton after which it will poison all zooplankton due to the high densities.

7.4. The impact of fish on water quality

In the reservoirs of Tigray, the predominant fish belongs to the genus *Garra*, an opportunistic benthivorous fish. It strongly depends on detritus, but it can also feed on cladocerans in addition to benthic invertebrates (Mekonen et al., in prep.). We reported from our enclosure experiment that *Garra* influences major environmental variables including total nitrogen, total phosphorus, water column transparency and suspended matter. Thus, it favoured increased phytoplankton productivity and increased biomass accumulation mediated by a bottom-up effect, through enhanced nutrient availability in the water column. Previous studies on other taxa of fish have observed that the presence of fish results in an increase in both phytoplankton

biomass and zooplankton densities (Rejas et al., 2005) and positive effects on nutrient (phosphorus and nitrogen) (Persson and Svensson, 2006).

7.5. General suggestions and recommendations

Based on the collected data on the environment gradients and the biota and their relationships in space and time in semi-arid highland reservoirs in Tigray, we can put some general suggestions and recommendations for sustainable utilization while maintaining the ecological integrity of the reservoirs. One major aspect we aim to recommend is the issue of water quality. We have reported the stocking of few of the reservoirs with commercial fish. But fish stocking in the northern part of Ethiopia as part of the farming activity has not been fully explored, although it can contribute to the economic use of the reservoirs and help to alleviate the chronic food insecurity in the region. But before engaging in stocking, some issues like the capacity of the reservoirs to sustain fish production, social aspects of fish consumption in the region and fish marketing, and the impact of fish on water quality need to be understood. Based on the data collected and the major finding we document, hereunder we forward some general suggestions and recommendations for sustainable management of the reservoirs:

1) Setting up of cyanobacteria monitoring and survey for hazardous effects: cyanobacterial blooms are already a prominent characteristic of many if not most of the reservoirs in the highlands of Tigray. Given that most reservoirs harbour toxic strains and people use the reservoirs for watering cattle and other household purposes, it is quite logic that potential hazards are in place. Awareness about the hazards of toxic cyanobacteria is minimal in Ethiopia. We, therefore, recommend the following measures to be implemented:

- a. Awareness creation to responsible authorities on the potential hazards of cyanobacteria and familiarization of local people with the appearance and emergence of cyanobacterial blooms. In contrast to several other waterborne microbial and toxicant health hazards, cyanobacteria are often readily apparent to the human eye and as such easy to monitor.
- b. Survey of animal and human health problems related to cyanobacterial blooms in localities with reservoirs infested with cyanobacteria. Whether the toxic effects of the existing blooms of cyanobacteria (mainly *Microcystis*) have already been translated into human and/or animal health problems in the vicinity of the reservoirs is not known. We believe it is quite important to undertake a survey and document the animal and/or human health.
- c. Installing mechanisms of monitoring blooms and emergence of toxic strains.
 We recommend this to be routinely introduced as *hazard ala*rming system beyond an academic interest. This, we believe, will help to take measures before heavy damage occurs.
- 2) Reduce nutrient loading and sediment input: the reservoirs are eutrophic and our survey indicates that they are impacted with phytoplankton blooms. It is imperative that nutrient loading is reduced. We can only make a very general recommendation as this issue requires a concerted effort of various stake holders, including local communities, the government and non government bodies. Catchment restoration is a global recommendation. Currently the major source of nutrient to the reservoirs is erosion losses and transport of sediment. It is, therefore, important to restore the terrestrial vegetation and reforestation of the valleys in which the reservoirs are situated. The activities of creating exclosures in some areas of the region are encouraging, and such activities along the hills can

allow vegetation re-growth and reduce erosion. In addition, we think it might be important to take measures to allow vegetation to develop in the immediate neighbourhood of the reservoirs, i.e. in a buffer area around the shore.

3) Reduce the impact of cattle: quite a large number of cattle visit freely the reservoirs. Cattle trampling and nutrient input by large amounts of livestock is likely contributing to the deteriorated water quality in the reservoirs. Complete prevention of access to livestock seems impractical. Restricting access to certain part of the reservoirs and to drink animals from canals can reduce the impact of trampling enormously. Restricting cattle to certain area of the reservoir may also serve another function: prevent or minimize the risk of intoxication of animals by toxic blooms of cyanobacteria if selection of site for watering animals is assisted by experts. Establishing cattle free zones is crucial to allow littoral vegetation development and for the establishment of reed land that may strongly reduce nutrient input in the reservoirs. The construction of drinking tubs or small ponds that are fed by water of upstream reservoirs would even be more efficient.

Cattle production in Ethiopia is not efficient, traditionally communities measure wealth by the number of cattle not by the quality of cattle they have. And this leads to higher density of cattle, with as a consequence negative impact on the environment. To recommend overall reduction of the cattle stock seems impractical in the current situation, and it is far beyond the scope of this study. We, however, recommend for policies to guide and pave the way towards a more limited number of cattle.

4) Reduce the impact of fish: our results indicate that the presence of fish, mainly Garra, has a negative impact on water quality by influencing turbidity, nutrient concentration and thereby abundance of cyanobacteria. Our results suggest that

fish contribute to the likelihood of cyanobacteria blooms. We see three potential ways of reducing the impact of *Garra* on the system, each with their own practical limitations, risks and potential benefits:

- a. *Removal of Garra by fishing or other appropriate methods*: this is one way to reduce the impact of *Garra*. Mechanical removal of the small fish may be practically very cumbersome and inefficient, but it would be a safe way to try to reduce cyanobacteria blooms. Advocating activities can be coupled with the removal to use the fish as animal (chicken) feed.
- b. *Introduction of predatory fish*: This might be a very efficient way to reduce densities of *Garra* to reasonable levels, but it is associated with some risks. Introducing predatory fish always entails a risk for damage either locally or in connected systems. Introduction of predatory fish should be considered as the last alternative to avoid the associated risks. It should be noted here that if introduction of predatory fish is considered, we strongly recommend indigenous fish to be used in order to avoid the unpredictable impact of exotic species on local biodiversity.
- c. Fish culture in the reservoirs: we know that quite a number of the reservoirs can support fish production. Introduction of commercial fish (thereby replacing *Garra*) like the well adapted *Tilapia* can bring an added value to the reservoirs. With respect to water quality, it requires an investigation to determine the density of *Tilapia* the reservoirs can support without a compromise to water quality. Some research results show that the presence of filter feeder fish may help in reducing cyanobacteria blooms. Stocking *Tilapia* has been recommended by Yirgaw et al (2000) from southern Ethiopia for management of a turbid lake. *Tilapia* can feed by filtering cyanobacteria. High

survival under low dissolved oxygen and high suspended matter is reported for this fish; thus it is a good candidate for stocking and evaluation of the trophic state of our systems.

Each option forwarded for the reduction of *Garra* has its own pros and cons. Thus, it is of paramount importance that whole lake manipulations with biological means of control need to be well thought and studied. Before widespread application of an option, we recommend thorough investigation and evaluation on selected study reservoirs. Finally we recommend for accumulation of knowledge and expertise on fisheries and fish to consciously implement corrective measures on the potential hazards of cyanobacterial blooms.

References

- Adebisi, A.A., 1981. The physico-chemical hydrology of a tropical seasonal river-Upper Ogun river. *Hydrobiology*, **79**: 157–165
- Akin-Oriola, G.A., 2003. On the phytoplankton of Awab reservoir, Ibadan, Nigeria. *Rev. Biol. Trop.*, **51**: 99-106
- Ariyadej, C., R. Tansakul, P. Tansakul, and S. Angsupanich, 2004. Phytoplankton diversity and its relationships to the physico-chemical environment in the Banglang Reservoir, Yala Province Songklanakarin. *J. Sci. Technol.*, 26: 595-607
- Bergquist, A.M., and S.R. Carpenter, 1986. Limnetic herbivory: effects on phytoplankton populations and primary production. *Ecology* 67: 1351-1360

- Bern, L., 1994. Particle selection over a broad size range by crustacean zooplankton. *Freshwater Biology.* 32: 105-112
- Brucet,S., D. Boix, R. López-Flores, A. Badosa, R. Moreno-Amich and X.D. Quintana, 2005. Zooplankton structure and dynamics in permanent and temporary Mediterranean salt marshes: taxon-based and size-based approaches. *Archiv für Hidrobiologie*, **162**: 535–555
- Burns, C.W., 1987. Insights into zooplankton-cyanobacteria interactions derived from enclosure studies. *New Zealand Journal of Marine and Freshwater Research*, 21: 477-482
- Carmichael, W.W., 1992. Cyanobacteria secondary metabolites the cyanotoxins. Journal of Applied Bacteriology, **72**: 445 – 459
- Carpenter S. R., J. F. Kitchell & J. R. Hodgson, 1985. Cascading trophic interactions and lake productivity. Bioscience 35: 634–639.
- Casamatta, D.A. and C.E. Wickstrom, 2000. Sensitivity of two disjunct bacterioplankton communities to exudates from cyanobacterim *Microcystis aeruginosa* Küutzing. *Microbial Ecology*, **41**: 64-73
- Chapman, L.J. and D.L. Kramer, 1991. Limnological observations of an intermittent tropical dry forest stream. *Hydrobiol.*, **226**: 153–166
- Constable, M., 1984. Resources of rural development in Ethiopia. Ethiopian highlands reclamation study. Working paper 17, FAO/Ministry of Agriculture, Addis Ababa; Ethiopia
- Carmichael, W.W., S.M.F.O., Azevedo, J.S. An, R.J.R. Molica, E.M. Jochimsen, S. Lau, K.L. Rinehart, G.R. Shaw, and G.K. Eaglesham, 2001. Human fatalities from Cyanobacteria: Chemical and biological evidence for cyanotoxins. *Environmental Health Perspectives*, 109: 663-668

- Carpenter, S.R., J.F. Kitchell, J.R. Hodgson, P.A. Cochran, J.J. Elser, M.M. Elser, D.M. Lodge, D. Kretchmer, X. He, and C.N. Von Ende, 1987. Regulation of lake primary productivity by food web structure. *Ecoloy*, 68: 1863-1876De Wit, J., 2003. Stuwmeren in Tigray (Noord-Ethiopië): kenmerken, sedimentatie en sedimentbronnen. Unpublished M.Sc. thesis, Department of Geography, K.U.Leuven.
- Chorus, I. and J. Bartram (eds.), 1999. Toxic cyanobacteria in water: a guide to their public health consequences, monitoring and management. St Edmundsbury Press, London.
- Dejenie, T., Tsehaye A., L. De Meester, Afework M., Abreha G., S. Risch, A. Pals, K.
 Van der Gucht, W. Vyverman, J. Nysen, S. Deckers, and J. Decklerck, 2008,
 Limnological and ecological characteristics of tropical highland reservoirs in
 Tigray, Northern Ethiopia. *Hydrobiologia*, 610:193–209
- DeMott, W.R., 1999. Foraging strategies and growth inhibition in five daphnids feeding on mixture of a toxic cyanobacterium and a green algae. *Freshwater Biology*, **42:** 263-274
- Deng, D., P. Xie, Q. Zhou, H. Yang, L. Guo and H. Geng, 2008. Field and experimental studies on the combined impacts of cyanobacterial blooms and small algae on crustacean zooplankton in a large, eutrophic, subtropical, Chinese lake. *Limnology*, 9: 1-11
- Dudgeon, D., A.H. Arthington, M.O. Gessner, Z.-I. Kawabata, D.J. Knowler, C. Leveque, R.J. Naiman, A.-H. Prieur-Richard, D. Soto, M.L.J. Stiassny and C.A. Sullivan, 2006. Freshwater biodiversity: importance, threats, status and conservation challenges. *Biological Reviews*, 81: 163-182

- Dumont, H.J., 1994b. On the diversity of the cladocera in the tropics. *Hydrobiologia*, **272**, 27–38
- Elier, A., and S. Bertilsson. 2004, Composition of freshwater bacterial communities associated with cyanobacterial blooms in four Swedish lakes. *Environmental Microbiology*, **6**: 1228 1243
- Elizabeth, K. and Amha, B., 1994. Species composition and phytoplankton biomass in a tropical African lake (Lake Awassa, Ethiopia). *Hydrobiologia*, **288**: 13-32
- El-Swaify, S.A and H. Hurni, 1996. Trans-boundary effects of soil erosion and conservation in the Nile basin. *Land Husbandry*, **1**: 6-21
- FAO, 1986. Highlands reclamation study Ethiopia. Final report, Volumes 1 and 2.Food and Agriculture Organization, Rome.
- Fernando, C.H., 1980a. The freshwater zooplankton of Sri Lanka, with a discussion of tropical freshwater zooplankton composition. *Internationale Revue der Gesamten Hydrobiologie*, **65**: 85–125
- Fernando, C.H., 1980b. The species and size composition of tropical freshwater zooplankton, with special reference to the Oriental region (South East Asia). *Internationale Revue der Gesamten Hydrobiologie*, **65**: 411-426
- Fernando, C.H., J.C. Paggi and R. Rajapaksa, 1987. *Daphnia* in tropical lowlands. *Memorie dell'Istituto Italiano di Idrobiologia*, **45**: 107–141
- Ferrao-Filho, A.S., and S.H.F.O., Azevedo, 2003. Effects of unicellular and colonial forms of toxic *Microcystis aeruginoa* from laboratory cultures and natural populations on tropical cladocerans. *Aquatic Ecology*, **37**: 23-35
- Ferrao-Filho, A.S., S.H.F.O., Azevedo, and W.R. DeMott, 2000. Effects of toxic and non-toxic cyanobacteria on the life history of tropical and temperate cladocerans. *Freshwater Biology*, **45**: 1-19

- Fulton, R.S., and H.W. Paerl, 1987. Toxic and inhibitory effects of the blue green algae *Microcystis aeruginosa* on herbivorous zooplankton. *Journal of Plankton Research*, 9: 837-855
- Gliwicz, Z. M. & W. Lampert, 1990. Food threshold in *Daphnia* species in the absence and presence of blue-green filaments. Ecology 71: 691-702.Gillooly, J.F., and S.I. Dodson, 2000. Latitudinal patterns in the size distribution and seasonal dynamics of new world, freshwater cladocerans. *Limnology and Oceanography*, 45: 22–30
- Geller, W. 1984. A device for the *in situ* measurement of zooplankton food selection, grazing and respiration rates. *Verhandlungen der internationalen Vereinigung fur theoretische und angewandte Limnologie*, **22**:1425-2431
- Green, J., 1976. Changes in the zooplankton of Lakes Mutanda, Bunyoni, and Mulehe (Uganda). *Freshwater Biology*, **6:** 433-436
- Gyllstrom, M., L.-A., Hansson, E. Jeppensen, F. Garci-Criado, E. Gross, K. Irvine, T. Kairesalo, R. Kornijow, M.R. Miracle, N. Nykanea, T. Noges, S. Romo, D. Stephen, E. Van Donk and B. Moss, 2005. The role of climate in shaping zooplankton communities of shallow lakes. *Limnol. Oceanogr.*, 50: 2008-2021
- Haile, T., 1988. Causes and characteristics of drought in Ethiopia. J. Agri. Sci., 10: 85-97
- Haney, J.F., and M.A. Traut, 1985. Size selective grazing by zooplankton in Lake
 Titicaca. Archiv fiir Hydrobiologie, Beiheft Ergebnisse der Limnologie, 21: 141160
- Hartmann, H.J., 1985. Feeding of Daphnia pulicaria and Diaptomus ashlandi on mixtures of unicellular and filamentous algae. *Verhandlungen Internationale Vereinigung für Theoretische und Angewandte Limnologie*, 22: 3178-3183

- Horris, G.P., and G. Baxter, 1996. Interannual variability in phytoplankton biomass and species biomass and species composition in sub-tropical reservoir. *Freshwater Biology*, 35: 545-560
- Hebert, P.D.N., 1978. The population biology of *Daphnia* (Crustacea, Daphniidae). *Biology Review*, 53:387-426

Huffnagel, H. P., 1961. Agriculture in Ethiopia. FAO, Rome, pp. 484.

- Hurni, H., 1993. 'Land Degradation, Famines and Resource Scenarios in Ethiopia', inPimental D. (ed.), World Soil Erosion and Conservation, Cambridge:Cambridge University Press.
- Huszar, V.L.M., L.H.S. Silva, P. Domingos, M. Marinho and S. Melo, 1998.
 Phytoplankton species composition is more sensitive than OECD criteria to the trophic status of three Brazilian tropical lakes. *Hydrobiologia*, 369/370: 59–71
- Jang, M.-H., K. Ha, G.-J. Joo, and N. Takamura, 2003. Toxin production of cyanobacteria is increased by exposure to zooplankton. *Freshwater Biology*.
 48: 1540-1550
- Konopka, A., T. Bercot, and C. Nakatsu, 1999. Bacterioplankton community diversity in a series of thermally stratified lakes. *Microb Ecol*, **38**: 126–135
- Kuiper-Goodman, T., I. Falconer, and J. Fitzgerald, 1999. Human health aspects, in I Chorus and J. Bartram (eds), Toxic cyanobacteria in water: a guide to their public health consequences, monitoring and management.
- Lampert, W. 1981. Inhibitory and toxic effects of blue-green algae on *Daphnia*. *Internationale Revue der gesamten Hydrobiologie*, 66: 285-298
- Langenheder, S., and K. Jürgens, 2001. Regulation of bacterial biomass and community structure by metazoan and protozoan predation. *Limnol. Oceanogr.*, 46:121–134

- Leibold, M.A., 1999. Biodiversity and nutrient enrichment in pond phytoplankton communities. *Evol. Ecol. Res.*, 1: 73-95
- Lo, C., D.C. Bicudo, C.E.M. Bicudo, and L.M. Bini, 2008. Phytoplankton biodiversity changes in shallow tropical reservoir during the hypertrophication process. *Braz. J. Biol.*, 68: 1061-1067
- McCann, J., 1995. People of the plow: An agricultural history of Ethiopia, 1800-1900. University of Wisconsin Press, Madison, USA. 298 pp.
- MNRDEP, 1994. National Conservation Strategy. Volume 2: National Policy in Natural Resources and the Environment. Ministry of Natural Resources Development and Environmental Protection. Addis Ababa.
- Mohamed-Saleem, M.A., 1995. Fragile East African highlands: A development vision for smallholder farmers in the Ethiopian highlands. *Outlook on Agriculture*, 24: 111-116
- Mur, L.R., O.M. Skulberg, and H. Utkilen, 1999. Cyanobacteria in the environment.
 In Toxic Cyanobacteria in Water: a Guide to Their Public Health Consequences, Monitoring and Management. Chorus, I., and J. Bartram, (eds).
 London: E & FN Spon, pp. 15–40.
- Muylaert, K., K. Van der Gucht, N. Vloemans, L. De Meester, M. Gillis, and W. Vyverman, 2002. Relationship between bacterial community composition and bottom-up versus top-down variables in four eutrophic shallow lakes. *Applied and Environmental Microbiology*, 68: 4740 4750
- Nabout, J.C., I. de S. Nogueira, L.G. de Oliveira, and R. R. Morais. 2007. Phytoplankton diversity (alpha, beta, and gamma) from the Araguaia River tropical floodplain lakes (central Brazil). *Hydrobiologia*, **557**:455–461

- Nigussie, H., J. Poesen, J. Nyssen, J. De Wit, Mitiku H., G. Govers and J. Deckers,
 2006. Reservoirs in Tigray: characteristics and sediment deposition problems. *Land Degradation and Development*, 17: 211–230
- Nizan, S., C. Dimentman, M. Shilo, 1986. Acute toxic effects of the cyanobacterium Microcystis aeruginosa on Daphnia magna. Limnology and oceanography **31**: 497-502
- Northcote, T.G., 1988. Fish in the structure and function of freshwater ecosystems: a top-down view. *Canadian Journal of Fisheries and Aquatic Sciences*. **45:** 361-379
- Nyssen, J., H. Vandenreyken, J. Poesen, J. Moeyersons, J. Deckers, Mitiku H., C. Salles, and G. Govers, 2005. Rainfall erosivity and variability in the Northern Ethiopian Highlands. *Journal of Hydrology*, **311**: 172-187
- Pace, M.L. and J.J. Cole, 1994. Comparative and experimental approaches to topdown and bottom-up regulation of bacteria. *Microb. Ecol.*, **28**: 181 – 193
- Persson, A., J.M. Svensson, 2006. Effects of benthivorous fish on biogeochemical processes in lake sediments. *Freshwater Biology*, 51: 1298-1309
- Rejas, D., S. Declerck, J. Auwerkerken, P. Tak & L. De Meester, 2005. Plankton dynamics in a tropical floodplain lake: fish, nutrients, and the relative importance of bottom-up and top-down control. Freshwater Biology 50: 52-69.
- Resson, R., F.S. Soong, J. Fitzgerald, L. Turczynowicz, O. El Saadi, D. Roder, and I.Falconer, 1994. Health Effects of Toxic Cyanobacteria (Blue Green Algae).Australian Government Publishing Service, Canberra, Australia.

- Reynolds, C.S., 1998. What factors influence the species composition of phytoplankton in lakes of different trophic status? *Hydrobiologia*, 369/370: 11 26
- Scheffer, M., R. Portielje, and L. Zambrano, 2003. Fish facilitate wave re-suspension of sediment. *Oceanography*, **48**: 1920 1926
- Sivonen, K. and G. Jones. Toxic ctanobacterial toxins; In Chorus, I. and J. Bartram (eds.), 1999. Toxic cyanobacteria in water: a guide to their public health consequences, monitoring and management. World Health Organization.
- Soininen, J., M. Kokocinski, S. Kotanen, and J. Heino, 2007. Naturality, niches, and determinants of plankton metacommunity structure across boreal wetland ponds. *Ecoscience*, **14**, 146-154
- Sommer, U., Z.M. Gilwicz, W. Lampert, and A. Duncan, 1986. The PEG model of seasonal succession of planktonic events in fresh waters. *Archiv für Hydrobiologie*, 106: 433-471
- Tsehaye A., Tadesse D., Declerck, S., Nyssen, J., Van der Gucht, K., Risch, S.,
 Rousseaux, S., De Wit, J., Afework M., Nigussie H., Abreha G., Poesen, J.,
 Deckers, J., Vyverman, W., De Meester, L., 2007. Ecological atlas of reservoirs
 in Tigray, Northern Ethiopia. Tigray Livelihood Papers No. 4, VLIR, Mekelle
 University IUC Programme and Zala-Daget Project, 80 pp.
- Tilzer, M.M., 1987. Light-dependence of photosynthesis and growth in cyanobacteria: implications for their dominance in eutrophic lakes. *New Zealand Journal of Marine and Freshwater Research*, **21**: 401-412
- Van der Gucht, K., K. Cottenie, K. Muylaert, N. Vloemans, S. Cousin, S. Declerck,E. Jeppesen, J.-M. Conde-Porcuna, K. Schwenk, G. Zwart, H. Degans, W.

Vyverman., L. De Meester, 2007. The power of species sorting: Local factors drive bacterial community composition over a wide range of spatial scales. *PNAS*, **104**, 20404-20409

- Van der Gucht, K., K. Sabbe, L. De Meester, N. Vloemans, G. Zwart, M. Gillis, and W. Vyverman, 2001. Contrasting bacterioplankton community composition and seasonal dynamics in two neighbouring hypertrophic freshwater lakes. *Environ. Microbiol.*, **3**: 680–690
- Van Hannen, E.J., G. Zwart, M.P. Van Agterveld, H.J. Gons, J. Ebert, and H.J. Laanbroek, 1999. Changes in bacterial and eukaryotic community structure after mass lysis of filamentous cyanobacteria associated with viruses. *Appl Environ Microbiol*, 65: 795–801
- Vanni, M.J., C.D. Layne and S.E. Arnott, 1997. Top-down trophic interactions in lakes: effects of fish on nutrient dynamics. *Ecology*, **78** : 1-20
- Vanni, M.J., and C.D. Layne, 1997. Nutrient recycling and herbivory as mechanisms in the top-down effect of fish on algae in lakes. *Ecology*, **78**: 21-40
- van Leeuwen, E., G. Lacerot, E.H. van Nes, L. Hemerik, and M. Scheffer, 2007.
 Reduced top-down control of phytoplankton in warmer climates can be explained by continuous fish reproduction. *Ecological Modelling*, 206: 205-212
- Vanormelingen, P., K. Cottenie, E. Michels, K. Muylaert, W. Vyverman, and L. DeMeester, 2008. The relative importance of dispersal and local processes in structuring phytoplankton communities in a set of highly interconnected ponds. *Freshwater Biology*, 53:2170-2183
- WHO, 2006. Guidelines for drinking-water quality: incorporating first addendum.Vol. 1, Recommendations. 3rd ed. Geneva

- Yirgaw, T., Demeke, A. and Seyoum, M., 2000. The condition factor, feeding and reproductive biology of *Oreochromis niloticus* L. (Pisces: Cichlidae) in Lake Chamo, Ethiopia. *SINET: Ethiopia.J.Sci.*, 23:13-23
- Zohary, T. and R.R. Robarts, 1989. Diurnal mixed layers and the long term dominance of *Microcystis aeruginosa*. In: Zurawell, R.W., H.R. Chen, J.M.
 Burke, and E.E. Prepas, 2005. Hepatotoxic cyanobacteria: A review of the biological importance of microcystins in freshwater environments. *Journal of Toxicology and Environmentl Health-Part B-Critical Reviews*, 8: 1-37

Summary

Ethiopia is the third largest country in Africa with an area of over one million km². It is one of the most populous countries in Africa (probably second), with more than 80 million inhabitants. The Ethiopian highlands represent about 43% of the country but support about 88% of the population. The highlands account for 95% of the regularly cropped land, more that 70% of the livestock population, and 90% of the economic activities of the country. They are considered to be amongst the most degraded lands in Africa by some authors. Rain fed agriculture is the main stay for most farmers. The frequent rainfall anomalies suggest that there are recurrent periods of drought every 3-5 years in the northern parts of Ethiopia and every 6-8 years over the whole country. Part of the variability in the seasonal and annual rainfall across time and space is known to be associated with the El Nino-Southern Oscillation (ENSO) phenomenon. The conditions in Tigray are worse. Extreme spatial and temporal variation in rainfall is characteristic for this region. To tackle the problem associated with the rain fall pattern, several small reservoirs have been constructed over the last two decades. Given the population intensity and long history of agriculture in the highlands of Ethiopia, massive erosion linked to land degradation is a prominent problem. This is expected to bring excessive nutrient to the reservoirs. And many of the reservoirs are expected to be characterized by high nutrient loads and phytoplankton blooms, including cyanobacteria blooms. This has indeed observed in a field survey of reservoirs, most of them suffer from heavy blooms of cyanobacteria.

In this study we started with a field survey of a set of 32 shallow semi-arid sub-tropical reservoirs in the highlands of Tigray, Ethiopia. This survey was carried out in both the wet and dry season to capture seasonal variations of phytoplankton communities and associated environmental variables. We assessed seasonal variation
in more details by monitoring eight selected reservoirs (sub sets of the 32 reservoirs) on a monthly basis during a whole year. We also carried out field enclosure experiments in an effort to better understand the trophic structure of the reservoirs and identify mechanisms that potentially lead to cyanobacterial blooms. First we tested the impact of fish on abiotic conditions in the water column as well as the dynamics of phytoplankton species composition and cyanobacteria biomass. In the second experiment we assessed the potential top-down effects of zooplankton on the phytoplankton communities including the toxic cyanobacteria.

The studied reservoirs were characterized by high nutrient concentrations and high turbidity. Most of the reservoirs harbor the riverine fish *Garra*. Overall, the local phytoplankton richness was low with most reservoirs dominated by a single genus of cyanobacteria (mostly Microcystis), chlorophytes, euglenophytes, cryptophytes or dinophytes. Similarly the bacterial community richness in the studied reservoirs was also low. Lower bacterial taxon richness was encountered in reservoirs with *Microcystis* blooms than bacterial communities in reservoirs without blooms. High altitude reservoirs were more nutrient-rich and associated with high abundances of green algae, euglenophytes or cyanobacteria other than Microcystis. Microcystis was associated with high pH in the rainy and high conductivity in the dry season. Additional factors correlated with Microcystis biomass were Daphnia biomass and possibly altitude and fish biomass. Environmental factors explained the bacterial community composition differently among season. Percentage contribution of Microcystis to the total phytoplankton biomass and copepod biomass showed significant association with the bacterial community composition in the wet season whereas variation in bacterial community composition was associated with total nitrogen (TN), total phosphorus (TP), oxygen, the number of cattle frequenting the reservoir, and fish biomass in the dry season.

Pronounced temporal variation was observed for both biotic and abiotic variables in our study systems. This variation involved both the intra-annual and interannual variations. For the intra-annual variation, the main limnological changes were associated with seasonal differences in rainfall, while also water temperature differed strongly between winter (sub-tropics) and the rest of the year. We observe two minima for phytoplankton biomass: one in winter and a more pronounced one during August. We also observed two main bloom periods for cyanobacteria: one in September-October and a more pronounced one in May-June. Seasonal variation in total phytoplankton and cyanobacterial biomass was, however, not significant.

The first field enclosure experiment was the experiment with fish. The results of this experiment showed that the presence of *Garra* in general increased the amount of suspended matter, nutrient concentrations (total nitrogen and total phosphorus), phytoplankton and to some extent also *Microcystis* biomass (including the proportion of *Microcystis* in the phytoplankton community), and reduced water transparency. The second experiment was carried out to study the effect of zooplankton grazing on phytoplankton community structure, including the relative abundance of toxic cyanobacteria. From this experiment top-down regulation by zooplankton was observed for some of the phytoplankton taxa, including *Anabaena*, Euglenoids, Chlorophytes and Cryptomonads, whereas the impact of the presence of zooplankton on *Microcystis* and *Peridinum* biomass was limited. From the same experiment we observed negative correlation between *Anabaena* and calanoid copepods and *Daphnia carinata*. We also detected microcystin from all experimental units in the second experiment; and higher concentrations were detected in the treatment with than without zooplankton.

We draw some important associations from the field observations and field enclosure experiments for *Microcystis* in our system. The results indicate an inter-play between bottom-up (possibility of *Microcystis* affecting the zooplankton composition) and top-down (zooplankton grazing) regulating the *Microcystis*. Negative association between *Microcystis* and *Daphnia* has been observed mainly from the field survey and to some extent the enclosure experiments with fish and fishless treatment demonstrated a top-down regulation. From these results we can conclude that the zooplankton grazing can not fully regulate *Microcystis*. But, the possibility remains, and is in fact quite realistic, that *Daphnia* also control *Microcystis*, mainly at lower biomasses of *Microcystis* until it reaches a certain biomass and *Microcystis* "escapes" potential control by zooplankton after which it may poison the major grazer zooplankton due to the high densities.

Based on our results of the present study, we put some general suggestions and recommendations for sustainable utilization, maintaining the ecological integrity of the reservoirs and protecting water quality deterioration. The recommendations follow in the following statements. 1) Cyanobacterial monitoring and survey for hazardous effects to assess if the toxins of the organisms are translated into problems of animal or human health should be set-up. 2) Reduction of nutrient loading and sediment input to the reservoirs to curb the eutrophication of the reservoirs. Catchment treatment with reforestation and setting up of exclosures can serve the purpose. 3) Reduction of cattle trampling by restricting cattle access to the reservoirs at selected sites of the reservoir. 4) Reduction of fish (mainly *Garra*). Here we recommend the use of methods to reduce the riverine fish with a high level care to protect the reservoirs from unpredictable consequences like the introduction of exotic fish.

© 2009 Faculteit Wetenschappen. (Gent)

Alle rechten voorbehouden. Niets uit deze uitgave mag worden vermenigvuldigd en/of openbaar gemaakt worden door middel van druk, fotokopie, microfilm, elektronisch of op welke andere wijze ook zonder voorafgaandelijke schriftelike toestemming van de uitgever.

All rights reserved. No part of the publication my be reproduced in any form by print, photo print, microfilm, electronic or any other means without written permission from the publisher.











